

LEUKOCYTE ELASTASE AND ANTI-ELASTASES IN PULMONARY EMPHYSEMA

A thesis submitted by

ROBERT LEO WALSH B.Sc. (Hons), M.Sc.

to the University of Adelaide

South Australia, Australia

for the degree of

DOCTOR OF PHILOSOPHY

Division of Clinical Biochemistry Institute of Medical and Veterinary Science

and

Department of Pathology University of Adelaide

JUNE 2000

ABSTRACT

Although the cause of emphysema is uncertain, the preferred theory to explain the aetiology points to an imbalance in the protease-antiprotease systems within the lung with human leukocyte elastase (HLE) and α_1 -protease inhibitor (α_1 -PI) being the main candidates. However an inherited deficiency of α_1 -PI, is responsible for only a small percentage of all cases of emphysema with the most frequent form occurring in some smokers who have normal α_1 -PI levels. This thesis examines some aspects of the protease-antiprotease theory.

The aim of the first part of this thesis was to examine the proposal that there may be a reduced capacity to inhibit HLE activity in serum and bronchoalveolar (BAL) fluid from emphysema patients and susceptible normal smokers when compared to normal individuals. Comparison groups included children, adult non-smokers, and patients with an acute lung disease, adult respiratory distress syndrome (ARDS). The serum α_1 -PI concentration was significantly increased in children and in adults with emphysema and ARDS. However, no reduction in the HLE inhibitory and the α_1 -PI functional capacity of serum samples from the emphysema and smoking groups was found when compared to the other groups studied, except in the ARDS group, where both were significantly reduced. BAL samples from volunteers were obtained from normal adult non-smokers and smokers and from patients with ARDS. The α_1 -PI concentration in BAL samples from ARDS patients was increased by more than 40 fold but was only 37% functional compared to the 70% and 85% seen in non-smokers and smokers respectively. However no significant reduction in the HLE inhibitory and the α_1 -PI functional capacity of BAL samples from smokers was noted when compared to non-smokers. The presence of other HLE inhibitors in addition to α_1 -PI was demonstrated by increased inhibition towards HLE in all BAL samples. In conclusion, no evidence was found for a reduction in the ability of serum and BAL fluid from emphysema patients and normal smokers to inhibit HLE activity that could explain the increased risk of development of emphysema in smokers.

The second part of this study was to assess the effectiveness of a number of nonspecific and specific inhibitors towards HLE under varying assay conditions including the presence of lung and other surfactants. Overall heparin and the specific HLE inhibitor, ICI 200355, were found to be the most potent inhibitors of HLE using MeO-Suc-A-A-P-V-NA as substrate, although heparin is only most effective as a partial inhibitor (approx. to 25% of residual HLE activity). However, nonspecific proteins such as elastin peptides and even albumin, if present at a high enough concentration (50% inhibition seen with 6 g/l), were found to be capable of partial inhibition of HLE. The effectiveness of glycosaminoglycan (GAG) inhibition of HLE was, under certain conditions, found to be dependent on factors, such as the type of GAG, the HLE assay substrate, the degree of GAG sulphation and the size of the GAG. For example, heparin was the more effective inhibitor with the soluble substrate Suc-A-A-NA while heparan sulphate was more effective with insoluble lung elastin as substrate. In general, less inhibition was found with GAGs that were less sulphated and with low molecular weight GAGs (<2000). Cathepsin G, is not completely inhibited by the major protease inhibitors, α_1 -PI or α_2 -M, unlike HLE, the other major inflammatory serine protease. Pulmonary surfactant had a mild inhibitory effect on HLE activity whereas non-ionic surfactants, such as Brij 35, were found to not only activate HLE but also reduce the effectiveness of some inhibitors, particularly α_2 -macroglobulin and those GAGs which are poor inhibitors of HLE.

In the third part of this study high affinity binding fractions towards HLE were isolated from heparin to determine if heparin has some specificity of binding towards HLE. Sulphate content of the heparin fractions fell with increased binding and inhibition of HLE, which suggests that maximum inhibition and binding depend on an ordered sulphate group sequence rather than simply having the strongest polyanion. Heparinase digestion of heparin bound to HLE allowed the isolation of the most tightly bound heparin saccharide fractions that are likely to contain the high affinity binding sequences for HLE.

In conclusion, the significant findings from this study has shown that there was no reduction in HLE and α_1 –PI functional capacity in adults who smoked or who had emphysema. There was also a significant contribution of other inhibitors in addition to α_1 –PI in all BAL samples examined. In addition to α_1 –PI, a number of other compounds were shown to inhibit HLE, including heparin and heparan sulphate, which is clinically significant because of their strong inhibition of HLE and the large amounts of heparan sulphate proteoglycans and heparin that are found in the lung. This study also highlights the importance of the modifying effects of surfactants, including pulmonary surfactant, towards HLE activity and the function of various inhibitors towards HLE. Finally, a study of the interaction of heparin with HLE has revealed new information on the type and degree of specificity of the binding of heparin for HLE.

DECLARATION

This thesis contains no material which has been accepted for the award of any other degree or diploma in any university or other tertiary institution and, to the best of my knowledge and belief, contains no material previously published or written by another person, except where reference has been made in the text.

I give consent to this copy of my thesis, when deposited in the University Library, being available for loan and photocopying.

Date 22/2/01

Signature

ACKNOWLEDGMENTS

I would like to thank my supervisors Associate Professor Geoffrey McLennan, Dr Edward Cleary, Dr John Edwards and Professor David Thomas. Special thanks are given to Associate Professor Geoffrey McLennan for encouragement and guidance throughout the course of the work and to Dr Edward Cleary for reviewing my thesis drafts. I also would like to thank Professor David Thomas, Head of the Division of Clinical Biochemistry at the outset of the project, for allowing me to undertake the project within the Division at the Institute of Medical and Veterinary Science and Dr John Edwards and his staff for their support and allowing me to use their laboratory in the Enzyme-Protein Unit for most of the project work. Finally I would like to thank my wife, Linette, and our children, James and Briony, for their love and support and tolerating my decision to undertake a part time PhD.

PUBLICATIONS

McLennan G, Walsh RL, Thomas PD, Stevens M. Soluble proteins in bronchoalveolar lavage fluid in the adult respiratory distress syndrome. Am Rev Resp Dis 1984; **129**: A123.

Walsh RL, Dillon TJ, McLennan G. Interference of functional α_1 -protease inhibitor activity by heparin. Am Rev Resp Dis 1988; 137: A208.

Dillon TJ, Walsh R, Scicchitano R, Cleary EG, McLennan G. Sensitive ELISA for measuring elastin breakdown products in plasma. Clin Chem 1990; 36: 392-393.

Walsh RL, Dillon TJ, Scicchitano R, McLennan G. Heparin and heparan sulphate are inhibitors of human leukocyte elastase. Clin Sci 1991; **81**: 341-346.

Walsh RL, Dillon T, Scicchitano R, McLennan G. Interaction of surfactants with heparin and heparan sulphate inhibition of human leukocyte elastase. Am Rev Resp Dis 1991; 143: A326.

Walsh RL, Dillon TJ, McLennan G. A rapid procedure for the measurement of elastase inhibitory function of plasma and bronchoalveolar lavage fluid. Clin Biochem 1992; 25: 263-270.

Dillon TJ, Walsh R., Scicchitano R, Eckhert B, Cleary EG, McLennan G. Plasma elastinderived peptide levels in normal adults, children and emphysematous subjects. Physiologic and computed tomograph scan correlates. Am Rev Respir Dis 1992; **146**: 1143-1148.

Walsh RL, Edwards JB, Holmes M, Scicchitano R, McLennan G. Inhibition of human neutrophil elastase by heparin fractions. Potential implications for therapy of inflammatory lung disease. Am Rev Resp Dis 1993; **147:** A304.

McLennan G, Walsh RL. Soluble proteins in bronchoalveolar lavage and serum in ARDS, IPF, sarcoidosis, and normals. Am Rev Resp Dis 1995; A205.

McLennan G, **Walsh RL**, Robinson, BWS. Bronchoalveolar lavage. In: Kradin RL, Robinson BWS, eds. Immunopathology of lung disease. Boston: Butterworth-Heinemann, 1996: 529-540.

Walsh RL. The role of proteases and antiproteases in emphysema. Clin Biochem Rev 2000; 21: 51-73.

Walsh RL, Dillon TJ, McLennan G. Interaction of surfactants and inhibitors with human leukocyte elastase. Clin Sci 2000; Submitted for publication.

Walsh RL, Wiebkin OW, McLennan G. Evidence for specificity of interaction of heparin with human leukocyte elastase. Clin Sci 2000; Submitted for publication.

PRESENTATIONS

Walsh R.L., Dillon T.J. and McLennan G. Interference of functional α_1 -protease inhibitor activity by heparin. Symposium on Recent Advances in Extracellular Matrix Biology, Lyndoch, South Australia (1988).

Walsh R.L., Dillon T.J. and McLennan G. Heparin inhibition of neutrophil elastase. Thoracic Society of Australia and New Zealand Meeting, Rotorua (1989).

Walsh R.L., Dillon T.J., Scicchitano R. and McLennan G. Heparan sulphate is an endogenous inhibitor of human leukocyte elastase. Thoracic Society of Australia and New Zealand Meeting, Adelaide (1990).

Walsh R.L. Glycosaminoglycan interaction with leukocyte elastase. Annual scientific meeting of the Thoracic Society of Australia and New Zealand, Victor Harbor (1991). An invited oral presentation.

Walsh R.L. Low molecular-weight heparin as anti-inflammatory agents. Symposium on Glycosaminoglycans, inflammation and cancer. An invited oral presentation, Adelaide (1995).

Walsh R.L. Role of heparin as an anti-inflammatory agent: Inhibition of human leukocyte elastase. Seminar. Department of Chemical Pathology, Womens and Childrens Hospital, Adelaide. An invited oral presentation, (1996).

ABBREVIATIONS

| α_1 -ACH | alpha-1-antichymotrypsin |
|------------------|--|
| α_{I} -PI | alpha-1-protease inhibitor |
| α_2 -M | alpha-2-macroglobulin |
| ARDS | adult respiratory distress syndrome |
| BAL | bronchoalveolar lavage |
| COPD | chronic obstructive pulmonary disease |
| CT | computed tomography |
| cpm | counts / minute |
| EC | Enzyme Commission |
| EDTA | ethylene diamine tetra acetic acid |
| ELISA | enzyme linked immunosorbent assay |
| FEV ₁ | forced expiratory volume in 1 second |
| FVC | forced vital capacity |
| GAG | glycosaminoglycan |
| HEPES | N-2-Hydroxyethylpiperazine-N'-2-ethanesulphonic acid |
| HLE | human leukocyte elastase |
| HPLC | high pressure liquid chromatography. |
| K | exponential function describing a pressure volume curve |
| kDa | kilodaltons |
| k _{ass} | binding association constant. |
| K _i | inhibition constant; concentration of inhibitor at which half maximal enzyme velocity is achieved under the stated conditions |
| K _m | Michaelis constant; concentration of substrate at which half maximal enzyme velocity is achieved under the stated conditions, also indicative of the affinity of the substrate. |

| MeO-Suc-A-A-P-V-NA | methoxysuccinyl-(L-alanine) ₂ -L-prolyl-L-valine- <i>p</i> -nitroanilide |
|--------------------|---|
| M/NEI | monocyte/neutrophil elastase inhibitor |
| M _r | molecular radius |
| PAGE | polyacrlyamide electrophoresis. |
| PMN | polymorphonuclear leukocyte |
| PPE | porcine pancreatic elastase |
| SLPI | secretory leukoprotease inhibitor |
| Suc-A-A-NA | n-succinyl-(L-alanine) ₃ -p-nitroanilide |
| TIMPS | tissue inhibitor of metalloproteinases |
| TRIS | tris(hydroxymethyl)aminomethane |
| VGVAPG | valine-glycine-valine-alanine-proline-glycine (elastin hexapeptide) |
| VPGVG | valine-proline-glycine-valine-glycine (elastin pentapeptide) |

ł

TABLE OF CONTENTS

ABSTRACT

DECLARATION

ACKNOWLEDGMENTS

PUBLICATIONS

PRESENTATIONS

ABBREVIATIONS

CHAPTER ONE

THE ROLE OF PROTEASES AND ANTIPROTEASES IN EMPHYSEMA - A REVIEW OF THE LITERATURE.

- **1.1 INTRODUCTION**
- 1.1.1 DEFINITION OF EMPHYSEMA
- 1.1.2. PREVALENCE
- 1.2 EPIDEMIOLOGY OF COPD AND EMPHYSEMA
- **1.3 AETIOLOGY OF EMPHYSEMA**
- 1.3.1 EARLY STUDIES OF EXPERIMENTAL EMPHYSEMA
- 1.3.2 CADMIUM AND EMPHYSEMA
- 1.3.3 EXPERIMENTAL CIGARETTE SMOKE EMPHYSEMA
- 1.3.4 CONNECTIVE TISSUE DEFECTS
- 1.3.5 OXIDANTS
- 1.4 THE PROTEASE-ANTIPROTEASE HYPOTHESIS IN ANIMAL MODELS
- 1.4.1 ENZYME INDUCED EMPHYSEMA

- 1.4.2 LEUKOCYTE ELASTASE INDUCED EMPHYSEMA
- 1.4.3 EMPHYSEMA INDUCED BY PULMONARY NEUTROPHILIA
- 1.4.4 α_1 -PROTEASE INHIBITOR (α_1 -PI) DEFICIENCY
- 1.5 THE PROTEASE-ANTIPROTEASE HYPOTHESIS IN HUMAN STUDIES
- 1.5.1 PULMONARY DEFENCE MECHANISMS
- 1.5.2 INFLAMMATORY CELLS OF THE LUNG
- 1.5.2.1 LEUKOCYTES
- 1.5.2.2 ALVEOLAR MACROPHAGES
- 1.5.3 PROTEASES
- 1.5.3.1 LEUKOCYTE PROTEASES
- 1.5.3.2 MACROPHAGE PROTEASES
- 1.5.3.3 OTHER CELL PROTEASES
- 1.5.4 DIRECT EVIDENCE FOR ELASTASE INVOLVEMENT IN EMPHYSEMA
- 1.5.4.1 PRESENCE OF ELASTASE IN EMPHYSEMATOUS LUNG
- 1.5.4.2 DEGRADATION PRODUCTS FROM ELASTASE ACTIVITY
- 1.5.5 ANTI-PROTEASES
- 1.5.5.1 ALPHA-1-PROTEASE INHIBITOR (α_1 -PI)
- 1.5.5.2 ALPHA-2-MACROGLOBULIN (α_2 -M)
- 1.5.5.3 ALPHA-1-ANTICHYMOTRYPSIN
- 1.5.5.4 SECRETORY LEUKOPROTEASE INHIBITOR
- 1.5.5.5 ELAFIN
- 1.5.5.6 MONOCYTE/NEUTROPHIL ELASTASE INHIBITOR
- 1.5.5.7 OTHER ENDOGENOUS INHIBITORS
- 1.6 THE PROTEASE-ANTIPROTEASE HYPOTHESIS: REDRESSING THE IMBALANCE

- 1.6.1 REDUCING THE PROTEASE LEVELS
- 1.6.2 INCREASING THE ANTIPROTEASE LEVELS
- 1.7 SUMMARY

CHAPTER TWO

MATERIALS AND METHODS

- 2.1 INTRODUCTION
- 2.2 MATERIALS
- 2.2.1 ENZYMES
- 2.2.2 NON-ENZYME PROTEINS AND PEPTIDES
- 2.2.3 GLYCOSAMINOGLYCANS
- 2.2.4 OTHER REAGENTS
- 2.3 METHODS
- 2.3.1 BRONCHOALVEOLAR LAVAGE
- 2.3.2 PREPARATION OF HUMAN LEUKOCYTE ELASTASE-SEPHAROSE GELS.
- 2.3.3 PREPARATION OF [³H]-HEPARIN AND [³H]-HEPARAN SULPHATE.
- 2.3.4 PREPARATION OF ANTITHROMBIN SEPHAROSE
- 2.3.5 PREPARATION OF 12-22 % ACRYLAMIDE GELS
- 2.3.6 ELASTIN PEPTIDES (α-ELASTIN) PREPARATION.
- 2.3.7 ISOLATION OF PULMONARY SURFACTANT.
- 2.3.8 RABBIT ANTISERUM TO ELASTIN PEPTIDES (α -ELASTIN) PREPARATION.
- 2.3.9 AFFINITY CHROMATOGRAPHY OF HEPARIN ON HUMAN LEUKOCYTE ELASTASE-SEPHAROSE

- 2.3.10 AFFINITY CHROMATOGRAPHY OF HEPARIN ON ANTITHROMBIN-SEPHAROSE
- 2.3.11 DIGESTION OF HEPARIN WITH HEPARIN LYASES.
- 2.3.11.1 HEPARIN BOUND TO HUMAN LEUKOCYTE ELASTASE-SEPHAROSE.
- 2.3.11.2 HEPARIN BOUND TO HUMAN LEUKOCYTE ELASTASE
- 2.3.12 ELECTROPHORETIC SEPARATION OF HEPARIN FRACTIONS WITH 12-22 % POLYACRYLAMIDE GEL ELECTROPHORESIS
- 2.3.13 HIGH PRESSURE LIQUID CHROMATOGRAPHY ESTIMATION OF THE SIZE OF ELASTIN PEPTIDES IN THE PRESENCE OF HEPARIN.
- 2.3.14 HIGH PRESSURE LIQUID CHROMATOGRAPHY OF HEPARIN FRACTIONS.
- 2.4 ASSAYS
- 2.4.1 α₁ -PROTEASE INHIBITOR AND ELASTASE INHIBITOR Y FUNCTION OF PLASMA, SERUM, AND BRONCHOALVEOLAR LAVAGE
- 2.4.1.1 INSTRUMENTATION.
- 2.4.1.2 ASSAY STANDARDS.
- 2.4.1.3 SUBJECTS
- 2.4.1.4 ASSAY SAMPLE HANDLING
- 2.4.1.5 ESTIMATION OF α₁-PROTEASE INHIBITOR WITH PORCINE PANCREATIC ELASTASE INHIBITION ASSAY.
- 2.4.1.6 ESTIMATION OF α_1 -PROTEASE INHIBITOR WITH HUMAN LEUKOCYTE ELASTASE INHIBITION ASSAY.
- 2.4.1.7 IMMUNO-TURBIDIMETRIC α_l -PROTEASE INHIBITOR ASSAY.
- 2.4.2 MEASUREMENT OF ELASTASE INHIBITION.
- 2.4.2.1 HUMAN LEUKOCYTE ELASTASE (2.5 μg) WITH SUC-A-A-A-NA AS SUBSTRATE.
- 2.4.2.2 HUMAN LEUKOCYTE ELASTASE (0.5 μg) WITH MeO-SUC-A-A-P-V-NA AS SUBSTRATE (15 μL sample).
- 2.4.2.3 HUMAN LEUKOCYTE ELASTASE (0.5 μg) WITH MeO-SUC-A-A-P-V-NA AS SUBSTRATE (75 μL sample).

- 2.4.2.4 HUMAN LEUKOCYTE ELASTASE (150 ng) WITH MeO-SUC-A-A-P-V-NA AS SUBSTRATE.
- 2.4.2.5 HUMAN LEUKOCYTE ELASTASE WITH HUMAN LUNG ELASTIN AS SUBSTRATE.
- 2.4.2.6 PORCINE PANCREATIC ELASTASE WITH SUC-A-A-NA AS SUBSTRATE.
- 2.4.3 MEASUREMENT OF CATHEPSIN G, CHYMOTRYPSIN AND TRYPSIN INHIBITION.
- 2.4.3.1 CATHEPSIN G AND CHYMOTRYPSIN ASSAY.
- 2.4.3.2 TRYPSIN ASSAY.
- 2.4.4 ELISA FOR ELASTIN PEPTIDES.
- 2.4.5 URONIC ACID CONTENT OF GLYCOSAMINOGLYCANS.
- 2.4.6 TOTAL SULPHATE CONTENT OF GLYCOSAMINOGLYCANS.
- 2.4.7 QUANTITATION OF SULPHATED GLYCOSAMINOGLYCANS.
- 2.4.8 ASSAY FOR THE DETERMINATION OF THE KINETIC CONSTANTS OF HLE.
- 2.5 DATA INTERPRETATION AND STATISTICAL ANALYSIS

CHAPTER THREE

ANTI-ELASTASE ACTIVITY OF PLASMA, SERUM AND BRONCHOALVEOLAR LAVAGE

- 3.1 INTRODUCTION
- **3.2 ASSAY EVALUATION**
- **3.3 EFFECT OF ANTI-COAGULANTS**
- 3.4 SERUM α_1 PROTEASE INHIBITOR CONCENTRATIONS
- 3.5 COMPARISON OF α_1 PROTEASE INHIBITOR CONCENTRATION IN SERUM WITH BRONCHOALVEOLAR LAVAGE FLUID

3.6 **DISCUSSION**

CHAPTER FOUR

ELASTASE INHIBITORS

- 4.1 INTRODUCTION
- 4.2 DETERMINATION OF HUMAN LEUKOCYTE ELASTASE ACTIVITY

4.3 **PROTEIN INHIBITION**

- 4.3.1 INHIBITION OF HUMAN LEUKOCYTE ELASTASE
- 4.3.2 INHIBITION OF CATHEPSIN G
- 4.4 HUMAN LEUKOCYTE ELASTASE SPECIFIC SYNTHETIC INHIBITORS
- 4.4.1 INHIBITION OF HUMAN LEUKOCYTE ELASTASE WITH MeO-Suc-A-A-P-V- NA AS SUBSTRATE
- 4.4.2 INHIBITION OF HUMAN LEUKOCYTE ELASTASE WITH HUMAN LUNG ELASTIN AS SUBSTRATE
- 4.4.3 INHIBITION OF CATHEPSIN G
- 4.5 GLYCOSAMINOGLYCAN INHIBITION
- 4.5.1 INHIBITION OF HUMAN LEUKOCYTE ELASTASE WITH Suc-A-A-NA AS SUBSTRATE
- 4.5.1.1 TYPE OF GLYCOSAMINOGLYCAN
- 4.5.1.2 SOURCE OF HEPARIN AND HEPARAN SULPHATE
- 4.5.1.3. MOLECULAR WEIGHT OF HEPARIN
- 4.5.1.4. ASSAY CONDITIONS
- 4.5.2 INHIBITION OF HUMAN LEUKOCYTE ELASTASE WITH HUMAN LUNG ELASTIN AS SUBSTRATE
- 4.5.3 INHIBITION OF CATHEPSIN G

4.6 INCUBATION TIME OF INHIBITORS

- 4.6.1 INHIBITION OF HUMAN LEUKOCYTE ELASTASE
- 4.6.2 INHIBITION OF CATHEPSIN G
- 4.7 ORDER OF ADDITION OF INHIBITOR, ENZYME AND SUBSTATE
- 4.8 INTERACTION OF HUMAN LEUKOCYTE ELASTASE AND CATHEPSIN G
- 4.9 COMPARISON OF EFFECTIVENESS OF INHIBITORS
- 4.10 **DISCUSSION**

CHAPTER FIVE

INTERACTION OF SURFACTANTS ON ELASTASE AND ELASTASE INHIBITORS

- 5.1 INTRODUCTION
- 5.3 EFFECT OF SURFACTANTS ON HUMAN LEUKOCYTE ELASTASE ACTIVITY
- 5.2.1 MeO-Suc-A-A-P-V-NA AS SUBSTRATE
- 5.2.2 HUMAN LUNG ELASTIN AS SUBSTRATE
- 5.3 EFFECT OF BRIJ 35 ON HUMAN LEUKOCYTE ELASTASE INHIBITORS
- 5.3.1 PROTEIN INHIBITION
- 5.3.2 HUMAN LEUKOCYTE ELASTASE SPECIFIC SYNTHETIC INHIBITORS
- 5.3.3 GLYCOSAMINOGLYCAN INHIBITION
- 5.4 EFFECT OF LUNG SURFACTANT ON HUMAN LEUKOCYTE ELASTASE INHIBITORS
- 5.5 **DISCUSSION**

CHAPTER SIX

ISOLATION OF HIGH AFFINITY HEPARIN FRACTIONS TOWARDS HUMAN LEUKOCYTE ELASTASE

6.1 **INTRODUCTION**

- 6.2 AFFINITY CHROMATOGRAPHY OF HEPARIN ON HUMAN LEUKOCYTE ELASTASE-SEPHAROSE
- 6.3 AFFINITY CHROMATOGRAPHY OF HEPARIN ON ANTITHROMBIN-SEPHAROSE
- 6.4 ISOLATION OF HLE BINDING FRACTIONS FROM HEPARINASE DIGESTION OF HUMAN LEUKOCYTE ELASTASE BOUND HEPARIN
- 6.4.1 HEPARIN BOUND TO FREE HUMAN LEUKOCYTE ELASTASE
- 6.4.2 HEPARIN BOUND TO HUMAN LEUKOCYTE ELASTASE-SEPHAROSE
- 6.5 **DISCUSSION**

CHAPTER SEVEN

CONCLUDING REMARKS

7.1 SYNOPSIS AND CONCLUSIONS OF THIS THESIS

- 7.1.1 FUNCTIONAL α₁ PROTEASE INHIBITOR CONCENTRATIONS IN SERUM AND BRONCHOALVEOLAR LAVAGE FLUID
- 7.1.2 ELASTASE INHIBITORS
- 7.1.3 EVIDENCE FOR SPECIFICITY OF INTERACTION OF HEPARIN WITH HLE
- 7.2 FUTURE STUDIES

BIBLIOGRAPHY

CHAPTER ONE



THE ROLE OF PROTEASES AND ANTIPROTEASES IN EMPHYSEMA - A REVIEW OF THE LITERATURE.

1.1 INTRODUCTION

Chronic obstructive pulmonary disease (COPD), of which emphysema is one component, is a major cause of morbidity and mortality. In Europe, together with asthma and pneumonia, COPD is the third most common cause of death (Siafakas et al, 1995). In the USA, COPD is currently the fourth leading cause of death (Mortality Patterns - United States, 1993). It is also a major cause of morbidity as demonstrated by the fact that more than 5.4 million Americans were estimated to have COPD in 1985, and it accounted for 13% of hospital admissions in the USA (Feinleib et al., 1989). The majority of cases occur in patients who have been or are currently smoking. Thus, prevention of the development of COPD by reducing the prevalence of smoking remains a priority. While the prevalence of smoking in developed countries, such as the USA, has fallen from 52.1% in 1965 (Smoking and health, 1964) to 25.7% in 1991 (Cigarette smoking among adults-United States, 1993), the recent large increase in smoking in many countries (MacKenzie et al., 1994) will see COPD becoming an even bigger worldwide problem. However, only a relatively small proportion of smokers develops symptomatic disease (Fletcher and Peto, 1977). As a consequence, there has been considerable interest in not only how smoking causes COPD, but also in identifying other factors (eg. genetic factors, environmental factors) that make individuals susceptible to COPD.

Pulmonary emphysema together with chronic obstructive bronchitis and some obstructive types of bronchiectasis form the group of conditions known as COPD, or alternatively as chronic obstructive lung disease (COLD) or chronic obstructive airways disease (COAD). Acronyms, such as COPD, have come into wide use because the pathophysiology of COPD is uncertain and it is difficult, with the usual clinical and epidemiological tools, to diagnose emphysema accurately and thus to determine the relative roles of emphysema, bronchitis and asthma in causing airway obstruction. However, such acronyms are not ideal terms as chronic bronchitis and emphysema, as they are currently defined, need not necessarily be accompanied by airway obstruction. This will be discussed further in relation to emphysema (See section 1.1.1). Despite the reservation, in this thesis the term COPD is used collectively to describe the clinical groupings.

Emphysema, bronchitis and asthma share a number of epidemiological characteristics such as a higher prevalence in men, smokers, urban dwellers, and poorer social classes. There is also considerable overlap between subjects with asthma and COPD in airway responsiveness, airflow obstruction and pulmonary symptoms (see Table 1.1). The so called "Dutch hypothesis", as first proposed by Orie et al (1961), includes asthma as one of the conditions in a single underlying lung disease mechanism that can lead to chronic airway limitation. However the theory needs to be more extensively tested before it is widely accepted.

The clinical diagnosis of COPD is usually based on the demonstration of irreversible airflow limitation after the exclusion of a few specific conditions (eg. localised disease of the upper airways, bronchiectasis, and cystic fibrosis) that can also cause fixed airway obstruction (American Thoracic Society, 1987). The airflow limitation in COPD is due to a variable combination of airway disease and emphysema. The airway disease component, often referred to as small or peripheral airway disease or chronic obstructive bronchiolitis, consists mainly of decreased luminal diameters resulting from various combinations of increased wall thickening, increased luminal mucus and changes in the lining fluid of the small airways. Emphysema, on the other hand, is defined anatomically by permanent, destructive enlargement of airspaces distal to the terminal bronchioles without obvious fibrosis (see 1.1.1. below). The relative contribution of airway disease and emphysema to airflow limitation is difficult to define *in vivo*.

At present the prognosis for patients with clinically evident COPD remains poor, particularly those with type II respiratory failure and cor pulmonale, where the 5 year survival may be less than 30 %. Some increased survival may be obtained by using continuous long-term, controlled oxygen therapy or with surgical intervention by way of lung transplantation or lung reduction. An understanding of the mechanisms of the development of COPD may allow treatment to arrest the further progress of established disease. More importantly, it could lead to treatment that prevents the development of COPD, or at least of the clinically debilitating stages of the disease.

Pulmonary emphysema is a major component contributing to the morbidity and mortality of COPD and occurs to a greater or lesser extent in all of the COPD subgroups. Despite the importance of pulmonary emphysema as a component of COPD, its pathogenesis remains uncertain. The current concepts about the pathogenesis of emphysema are based on essentially two classical studies. One report demonstrated an increase in emphysema in patients with a decrease in the antiprotease, α_1 -proteinase inhibitor (α_1 -PI) (Laurell & Eriksson, 1963) and the other the induction of emphysema in experimental rats after the intratracheal instillation of the protease, papain (Gross et al, 1964). These findings led to a simple protease-antiprotease model in which an imbalance resulted in increased proteolytic activity, which destroyed the elastic framework and the air space walls of the lung. However, while most of the experimental and epidemiological data support the case for this model, the pathogenetic mechanisms, in practice, appear to be much more complex. For example, this simple model explains the very early development of emphysema in patients with a homozygous α_1 -PI deficiency (PI Z) who smoke, but the situation is not as definitive in those deficient patients who do not smoke where emphysema may not develop for another 10-20 years or not at all, as is the case for most individuals with normal levels of α_1 -PI. Overall, it has been estimated that only 10-15% of individuals with the homozygous form of α_1 -PI deficiency develop emphysema (Knight et al., 1997). Thus the absence of α_1 -PI does not necessarily lead to the development of emphysema

and suggests that factors additional to α_1 -PI deficiency are required for the development of the disease.

A long series of experiments with animal models has demonstrated that elastolytic enzymes, introduced via the trachea, can produce emphysema (Snider et al., 1986; Evans and Pryor, 1994). The majority of the evidence suggests that human leukocyte elastase (HLE), the physiological target of α_1 -PI (Beatty et al., 1980; Longstaff and Gaffney, 1991), is the elastolytic enzyme responsible for the degradation of lung tissue in emphysema, although any of the elastolytic enzymes that can potentially be released in the lung could have a role.

The theory has been modified to explain smoking-induced emphysema as discussed later in this chapter (See section 1.5). This includes a probable increase, in the lung, of oxidants and free radicals from tobacco smoke as well from greater number of activated neutrophils and macrophages found in the lungs of smokers. These oxidants may produce tissue damage directly or by partially inactivating α_1 -PI (Evans and Pryor, 1994; Hoidal et al., 1981). The elastolytic load on the lungs is also probably increased due to the increased number of activated neutrophils (McGowan et al., 1983; Hunninghake and Crystal, 1983; Hoidal and Niewoehner, 1983a). While experiments with animal models have demonstrated that exposure to tobacco can increase the amount of emphysema in the presence of tracheally-induced elastolytic enzymes (Hoidal and Niewoehner, 1983b, Janoff, 1985), tobacco smoke, by itself, has not been shown conclusively to induce emphysema in animal models. The majority of the studies of the pathogenesis of smoking-induced emphysema in humans have simply compared smokers with non-smokers, but because of the wide variability in susceptibility of individual smokers to develop progressive COPD, most of the results have been inconsistent. Several studies have suggested that genetic factors, other than α_1 -PI type, may be involved in the susceptibility to develop COPD. Other than smoking and α_1 -PI deficiency, the other major risk factor for which there is clear evidence is occupational exposure, in particular, coal and gold mining (Oxman et

al., 1993). Until some of the other unidentified host factors for the development of COPD are identified, the exact pathogenesis of smoking-induced emphysema will continue to be unknown.

The elucidation of the mechanisms involved in the development of emphysema may depend on the use of suitable animal models such as transgenic mice in which the protease and antiprotease levels can be changed. While the expression of α_1 -PI in mice appears to be different from that which occurs in humans (Tardiff and Krauter, 1998), the different forms of genetic emphysema seen in the tight-skin, pallid, and beige strains of mice (Keil et al, 1996) lend support the notion that the pathogenesis of emphysema is multifactorial.

This chapter reviews the epidemiology of emphysema, and discusses the mechanisms proposed to explain its pathophysiology. The protease-antiprotease model is discussed and the evidence supporting the hypothesis in humans and in animal models is reviewed and evaluated. In particular, the central position in this hypothesis of the protease human leukocyte elastase (HLE) and the antiprotease α_1 -PI will be carefully considered, as the relationship between these two components is the basis for a major part of the work reported in this thesis.

1.1.1 DEFINITION OF EMPHYSEMA

Emphysema has been recognised since the 1800's, but a definition that was widely accepted did not evolve until the 1959 Ciba Guest Symposium where the participants agreed to adopt, as a working definition of emphysema, "a condition of the lung characterised by increase beyond the normal in the size of the air spaces distal to the terminal bronchiole either from dilatation or from destruction of their walls". This anatomical rather than clinical diagnosis, while being somewhat difficult to confirm during life, nevertheless led to a useful distinction from the often associated chronic bronchitis, both as regards aetiology and treatment. However, this definition did not distinguish between overinflation such as is seen in acute asthma, the dilatation of the peripheral airspaces seen in the ageing lung, or the destruction of the lung architecture which occurs in smoking-related emphysema.

Since then the agreed definition has been modified on two occasions to emphasise the destructive process in emphysema (American Thoracic Society, 1962; Snider et al., 1985). The most recent definition states that emphysema is "A condition of the lung characterised by abnormal, permanent enlargement of the air spaces distal to the terminal bronchiole, accompanied by destruction of their walls and without obvious fibrosis" (Snider et al., 1985). The later definitions exclude many conditions which were originally included under the original definition of emphysema (Reid, 1967) and they require a tighter definition of lung normality. For example, senile lungs are characterised by a homogeneous enlargement of the alveolar airspace with some functional characteristics of emphysematous lungs (Verbeken et al., 1992), but emphysema according, to the later definitions, can only be said to be present where destruction of alveolar walls can be demonstrated.

Although tightening the definition in this way has helped promote research and understanding as to the causes of emphysema, it has brought with it additional problems in demonstrating the presence of emphysema. At present, the diagnosis of emphysema, based on the most recent definition (Snider et al., 1985), relies on the interpretation of structural changes in lung tissue. In practice, this is only available on surgically resected material, such as lung removed prior to lung transplantation, and on lungs studied at postmortem. In living subjects, chest X-rays only show a moderate correlation with macroscopic emphysema (Thurlbeck and Simon, 1978). However computed tomography (CT) scan methods correlate well with quantitative histologic assessment (Gould et al., 1988; MacNee et al., 1991) even within the range of those showing normal age change or early emphysema. Despite this, correlation of CT data with clinical assessment is poor, although it does have a better correlation with the results of pulmonary function tests (Gould et al., 1988; MacNee et al., 1991; Heremans et al., 1992; Hruban et al., 1987).

The early stages of emphysema are virtually impossible to detect clinically and currently detection at this level can only be effectively achieved at the microscopic level. This is because normal alveolar air spaces are approximately 250 µm in diameter and are thus not visible to the naked eye. For emphysema airspaces to be visible to the naked eye, they must be greater than approximately 1 mm in diameter, at which stage three-quarters of the alveolar surface is destroyed (Weibel, 1963; Schreider and Raabe, 1957). At present, high resolution computed tomography (CT) scans are unable to detect the early stages of emphysema so that improved CT or other means will need to be developed to allow early detection of emphysema.

Emphysema is subclassified according to the distribution of abnormal airspaces within the acinus or the respiratory tissue airspaces at a single terminal bronchiole (Ciba Guest Symposium, 1959). For example, centriacinar emphysema begins at the level of respiratory bronchioles and is seen in the form of focal emphysema associated with heavy exposure to coal dust and in the more generalised centrilobular emphysema, which is the most common form seen in smokers. Panacinar emphysema involves all the acinus and is the usual pattern that occurs in homozygous α_1 -PI deficiency and in smokers (Thurlbeck, 1988). The varying distribution of the lesions in emphysema is not only dependent on the causative agent but suggests that they may differ in pathogenesis.

Emphysema may occur with or without airways obstruction. The relative contributions of emphysema and small airways abnormalities in fixed airflow obstruction in COPD remain controversial (Snider, 1992; Wright, 1992). While a number of studies have concluded that emphysema is the most important lesion causing expiratory airflow limitation, an almost equal number concluded that it did not. The most recent study concluded that the cause of the lesions responsible for small airway obstruction still needs to be identified (Gelb et al., 1996).

The destruction of the lung architecture that occurs in emphysema includes elastic fibre damage, an observation that was noted as early as 1907 (Orsos). This results in a loss of lung elasticity and elastic driving force, as reflected in reduced elastic recoil pressure measurements (Silvers et al., 1980; Petty et al., 1987) leading to expiratory airway collapse and chronic airways obstruction (Wilson et al., 1980). The loss of lung elasticity reflects the shift in the lung's pressure volume curve that is best described by an index of lung distensibility, K, an exponential function that describes the shape of the lungs' pressure-volume curve. Colebatch et al. (1979a, 1979b) originally demonstrated in patients with a clinical diagnosis of emphysema that values of K consistently fall well outside the age-related normal range. Later studies have demonstrated that K is a better predictor of emphysema than other measures of elastic recoil (e.g. Pare et al., 1982) but that it has poor sensitivity for detecting mild disease. This loss of lung elasticity together with the loss of alveolar capillaries as part of the destructive process in emphysema results in a mismatch of ventilation and perfusion and a widened arterial-alveolar difference in oxygen tension.

In conclusion, the more recent definitions of emphysema that include the destructive component of this disease are preferable, as they emphasise the damage to lung tissue that results in a loss of lung elasticity. As emphysema is difficult to detect in its early stages it is included for clinical convenience under COPD, even though emphysema may prove not to have an obstructive component.

1.1.2 PREVALENCE

Figures for incidence and prevalence of emphysema are very unreliable because, as mentioned above, it has been difficult to agree on criteria by which to make the diagnosis during life. Since the diagnosis is defined in anatomical rather than physiological terms, the best data come from post mortem surveys, where it has been found that most adult lungs contain some degree of

8

emphysema with the incidence increasing in the fifth to seventh decade (Thurlbeck et al., 1974). Approximately 2/3 of adult males and 1/4 of adult females will have well-defined emphysema at autopsy, although the emphysema is often limited in extent (Ingram, 1994). Therefore, it is likely that almost half of all living elderly adults have well-defined emphysema, with the remainder having some degree of emphysema with, in most cases, no apparent disability or symptoms.

The measurement of lung function with FEV_1 (forced expiratory volume in 1 sec.) has been shown to be a strong predictor of mortality from COPD (Peto et al., 1983). In a large prospective community health study (Higgins et al., 1982) airway obstruction was defined as an FEV_1/FVC (forced vital capacity) ratio < 65%. In this study airway obstruction was found in 16.3% of males aged 45-54 years. An unknown proportion of this community would be expected to have emphysema, but no figures are available in this study for the incidence of clinically significant emphysema.

The death rate attributed to COPD in different countries can vary by as much as four-fold from the lowest in Southern European countries to the highest in the United Kingdom and Eastern European countries, with the U.S.A. lying in the middle (Lamb, 1996). The incidence of emphysema varies from that of the COPD death rates, for which the lowest rates were reported in Japan and New Guinea and the highest rates were reported in the United Kingdom and the U.S.A. (Lamb, 1996). The cause of these international variations is unclear but may result in part from differences in diagnostic accuracy, as discussed in a recent review by Lamb (1996). While tobacco smoking is a major cause of COPD, international differences associated with atmospheric pollution and regions of heavy industry probably also play a role.

 Table 1.1: SUMMARY OF RISK FACTORS FOR COPD

 (Adapted from a review by Silverman and Speizer (1996). Other references listed below.)

| RISK FACTOR | COMMENT |
|-------------------------------|--|
| Cigarette smoking | The major risk factor responsible for an estimated 95% of emphysema cases (Snider, 1986). Only 10-20% of tobacco smokers develop disease (Tetley, 1993). Other factors (Bascom, 1991; MacKenzie et al., 1994)? |
| Airway responsiveness | Significant correlation with decline in pulmonary function (Thurlbeck, 1995). Increased incidence in α_1 -PI deficiency (Colp et al, 1993; Eden et al, 1997). Possible common mechanism to asthma ? |
| Respiratory infections | Significant correlation with childhood acute respiratory infections but less so in adults (Barker et al., 1991; Johnston et al., 1998; Shaheen et al., 1998; Berglund et al., 1999) |
| Occupational exposures | Increased risk shown with coal and gold mining, mineral dust, cadmium fumes, and wood dust (Noertjojo et al., 1996; Hendrick, 1996). Appears additive to smoking effect. |
| Ambient air pollution | Increased risk in urban compared to country environments. No clear link to any specific pollutant except, perhaps, domestic wood smoke (Chen et al., 1990; Pérez-Padilla et al., 1996). |
| Passive smoking exposure | Increased risk in children and <i>in utero</i> for reduced pulmonary function. Uncertain in adults. |
| α ₁ -PI deficiency | Definite risk with PI types ZZ, Znull, and null-null, particularly in smokers (Seersholm et al., 1995; Piitulainen and Eriksen, 1999). Some evidence for risk with MZ and SZ (Tarjan et al., 1994; Turino et al., 1996). Variable expression – other factors (Poller et al, 1990a)? Accounts for about 1% of all emphysema (Tetley, 1993). |
| Other genetic factors | Association with α_1 -ACH deficiency (Eriksson et al., 1986; Lindmark et al., 1990; Poller et al., 1993). Possible association with ABO, secretor and Lewis blood group antigens (Kauffmann et al., 1996). Other inherited factors as yet unidentified. |
| Age | Increased risk due to normal loss of pulmonary function (Verbeken et al., 1992). Cumulative effect with smoking. |

1.2 EPIDEMIOLOGY OF COPD AND EMPHYSEMA

A comprehensive list of the risk factors associated with COPD, of which emphysema is one component, is shown in Table 1.1. It can be seen from this summary, that eigarette smoking is clearly the most important in not only being responsible for the majority of COPD, but that its effect appears to be additive to many of the other risk factors. Longitudinal studies have shown an accelerated decline in FEV₁, in a dose-response relationship to cigarette smoking, associated with both the duration of smoking and the amount smoked (Burrows et al., 1977; Dockery et al., 1988). However, cigarette smokers show wide variation in their susceptibility to develop progressive COPD, with only 10-20% of tobacco smokers developing clinically evident emphysema (Bascom, 1991; Tetley, 1993). Of the factors evaluated, the pack-years of cigarette smoking were the most highly significant predictor, with a correlation coefficient of 0.38. This low correlation suggests that additional environmental factors and familial or genetic factors may contribute to the impact of smoking on the development of airflow obstruction (Pryde and Burrows, 1995). The additional risk factors that can be identified, such as occupation and environment, chest infections and α_1 -PI phenotype, do not explain the individual susceptibility of some smokers to develop COPD (Snider, 1986).

Of the genetic factors, there is the possibility that some forms of α_1 -ACH deficiency, the possession of some blood group antigen combinations and other as yet unidentified familial components, may contribute to the risk of COPD and emphysema. Nevertheless no additional clinically proven genetic factors other than α_1 -PI have, as yet, been described. In most of the studies of α_1 -PI deficiency, smoking is the overriding risk factor in that it increases the incidence of COPD over and above that due to the effect of the particular α_1 -PI phenotype. Thus cigarette smoke appears to be able to tilt the balance of the protease-antiproteases towards lung injury particularly where antiprotease function is compromised such as in α_1 -PI deficiency.

In conclusion, while the overwhelming majority of emphysema is due to smoking, it is not a simple relationship as demonstrated by the fact that not all individuals who smoke develop clinically evident emphysema. A much better understanding of the mechanisms of how emphysema develops in susceptible smokers in comparison to non-susceptible individuals is required, together with the discovery of the interplay of other, perhaps unknown, risk factors in each individual.

1.3 AETIOLOGY OF EMPHYSEMA

1.3.1 EARLY STUDIES OF EXPERIMENTAL EMPHYSEMA

There were two main theories regarding the development of emphysema during the 19th and early 20th centuries: the mechanical and the ischaemic (or atrophic) theories. The mechanical theory reasoned that upper lobe alveolar destruction and dilatation were caused by repeated coughing against obstructed airways, which trapped air distally under pressure, stretching and rupturing the alveoli (Laennec, 1835; Mendelssohn, 1845). Experimental work in animals with mechanical or surgical narrowing of the upper airways or ball valve implantation into the bronchi led to the development of alveolar distension distal to the obstruction (Loeb LM, 1930). However, it was concluded that the animals did not develop the appropriate mechanical or histological features of emphysema and that large airway obstruction alone was not sufficient to cause emphysema.

The ischaemic theory proposed that thrombosis of alveolar capillaries traversing dilated air sacs resulted in impaired nutrition and ischaemic necrosis of the alveolar septa (Waters, 1862). Cigarette smoking was thought to cause *in situ* thrombosis of these capillaries. To test this theory, glass microspheres were used to occlude small lung blood vessels in an animal model study but the resultant ischaemia failed to produce emphysema (Wright and Klienerman, 1963).

In another study, chlorpromazine infusion into the bronchial arteries of horses was reported to be able to produce an emphysema-like lesion (Edwards, 1961). However, the development of emphysema was preceded by a severe arteritis of the bronchial arteries, which is not seen in natural human panlobular emphysema.

In another study a third mechanism was proposed, whereby the emphysema was caused by an immunological response to lung tissue damaged by some unknown agent. In this study, homogenised guinea pig lung tissue, which had been damaged by NO_2 , was injected together with adjuvant, back into guinea pigs. However, this caused an interstitial pneumonitis to develop rather than true emphysema (Crowle, 1959).

In conclusion, none of the above theories is now in vogue, as they do not adequately explain the development of human emphysema. The most likely mechanism proposed that best describes the development of emphysema is provided by the protease-antiprotease hypothesis. The evidence for this conclusion is set out in Sections 1.4 and 1.5.

1.3.2 CADMIUM AND EMPHYSEMA

Cadmium is one of the few occupational agents where the development of COPD appears to be due solely to emphysema. While early reports were controversial due to conflicting data, a definitive study in 1988 (Davison et al.) of 101 workers and ex-workers from a cadmium alloy factory in England clearly linked cadmium fumes to the development of emphysema. However, the mechanisms involved in the development of emphysema due to the presence of cadmium are unclear. Animal studies demonstrate airspace enlargement with fibrosis after intratracheally instilled cadmium chloride (Thurlbeck, 1963) and emphysema-like lesions after multiple aerosol treatment with cadmium chloride (Snider et al., 1973). However, the lesions due to cadmium are not typical of those that are induced by elastolytic enzymes in animal studies as suggested by the finding that elastin was not destroyed in a study with cadmium chloride using radiolabelled neonatal lung elastin in hamsters (Snider et al., 1988).

HLE, the major protease secreted by neutrophils, is now generally considered to be the elastolytic enzyme implicated in pulmonary emphysema (Janoff, 1985; Travis and Salvesen, 1983). However, HLE does not appear to be involved in cadmium-induced lung disease, as demonstrated by a study where hamsters that had been made neutropenic with anti-neutrophil globulin were still able to develop cadmium chloride-induced airspace enlargement (Hoidal et al., 1985).

Cigarettes contain significant amounts of cadmium and as a consequence smokers have been shown to have higher cadmium levels than non-smokers in their lungs and kidneys (Hirst et al., 1973). The authors report that cadmium concentration in the emphysematous lungs of smokers was in direct proportion to the severity of the emphysema. In a later study, Paakko et al. (1989) reported that the cadmium concentration in lung tissue of smokers was in direct proportion to the amount of tobacco smoked and that pulmonary cadmium levels were seen to return to the level of non-smokers in 21-22 years after cessation of smoking. While the presence of cadmium in cigarettes may not produce emphysema on its own, it may contribute to the smoking-induced disease by causing airspace enlargement.

1.3.3 EXPERIMENTAL CIGARETTE SMOKE EMPHYSEMA

As the incidence of emphysema is highest in smokers there have been a number of studies to develop an animal model of smoking-induced emphysema. However, attempts to cause emphysema in animals by prolonged exposure to cigarette smoke have met with mixed success, often producing only goblet cell metaplasia and bronchitis in early studies (Park et al., 1977; Holland et al, 1963). Some of the strongest evidence that laboratory animals develop emphysema is found in a series of studies using beagle dogs (Auerbach et al., 1967; Hammond

et al., 1970; Frasca et al., 1971; Carter et al., 1976). In one study, dogs smoked through tracheostomies for periods of over a year and developed what was thought to be emphysema, but the distribution of the lesions seen was not typical of panlobular emphysema (Auerbach et al., 1967). In a later study, also with tracheostomized dogs, air space enlargement and alveolar wall fibrosis were induced by exposure to the smoke from 2-7 cigarettes per day for 2-4 months (Frasca et al., 1983). Whilst the lungs of these dogs showed significant emphysematous changes, the data were not conclusive because of inadequate controls.

In a carefully controlled study of smoking in rats, both morphometric and pulmonary function tests indicated changes consistent with the development of emphysema in the smoke-exposed animals. However, the period of study was not long enough (6 months) to allow the emphysema-like changes to develop sufficiently so as to be significantly different from the non-smoking controls (Huber et al., 1981). One report has shown the development of significant emphysematous lesions in rats but these developed only after exposure of the rats for more than 12 months to tobacco smoke at the relatively high rate of 10 cigarettes a day, which resulted in a high mortality rate (Heckman and Dalbey, 1982).

More recent studies using the rat, guinea pig or mouse as an animal model of smoking-induced emphysema have given more promising results. When rats were exposed to cigarette smoke for 6 months in one study, there was evidence of emphysema lesions as early as 2 months based on morphometric data from histological sections of the rat lungs (Ofulue and Abboud, 1998). In this last study, the destructive component of emphysema in the smoke-exposed rats was demonstrated by the detection of lung elastin degradation, as measured by the presence of increased levels of elastin breakdown products (elastin derived peptides and desmosine). With the guinea pig as the animal model, long term exposure to cigarette smoke at the rate of 10 cigarettes a day for up to 12 months, has been reported to produce airflow obstruction associated with emphysematous air space enlargement similar to that seen in humans (Wright and Churg, 1990; Wright and Sun, 1994). Mice develop similar pathologic changes to human emphysema after being exposed to two cigarettes a day for 6 months (Shapiro, 1995; Hautamaki et al., 1997).

In summary, it has yet to be conclusively shown that any laboratory animal has developed emphysema, as it occurs in humans, solely in response to cigarette smoke exposure. The failure of tobacco smoke definitively to produce all the characteristics of human emphysema in animal models is probably due to a combination of species difference and the short duration of the studies in comparison to smoking in humans, where it is usually a lifetime habit. For example, the mouse does not have mucus glands in the larger airways and lacks respiratory bronchioles. Besides species differences, individual susceptibility is also a factor, as demonstrated by the fact that only 10-20% of human smokers develop emphysema.

1.3.4 CONNECTIVE TISSUE DEFECTS

The connective tissue in lung has a high content of elastin in numerous elastic fibres that give the lung the elastic properties it requires for expansion and contraction functions, as part of each respiratory cycle. The stronger collagen fibres present are thought to function as the restraining framework providing overall structural stability to the organ. The extensive alterations in the lung's connective tissue framework, that are seen as air space enlargement and loss of elasticity in emphysema, are generally thought to be destructive and form one of the requirements of the more recent definitions of emphysema (See section 1.1.1). However, studies of the connective tissue framework in human emphysematous lungs have provided little information as to the specific nature of the defect. It is still not clear whether there are any changes in the whole lung content of elastin and collagen (Pierce et al., 1961; Chrzanowski et al., 1980). Although defects have been identified in the elastic fibre network of emphysematous lungs, the abnormalities appear to be localised and their impact on the overall function of the intact lung is not clear (Wright, 1961; Niewoehner and Kleinerman; 1977). Emphysema-like lung abnormalities have been reported to be associated with some rare heritable defects in connective tissue synthesis, as reported in the Marfan syndrome (Dwyer and Troncale, 1965) and in certain extreme recessive forms of 'cutis laxa' (Turner-Stokes et al., 1983). In the case of the Marfan syndrome, it is not clearly established that the occasional bullous emphysema found in this syndrome is a consequence of the disease or some other causes (Streeten et al., 1987). Cutis laxa describes a heterogenous group of clinical and genetic conditions where pulmonary emphysema is not always found (Rybojad et al., 1999). Because of the rarity of these conditions and their diverse presentation, these conditions have provided little information on the connective tissue defects in emphysema.

Experimental animal model studies suggest that extensive tissue remodelling occurs in emphysema induced by intra-tracheal instillation of elastases, after which, rates of connective tissue degradation may be matched by synthesis. For example, exposure of the hamster lung to elastase causes a rapid loss of elastin, but collagen and elastin contents then rapidly increase to levels above baseline, during which time there is continuing airspace enlargement (Kuhn et al., 1976). Thus changes in the structural and spatial arrangements of the connective tissue components in the emphysematous lung may be more relevant than changes in the total amount of the constituents present.

The key role of elastin in maintaining the integrity of the connective tissue in the lung can be inferred from experimental animal studies. For example, blotchy mice have genetically impaired copper transport that affects lysyl oxidase, the copper dependent enzyme responsible for the lysine derived cross-linking of collagen and elastin. These mice develop progressive emphysema (Fisk and Kuhn, 1976). The tight skin mouse has a similar phenotype and has also been used in studies of inherited emphysema (Szapiel et al., 1981). Similar defects can be induced in young animals by using copper deficient diets or by a diet containing β -aminopropionitrile, an inhibitor of lysyl oxidase (Kida and Thurlbeck, 1980; O'Dell et al., 1978). However, no significant effects are apparent in normal, mature animals, given copper

deficient diets or a diet containing β -aminopropionitrile, presumably because of the low turnover rates of lung elastin and collagen in mature animals. Other agents that induce emphysema in animal models, if given with these types of diets, can induce the development of emphysema lesions of far greater severity than those produced in animals on a normal diet. Examples include the intra-tracheal installation of porcine pancreatic elastase (Kuhn and Starcher, 1980) and dietary cadmium (Niewoehner and Hoidal, 1982) when given together with β -aminopropionitrile in the diet. These animal model studies demonstrate that impaired synthesis and cross linking of elastin can lead to emphysema. There is also some evidence that cigarette smoke impairs synthesis of lung collagen and elastin under certain conditions (Laurent et al., 1983; Osman et al., 1985). Whether similar changes occur in human smokers during a lifetime of exposure to the components of tobacco smoke such as cadmium is not known.

1.3.5 OXIDANTS

Nitrogen dioxide (NO₂) is a product of most operations requiring combustion such as motor vehicle emissions (Fishbein, 1993) and is a major component of indoor pollution involving tobacco smoking and the burning of indoor cooking and heating fuels, such as natural gas and kerosene (Samet et al., 1987; Leaderer, 1986). While there are no consistent data linking NO₂ exposure to the reduction of pulmonary function in humans, emphysema-like injuries with destruction of alveoli similar to human centriacinar emphysema have been observed in a number of animal models after chronic long-term exposure to NO₂ (Freeman et al., 1972; Juhos et al., 1980; Freeman and Haydon, 1964; Glasgow et al., 1987; Chitano et al., 1995). A decrease in lung elastin and collagen has been shown in male hamsters, within 4 and 10 days respectively, after exposure to 30 ppm of NO₂ with elastin not returning to control levels until after NO₂ was stopped (Kleinerman and Ip, 1979). During NO₂ exposure, neutrophils accumulate rapidly in the lungs where there is some evidence that elastase is released as they migrate through the lung (Glasgow et al., 1987).
Airspace enlargement has also been shown with exposure of rats to hyperoxia (Riley et al., 1980) and in hamsters exposed to ozone (Lucey et al., 1990) but emphysema-like injuries have not been described. A recent review (Chitano et al., 1995), of studies of the chronic long-term exposure to ozone in experimental animals, has found that the animals mainly develop pulmonary fibrosis. One study has reported the induction of emphysematous-like lesions in dogs by using chloramine-T, a chemical oxidant that can inactivate α_1 -PI (Abrams et al., 1981).

Besides the removal of the source of the oxidants, the prevention of oxidant inactivation to the lung could theoretically be prevented or reduced by giving a drug that has antioxidant properties. However, little work has been undertaken with antioxidants, possibly because the evidence for the mechanisms involved remains controversial (Snider et al., 1992). One study using supplementation of the antioxidants, α -tocopherol and β -carotene, did not improve symptoms in patients with COPD (Rautalahti et al, 1997).

The increased risk of COPD in individuals exposed to increased NO_2 , such as in tobacco smoke, wood smoke and industrial pollution (see Table 1.1), is consistent with animal model studies mentioned in this section. Most of the evidence from animal studies supports a similar process to that occurring in human emphysema such that NO_2 induced emphysema has been used as an animal model to study the disease process of emphysema (Snider at al., 1986). The evidence indicates that exposure to increased NO_2 levels could be an important factor in the development of emphysema in humans.

1.4 THE PROTEASE-ANTIPROTEASE HYPOTHESIS IN ANIMAL MODELS

1.4.1 ENZYME INDUCED EMPHYSEMA

The association of low levels of the antiprotease α_1 -PI with emphysema, as found in the homozygous form of α_1 -PI deficiency, led to the proposal that unopposed proteolysis by enzymes released from neutrophils and macrophages was responsible for the development of emphysema. This proteolytic imbalance became known as the protease-antiprotease hypothesis. These observations led to a search for a responsible *in vivo* enzyme, although at that time no mechanism of enzymic digestion of the lung structure had been shown.

A new direction of investigation into the pathogenesis of emphysema was opened up when an experimental model imitating natural emphysema was created, almost accidentally, by Gross (Gross et al., 1964; Gross et al., 1965). He had been attempting to ameliorate the fibrotic effects of inhaled quartz dust by intra-tracheal instillation of the plant enzyme papain, and was surprised instead to note the development of a patchy airspace lesion resembling emphysema. Subsequent investigations of enzyme instillation into the lungs of animal models demonstrated that only elastolytic enzymes can induce experimental emphysema. The elastolytic enzymes obtained from cells that have access to the lung and that can cause experimental emphysema include HLE (discussed below) and more recently, another leukocyte enzyme, proteinase 3 and a macrophage enzyme, cathepsin B. Proteinase 3 (Kao et al., 1988) and cathepsin B (Lesser et al., 1992) have been shown to produce emphysema in hamsters following intratracheal installation. However, trypsin, an enzyme with no elastolytic activity, has also been shown to cause emphysema in rats after a single dose, whether given intravenously, intraperitoneally or endotracheally (Reichart et al., 1992). An explanation for this apparent contradiction, as suggested by the authors, is that the observed pulmonary leucostasis found in these rats may lead to the release of neutrophil products in the lung. These products, which include elastase.

could damage the lung in the presence of depleted antiproteases resulting from the trypsin overload.

Although early findings showed a correlation between *in vitro* elastolytic activity and *in vivo* emphysema-inducing ability (Blackwood et al., 1973; Snider et al., 1974), there are exceptions. One exception is HLE, which causes considerably less severe lesions in the lung than those induced by pancreatic elastase with comparable *in vitro* elastolytic activity. This difference is thought to be due to differences in susceptibility of various lung tissues to the two enzymes and accessibility of the two enzymes to the elastin in the lung tissue as compared to the unimpeded access of the elastases to purified elastin used *in vitro*. An example is the presence in lung tissue of sulphated proteoglycans, which are able to bind to HLE and inhibit HLE activity towards elastin (Lucey et al., 1991). On the other hand, pancreatic elastase is not inhibited by the presence of sulphated proteoglycans and thus is able to maintain its elastolytic activity *in vivo*.

1.4.2 LEUKOCYTE ELASTASE INDUCED EMPHYSEMA

Of all the elastolytic enzymes, HLE has been the most studied and is now generally considered to be the major destructive enzyme implicated in the development of pulmonary emphysema (Janoff, 1985; Travis and Salvesen, 1983). The initial evidence for HLE involvement was suggested by the development of emphysema in some individuals with an inherited deficiency of α_1 -PI, the major serum protease inhibitor of this enzyme. HLE has a broad substrate specificity, which includes the capability to degrade a wide range of extracellular matrix proteins as well as a number of serum proteins (Bieth, 1986; Hubbard et al., 1991b). Thus, HLE activity could account for much of the tissue damage observed in emphysema.

The instillation of HLE into animal lungs has been shown to cause destructive lung disease (Janoff et al., 1977; Senior et al., 1977; Snider et al., 1984). Initially, there is a reduction of lung elastin content at 24 h. followed by a rapid restoration of total lung elastin to normal levels but

with the anatomic arrangement of the new elastic fibres now being grossly disordered (Karlinsky et al., 1983). In elastase-induced emphysema there is also evidence of damage and subsequent repair to the collagen network in the lungs of experimental animals (Kuhn et al., 1976). The result is a picture of permanently increased airspace size and lung volumes, which is typical of human pulmonary emphysema.

1.4.3 EMPHYSEMA INDUCED BY PULMONARY NEUTROPHILIA

Emphysema has been produced both in rhesus monkeys (Wittels et al., 1974) and in dogs in response to repetitive intravenous injections of endotoxin, which caused sequestration of polymorphs in the lung vascular bed (Guenter et al., 1981). Similar observations were made in hamsters, in which the endotoxin was given as multiple intratracheal installations (Rudolphus et al., 1993). Repeated intravenous injections of endotoxin administered to rats over a 10 week period caused emphysema in rats made α_1 -PI deficient by the administration of D-galactosamine (Blackwood et al., 1984). Treatment with endotoxin alone produced milder changes in the rat lungs. In all these experiments it is postulated that the emphysema is induced by the release of enzymes from sequestered neutrophils; however, it has not been proven that elastin destruction occurred in these models.

1.4.4 α_1 -PROTEASE INHIBITOR (α_1 -PI) DEFICIENCY

The only animal models that have been reported to spontaneously develop air space enlargement reminiscent of human emphysema are some strains of mice (pallid, tight-skin, blotchy and beige). Of these strains, only the pallid mouse has been reported to have reduced levels of serum α_1 -PI with spontaneous emphysema occurring late in life, similar to that seen in the inherited α_1 -PI deficiency form of human emphysema. Histological examination of pallid mouse lungs has demonstrated some patchy areas of air-space enlargement with destruction of alveolar septa, a decrease in lung elastin content and the immunological detection of an increased level of elastase bound to alveoli (Martorana et al., 1993). However some differences have been reported between the form of emphysema that is seen in the pallid mouse when compared to human form. For example, there is no increase in neutrophils in BAL (bronchoalveolar lavage) fluid (de Santi et al., 1995) as is seen in non-smoking individuals with the PI Z form of α_1 -PI deficiency (Gadek et al., 1981a; Hubbard et al., 1991c). Another difference is that, unlike humans, there does not appear to be any significant expression of α_1 -PI in the macrophages of mice (Tardiff and Krauter, 1998). Nevertheless, the histological changes reported in pallid mouse are consistent with those found in the inherited form of human emphysema and may provide new insights into the protease-antiprotease hypothesis for the development of the inherited form of human emphysema where there is an increase in protease (elastase) activity due to insufficient antiprotease (α_1 -PI).

1.5 THE PROTEASE-ANTIPROTEASE HYPOTHESIS IN HUMAN STUDIES

The protease-antiprotease or elastase-antielastase hypothesis for the pathogenesis of emphysema arose out of the original report in 1963 by Laurell and Erikssen of an association between emphysema and inherited deficiency of α_1 -antitrypsin (or α_1 -PI in this thesis). Much of the subsequent research on the pathogenesis of emphysema is based on the hypothesis that emphysema results from an imbalance in the protease-antiprotease mechanism which leads to uninhibited activity of enzymes that can hydrolyse elastin in the lung.

Other than the association of a severe form of emphysema in some individuals with rare congenital deficiencies of α_1 -PI, most of the initial support for this theory has come from intrapulmonary instillation of proteases, in particular HLE, into experimental animals leading to anatomic derangements in lung architecture which resemble human emphysema. However,

validation of this hypothesis can only ultimately come from studies of the mechanisms leading to the development of the human form of emphysema with *in vitro* and, if possible, human population studies. In this section, the evidence for an imbalance in the various proteases and antiproteases in the development of human emphysema is examined. As the protease antiprotease imbalance is thought to occur primarily in the lung, the cells in the lung that are capable of releasing proteases, the individual proteases and the antiproteases present in the lung are examined, to establish from current literature, their normal function and possible roles in the pathogenesis of emphysema.

1.5.1 PULMONARY DEFENCE MECHANISMS

The respiratory epithelial surface is continuously exposed to microorganisms, particulates, gases, fumes and aerosols as a result of processing 6-10,000 litres of air daily in an adult. Tobacco and marijuana smoking result in an additional burden of an aerosol containing gas and tar particles. A variety of mechanical, chemical and cellular mechanisms are used to protect the lung surface. During the host defence response, inflammatory cells are attracted to the lung to facilitate the clearance of foreign agents (Harada and Repine, 1985). The three major mediators produced by inflammatory cells are oxygen metabolites, proteases and cationic proteins. The protease component forms a significant part of the defence strategy of inflammatory cells in removing foreign material. However, in excess it can also destroy the extracellular frameworks of the lung, disrupt resident cells, and stimulate further inflammation (Harada and Repine, 1985; Henson and Johnston, 1987). The protease activity is normally tightly controlled by inhibitors so that, if tissue damage has occurred, repair occurs rapidly and little if any permanent damage is seen, as for example, following pneumococcal pneumonia (Haslett, 1992). However, inadequate inhibition or control of protease activity can result in the generation of more inflammatory mediators with continued inflammation and impeded repair of tissue damage.

1.5.2 INFLAMMATORY CELLS OF THE LUNG

1.5.2.1 LEUKOCYTES

All circulating neutrophils pass through the pulmonary capillaries many times each hour and may enter the lungs in large numbers in response to chemotactic gradients (Baggiolini et al., 1992; Oppenheim et al., 1991). These cells have a short half-life of about one to two days after leaving the bone marrow (Bainton, 1992) and are removed by apoptosis, a regulated process which prevents dispersion of harmful cellular components, including proteases, into the surrounding tissue (Haslett, 1992; Haslett, 1997). The rate of apoptosis of neutrophils has been proposed to be regulated by the high concentrations of proteases, such as HLE, at inflammatory sites (Trevani et al., 1996). In the lung, neutrophils are normally present in small numbers in the alveoli, small airways, and pulmonary parenchyma and form only a small percentage of the phagocytic migratory cells in the healthy lung. However, during acute inflammatory stimuli large numbers of these cells rapidly accumulate (Wright et al., 1988). Neutrophils are also found to be increased in lung lavage samples from certain pulmonary inflammatory diseases, such as some acute forms of the adult respiratory distress syndrome (70-95% increase; Mcguire et al., 1982; Lee et al., 1981; Weiland et al., 1986; Idell et al., 1985; Fowler et al., 1982) and cystic fibrosis (50-80% increase; Davis et al., 1983; Fick et al., 1983). The presence of increased numbers of neutrophils should also be obvious in the inherited form of emphysema, if the neutrophil and its proteases are seen to play a role in the development of this long-term chronic lung disease. In non-smoking individuals with the PI Z form of α_1 -PI deficiency, more neutrophils were found in the lower respiratory tract than in normals (Gadek et al., 1981a: Hubbard et al., 1991c). The authors also refer to unpublished data suggesting that individuals with emphysema have a higher number of neutrophils than those free of the disease. They suggest that the protease-antiprotease imbalance in PI Z subjects, where α_1 -PI levels are low, is further exaggerated by an increased HLE burden from the increased numbers of neutrophils.

While the protease burden in the lung is increased with an influx of neutrophils, neutrophil enzymes other than proteases can also contribute to a protease imbalance. An example is myeloperoxidase, whose concentration in the neutrophil is at least twofold higher than that of the serine proteases. Myeloperoxidase, in the presence of H_2O_2 and CI^- , can produce potent oxidising agents, which have the ability to inactivate α_1 -PI and secretory leukoprotease inhibitor (SLPI), resulting in a potential protease-antiprotease imbalance. The oxidants produced by myeloperoxidase are normally regulated by the presence of antioxidants such as superoxide dismutase, glutathione, and catalase thus protecting the lung against both possible proteolytic and oxidant injury (Snider, 1989).

It has long been recognised that in smokers significantly increased numbers of circulating neutrophils are attracted to the lung to facilitate the clearance of irritants in the cigarette smoke (Cone et al., 1971). While in the lungs of smokers there can be an approximately 10 fold increase of absolute neutrophil numbers, the percentage of neutrophils remains unchanged relative to an overall increase in the total number of cells recovered in bronchoalveolar fluid (McGowan et al., 1983; Hunninghake and Crystal, 1983; Hoidal and Niewoehner, 1983a). Thus, in smokers, the lung has not only an increased burden of neutrophils with the accompanying inflammatory products but also a much larger content of another inflammatory cell, the macrophage, as discussed in the following section.

1.5.2.2 ALVEOLAR MACROPHAGES

The most prominent phagocytic cell in normal lung is the alveolar macrophage (almost 90% of total phagocytes), a local cell responsible for the normal function of removing foreign material deposited in the peripheral lung. In contrast to neutrophils, macrophages are long lived; their half-life varies from several days to weeks, depending on the tissue involved. Within the lung, they live about 10 days (Van Furth, 1992). Macrophages, like neutrophils, are attracted to sites of inflammation under the influence of locally produced chemotactic factors, some of which are

selective for macrophages (Baggioloni et al., 1992; Oppenheim et al., 1991). Alveolar macrophages are capable, when activated by surface stimuli, of releasing chemotactic mediators which are likely to account for the increase in neutrophils at the site of inflammation (Baggiolini et al., 1992; Oppenheim et al., 1991). Macrophages also may regulate unwanted neutrophil enzyme activity during resolution of inflammation via apoptosis (Haslett, 1992; Savill et al., 1992). Besides chemotactic factors, macrophages also secrete a neutrophil degranulating agent and can ingest, and later release, HLE.

Cigarette smoking is associated with an approximately 5 to 10 fold increase in cells recovered by BAL. Of these cells, macrophages comprise over 98 %, while leukocytes represent only 1% (Merchant et al., 1992). Macrophages are also prominent in the respiratory bronchioles of cigarette smokers where emphysematous changes are first apparent (Niewoehner et al., 1974). Whilst macrophages appear to be involved in the control of the number of neutrophils in the lung, the extent of their direct involvement in the degradation of connective tissue, including elastin, remains unknown.

1.5.3 PROTEASES

1.5.3.1 LEUKOCYTE PROTEASES

The azurophil (or primary) and specific (or secondary) granules of neutrophils contain compounds responsible for the hydrolytic, proteolytic and oxidative breakdown of phagocytosed material. HLE is the major protease stored in the azurophil granules and can potentially be secreted in millimolar amounts in response to various inflammatory stimuli (Janoff, 1985, Campbell, 1986). HLE is a neutral serine proteinase with a broad substrate specificity which includes the capability of degrading a wide range of extracellular matrix proteins of the lung including elastin, collagen, proteoglycan, fibronectin, and laminin (Hubbard et al., 1991b) as well as lung surfactant (Pison et al., 1989, Lewis et al., 1992), a number of serum proteins such as fibrinolytic and coagulation factors, complement, and immunoglobulins G and M (Hubbard et al., 1991b). Thus uncontrolled HLE activity on its own could account for much of the tissue damage observed in emphysema and in other acute and chronic inflammatory processes (Sanborg and Smolen, 1988; Travis, 1988).

In addition to HLE, azurophil granules also store other neutral serine proteinases (cathepsin G, azurocidin, and proteinase 3) as well as other components such as acid hydrolases, myeloperoxidase, and the defensins. All these components of the azurophil granules are probably released alongside, and work together with, HLE. For example, cathepsin G has been shown to work synergistically with HLE in the degradation of extracellular matrix (Boudier et al., 1991). Like HLE, proteinase 3 (Rao et al., 1991) and, to a lesser degree, cathepsin G (Hubbard et al., 1991b) both have the capacity to degrade extracellular matrix components including elastic fibers. Azurocidin appears to have no enzymatic activity but, like cathepsin G, acts as an antimicrobial agent (Gabay et al., 1989).

Stored within the specific granules of neutrophils are two metalloproteinases; gelatinase B and neutrophil collagenase. Metalloproteinases comprise a family of structurally related zinc dependent enzymes that are able to degrade components of the extracellular matrix (Birkedal-Hansen et al., 1993). These neutrophil proteinases are stored in a proenzyme form, which require activation after secretion. This is usually effected by other neutrophil products such as oxidants generated by stimulated neutrophils (Weiss, 1989). Collagenase can also be activated by plasmin (Murphy and Docherty, 1992) and by limited HLE and cathepsin G activity (Hubbard et al., 1991b). Metalloproteinase activity is specifically inhibited by the tissue inhibitors of metalloproteinases (TIMPS 1 and 2) and by α_2 -macroglobulin, which is found in small amounts in the lung. In combination, activated neutrophil collagenase and gelatinase B can degrade fibronectin and elastin and a wide range of collagens (Murphy and Docherty, 1992). Singularly, however, they are less effective, as each enzyme has a restricted range of extracellular matrix components as substrate. For example, of the metalloproteinases, only gelatinase B can degrade elastin (Senior et al., 1991).

Phagocytosis by neutrophils involves direct recognition or receptor-mediated binding of compounds, particles, or microbes on the neutrophil cell surface, and their uptake into a phagolysosome that has been built by fusion of the intracellular granules with neutrophil cell surface membrane (Sandborg and Smolen, 1988). The phagocytosed compound is then digested in the phagocytic vacuole filled with lysosomal enzymes and oxygen radicals. Neutrophils have also been found to form protected extracellular compartments between their plasma membranes and proteins such as fibrinogen, fibronectin or elastin-coated surfaces. This extracellular compartment is inaccessible to plasma protease inhibitors and allows unrestrained proteolytic activity to occur (Loike et al., 1992). However, neutrophil enzymes and oxygen radicals also appear to escape or be actively exocytosed in regulated amounts to the extracellular space as part of their normal function. For example, the release of extracellular enzymes and oxidants from neutrophils may occur in the following situations:

- In the process of neutrophil migration, to assist the neutrophil's passage through barriers of cells and extracellular matrix in response to chemotactic signals (Glasgow et al., 1987).
- During successful phagocytosis, where the phagolysosyme is not completely closed, allowing granule contents to be released (Klebanoff and Clark, 1978).
- After disintegration of the neutrophil following cell death (Goldstein, 1976).
- During so called "frustrated phagocytosis" where the neutrophil is unable to ingest large foreign bodies (e.g. cigarette smoke tars, dust particles, etc.) or kill and destroy foreign organisms (eg., resistant bacteria) (Klebanoff and Clark, 1978).

In emphysema, there is a continual influx of neutrophils into the lungs in response to the inhalation of cigarette smoke and other likely causative agents, such as environmental and occupational dust and smoke particles. This is thought to result in the release of an excess of extracellular neutrophil proteases and oxidants with the most likely mechanism being "frustrated phagocytosis". In the protease-antiprotease hypothesis, the lungs of emphysematous individuals are in a chronic inflammatory state in which the excess extracellular neutrophil

proteases and oxidants cause damage to lung connective tissue leading to the characteristic irreversible changes seen in emphysema.

1.5.3.2 MACROPHAGE PROTEASES

Human mononuclear phagocytes synthesise and secrete a range of proteinases capable of degrading extracellular matrix. However, the type of proteinase that the phagocyte is capable of producing changes on differentiating from a monocyte to a macrophage. Whereas monocytes produce serine proteinases of the same type that are found in neutrophils, macrophages produce mainly metalloproteinases (Shapiro, 1994). Currently nine macrophage metalloproteinases have been identified, including collagenases, stromelysins, gelatinases, matrilysin and macrophage elastase. Like leukocyte metalloproteinases, the macrophage enzyme activity is zinc dependent and is specifically inhibited by the tissue inhibitors of metalloproteinases (TIMPs 1 and 2). Presumably, such a wide range of metalloproteinases is required by the macrophage for complete tissue degradation as discussed further below.

Macrophage proteinases are thought to be involved in certain chronic inflammatory conditions such as rheumatoid diseases (eg. there is increased collagenase activity in rheumatoid synovial fluids; Birkedal-Hansen et al., 1993) and could play a role in the development of other chronic inflammatory diseases such as emphysema. Cathepsin B, D, L and S are synthesised and secreted by macrophages and may also play a role in elastin degradation (Chapman and Stone, 1984). Macrophages may be involved in proteolytic mechanisms of lung injury since they synthesise and secrete enzymes which, when working together, degrade a spectrum of extracellular matrix components similar to that degraded by neutrophil proteinases (Murphy and Docherty, 1992; Shapiro, 1994; Senior et al., 1991; Murphy et al., 1991).

The finding, that several metalloproteinases have elastolytic activity when working together, and the fact that freshly harvested blood monocytes contain the serine proteinases HLE and

30

cathepsin G (Welgus et al., 1990) suggests that macrophages and monocytes can degrade elastin, and hence contribute to the development of emphysema. It has been found that a subpopulation (20-30%) of circulating monocytes is capable of being recruited rapidly by chemoattractants to sites of inflammation. This subpopulation of monocytes has neutrophil-like pro-inflammatory properties which includes avid adherence to the extracellular matrix, the ability to produce reactive oxygen species, high proteinase content, and proteolytic activity against elastin and fibronectin (Owen et al., 1994a; Owen et al., 1994b). This subpopulation of monocytes may have a role in the lung injury that occurs in emphysema.

Two recent studies by Shapiro and co-workers (Zheng et al., 2000; Wang et al., 2000) in mice lend support to the proposal that macrophage metalloproteinases may have a more significant role in the development of human emphysema. In the first study (Zheng et al., 2000), IL-13, a critical cytokine in asthma, caused emphysema in mice with enhanced lung volumes and compliance, mucus metaplasia and inflammation. The mouse metalloproteinases, MMP-2, -9, -12, -13, and -14 and cathepsins B, S, L, H, and K were all found to be induced by IL-13. In addition, treatment with MMP or cysteine proteinase antagonists significantly decreased the emphysema and inflammation. In the second study (Wang et al., 2000), interferon gamma, a product of CD8(+) lymphocytes, was found to cause emphysema in mice where there was induction and activation of MMP-12 and cathepsins B, H, D, S, and L. Despite there being significant differences in physiology between mice and humans (Shapiro, 1997; Tardiff and Krauter, 1998), these studies suggest that the investigation of macrophage metalloproteinase function in human emphysema should be more closely examined.

1.5.3.3 OTHER CELL PROTEASES

Other inflammatory cells, such as lymphocytes, eosinophils and mast cells, contain proteolytic enzymes, although it is unclear whether they have a role in extracellular matrix turnover. Fibroblasts and other similar mesenchymal cells synthesise metalloproteinases which are required for pulmonary morphogenesis, and remodelling and repair of damaged or regenerating lung tissue (Murphy and Docherty, 1992). Resident alveolar cells of the lung may also be induced to secrete proteinases in inflammatory situations, as has been demonstrated with gelatinase B expression from unidentified cells within the alveolar wall of lungs of cigarette smokers (Shapiro, 1995).

1.5.4 DIRECT EVIDENCE FOR ELASTASE INVOLVEMENT IN EMPHYSEMA 1.5.4.1 PRESENCE OF ELASTASE IN EMPHYSEMATOUS LUNG

An excess of elastase activity in the lung is proposed to lead to emphysema as part of the protease-antiprotease or elastase-antielastase hypothesis. Thus the presence of HLE, the most likely candidate, should be able to be demonstrated in the lungs of emphysematous patients and in smokers, at least in the 10-20% of smokers who develop clinically significant emphysema. However, there are conflicting reports in the literature, not only concerning the role of HLE in the pathogenesis of emphysema, but also the detection of whether there are increased levels of HLE in the emphysematous lung. Most free HLE released into the lung would be rapidly bound by antiproteases with any excess likely to be found attached to the connective tissue of the lung surface. Damiano et al. (1986) has demonstrated the immuno-localisation of HLE on elastic fibres in surgically resected tissue from the lungs of smokers and showed that the amount of HLE bound was proportional to the amount of emphysema that was present. These findings were supported by Ge et al. (1990) but in similar study Fox et al. (1988) reported no increase in HLE in association with elastin in emphysematous lung tissue. It is important to note, in interpreting the data, that the immunological presence of HLE does not necessarily indicate that the HLE is functionally active.

As most free HLE in the lung is rapidly bound by antiproteases, such as α_1 -PI, the simplest way of detecting HLE excess is the immunological measurement of the complexed form of HLE in BAL fluid. A number of studies have used this type of assay. One study has demonstrated a rapid increase in elastase values in BAL from smokers, minutes after cigarette exposure (Fera et al., 1986). Another study demonstrated a relationship between elevated HLE and reduced antielastase levels in BAL fluid and emphysema (Fujita et al., 1990). Increased HLE - α_1 -PI complexes were also found in BAL from both patients with emphysema (Trefz et al., 1992) and subjects with subclinical emphysema (Yoshioka et al., 1995). The condition of subclinical emphysema was defined as those individuals who showed low attenuation areas on CT scans, consistent with the early changes seen in emphysema but who did not demonstrate clinical signs of emphysema (Yoshioka et al., 1995).

Unbound leukocyte proteolytic enzymes, including HLE, are capable of being taken up or "internalised" by macrophages via receptors on the macrophage surface (Campbell et al., 1979; Campbell, 1982; McGowan et al., 1983). The macrophage can subsequently release these "internalised" enzymes in their active form; a process which may contribute to an elastase burden in the lungs of emphysematous individuals. This proposal is supported by the finding that higher elastase activity was found in the alveolar macrophages isolated from individuals with CT scans showing low attenuation areas when compared to alveolar macrophages from individuals with normal CT scans (Betsuyaku et al., 1994).

While some attempts to show increased elastase activity in cigarette smokers have often been inconclusive (Gadek et al., 1979; Carp et al., 1982; Stone et al., 1983), other studies have found small amounts of elastase-like esterase activity in the BAL fluid of smokers compared to nonsmokers (Stockley and Ohlsson, 1982; Janoff et al., 1983a; Niederman et al., 1984). However, in most of these studies the type and source of the elastase activity has not been clearly defined. For example, Janoff et al. (1983a) found that the elastase-like activity was suppressed partially and equally by inhibitors of both metalloproteinases and serine proteases, but surprisingly, not by α_1 -PI. In another study, where elastin was used as substrate, the elastase activity was inhibited more effectively by inhibitors of metalloproteinases than by serine proteases inhibitors (Niederman et al., 1984). The above reports indicate that free elastase activity could be due either to serine proteases or metalloproteinases and that the source might be the leukocyte or the alveolar macrophage.

Some support for macrophage activity in the development of emphysema is suggested by reports, which demonstrate increased levels of cathepsin enzymes in response to tobacco smoke. Cathepsin B levels were found to be approximately three times higher in the macrophages of healthy smokers and approximately ten times higher in the BAL fluid of smokers when compared to non-smokers (Chang et al., 1986). Cathepsin D levels were found to be elevated in the macrophages and BAL fluid of smokers (Chang et al., 1989). Cathepsin L activity is also increased in rat macrophages in response to tobacco smoke (Lesser et al., 1989). While the above reports suggest that the cathepsins may play a role in the inflammatory response of macrophages to tobacco smoke it is not known whether cathepsins have a role in the development of emphysema.

The prevailing evidence suggests that HLE is the prime candidate responsible for the degradation of lung tissue in emphysema. However, because of the chronic nature of the disease, any or more than one of the elastolytic enzymes, that can potentially be released in the lung over a long period of time, may have a role in the development of emphysema (Table 1.2; adapted from Shapiro, 1995; Chapman et al., 1994). The enzyme(s) involved in the degradation of the extracellular matrix of the lung in emphysema may be dependent on the conditions created in the tight junction or "microenvironment", between inflammatory cells such as neutrophils or macrophages and extracellular matrix surfaces. In this "microenvironment", inhibitors such as α_1 -PI may be at reduced levels or excluded and normal physiological conditions, such as pH and oxidant levels, may be altered (Campbell and Campbell, 1988; Weiss and Regiani, 1984). In the unique conditions created in this "microenvironment", certain elastolytic enzyme activity may be favoured which would not be seen under more normal physiological conditions. For example, the neutrophil, which generates oxygen free radicals

when activated, can therefore, at the same time as releasing granule proteinases, "switch off" α_1 -PI and SLPI function by oxidative inactivation of these inhibitors to allow unrestricted HLE activity in this "microenvironment" (Kramps et al., 1991).

1

1.5.4.2 DEGRADATION PRODUCTS FROM ELASTASE ACTIVITY

Because emphysema is a destructive process, quantitation of degradation products from alveolar wall components has been studied as a measure of elastase activity within the lungs. As elastin is a prominent component of the alveolar wall, most of the assays have been developed to measure either elastin-derived degradation products such as desmosine, a cross-linking amino acid unique to elastin, or elastin-derived peptides. A third approach has been to measure the presence of specific fibrinopeptides, which are generated only by the unique action of HLE on fibrinogen.

When elastin is degraded, desmosine is excreted in the urine as one of the components of the peptides produced by elastin degradation. A radioimmunoassay for desmosine in urine reported increased levels in experimental animals treated with intratracheal elastase (Goldstein and Starcher, 1978; Janoff et al., 1983b). However, the same assay for urinary desmosine in human subjects with suspected elastolytic degradation of the lung has given conflicting results. One study showed increased 24 hour urinary excretion of desmosine in 11 subjects with COPD as compared with 23 healthy non-smokers (Harel et al., 1980) but increased levels were not found in two later studies using this assay. In the first of these, on a group of 157 subjects, elevated desmosine levels were not found in the urine of smokers with COPD when compared with normal smokers or non-smokers (Davies et al., 1983). In the second study, no difference in urinary desmosine levels was found between normal individuals, PI Z subjects with emphysema and subjects with interstitial lung diseases; nor were elevated levels found in PI Z children when compared with age-matched control children (Pelham et al., 1985). In a re-evaluation of the radioimmunoassay method, by including preliminary purification steps as part of the assay.

Starcher and Scott (1992) showed that the original radioimmunoassay overestimated urinary desmosine due to the presence in urine of cross-reacting material, whose presence could mask significant differences between subject groups.

At about the same time, Stone et al. (1991) reported an improved method for the assay of urinary desmosine, based on HPLC with preparative chromatography and isotope dilution steps. This procedure was claimed to avoid the measurement of interfering substances detected in the earlier radioimmunoassay method. It gave values that were 10-20% of the previously reported radioimmunoassay values (Stone et al., 1991). Using this HPLC assay, Stone et al. (1995a) reported that urinary desmosine excretion in apparently healthy smokers was 50% higher than in a control group of healthy subjects of similar age who had never smoked. In a later study using this improved assay, together with lung function data from a 12 year longitudinal study, it was shown that smokers with a rapid decline in lung function excreted significantly more desmosine in their urine than did smokers experiencing normal rates of decline (Gottlieb et al., 1996). The results from the improved assay suggest that elevated urinary excretion of desmosine may be an early biologic marker of the development of COPD and lend support to the claim that elastase degradation of elastin is important in the pathogenesis of emphysema.

Besides measurement of desmosine content, elastin peptides have been measured directly with ELISA methods using specific antibodies for elastin peptides. Some of the reported assays have been more successful than others in differentiating COPD subjects from other groups. Patients with COPD have been shown, in several studies, to have elevated plasma elastin peptide levels compared with levels in normal non-smokers (Darnule et al., 1982; Kucich et al., 1985; Dillon et al., 1992; Schriver et al, 1992; Akers et al., 1992). In some of these studies smokers were shown to have higher levels than non-smokers (Kucich et al., 1985; Schriver et al, 1992; Akers et al., 1992). Dillon and co-workers (1992) have reported that plasma elastin peptide levels correlate with the lung function parameter, K, but no correlation was shown in this and another study with the lung function parameter, FEV₁ (Dillon et al., 1992; Frette et al., 1992). Schriver

et al (1992) found significant differences in elastin peptides levels (both in plasma and urine) between subjects with COPD, smokers and non-smokers. Subjects with COPD had the highest levels and smokers had intermediate levels compared to non-smokers. Elastin peptides have also been measured in BAL, where levels were significantly elevated in current smokers compared with age-matched current non-smokers (Betsuyaku et al, 1996).

In contrast to the above reports, one assay for elastin peptides was unable to differentiate between any of the patient and control groups studied which included non-smokers, smokers and individuals with α_1 -PI deficiency (Frette et al., 1992) and patients with emphysema (Frette et al., 1996). In addition, no relationship was found with FEV₁ in non-smokers, smokers and individuals with α_1 -PI deficiency (Frette et al., 1992) and in coal miners (Frette et al., 1997). Although an explanation for these discordant results is not known, notable differences with this assay are that aortic kappa elastin (ethanolic KOH generated elastin peptides) was used to prepare the antisera and that the serum elastin peptide levels found were detected in ug/ml range compared to ng/ml quantities as found in most of the other major reported assays (Kucich et al., 1985; Dillon et al., 1992; Schriver et al, 1992). One possible explanation is that the different antigen used could result in an antibody preparation with significantly different specificities to that used in the other assays.

As neither desmosine nor elastin peptide assays are tissue specific, the site of the excess elastin degradation in smokers and individuals with COPD has not been established. However, in view of the known inflammatory response in the lungs of cigarette smokers, and as the lung is rich in elastic tissue, it must be a favoured site of origin for elastin peptides in smokers. This is supported by the observation that significant increases in elastin degradation products have only been reported in smoking and COPD subjects, and in patients with another degenerative lung disease, cystic fibrosis (Stone et al., 1995b). Elastin peptide levels were reported to be significantly higher in subjects with COPD than in the other lung disease groups studied (pneumonia, asthma, tracheobronchitis). They were also significantly higher than in disease

groups with the potential for non-pulmonary elastin degradation, such as, inflammatory arthritis and atherosclerotic vascular disease (Akers et al., 1992). Further support for the concept that increased elastin degradation products are derived from the lungs in emphysema comes from preliminary evidence that there is a decrease in urinary desmosine excretion in PI Z patients treated with α_1 -PI (Snider et al., 1993; Stone et al., 1995c). All these studies suggest that elastin degradation products are of lung origin and thus would appear to be useful markers of degradative lung disease.

The current working hypothesis for the pathogenesis of emphysema has HLE as the most likely elastolytic enzyme responsible for the degradation of lung tissue in emphysema (Janoff, 1985). To demonstrate that there is increased HLE activity in the pathogenesis of emphysema, Weitz et al. (1986) have developed an assay based on the measurement of fibrinopeptide A α 1-21 and its degradation products which result from the unique capacity of HLE to cleave the Aa21(Val)- Aa22(Glu) bond of fibrinogen. Using this assay as an index of in vivo HLE activity, a five fold elevation of plasma elastase-specific fibrinopeptides was found in a group of smokers compared with non-smokers (Weitz et al., 1987). Further, individuals with α_1 -PI deficiency had plasma elastase-specific fibrinopeptides markedly higher than those in smokers (Weitz et al., 1986; Weitz et al., 1987). In a later study, Weitz et al. (1992) reported that fibrinopeptide levels significantly correlated with the degree of α_1 -PI deficiency, with PiZ individuals having the highest levels and PiMZ individuals having intermediate levels compared to normal PiM subjects. In addition, this study suggests that α_1 -PI deficiency and cigarette smoking have additive effects on plasma elastase-specific fibrinopeptides levels, which would explain why PiZ and some PiMZ subjects are at greater risk for the development of emphysema if they smoke (Weitz et al., 1992). Elastase-specific fibrinopeptides levels in non-smoking PiZ subjects were also found to be inversely related to FEV_1 (Weitz et al., 1992). The assay data from this group provide some of the best direct evidence of in vivo elastolysis,

due to the activity of HLE and is consistent with the concept that unregulated HLE activity causes lung destruction leading to emphysema.

1.5.5 ANTI-PROTEASES

While there is good evidence for the role of α_1 -PI in the inherited form of emphysema, there is still controversy over the relative contribution of other anti-proteases to the more common type of emphysema where α_1 -PI levels are normal. In this section, a number of the candidate anti-proteases, including α_1 -PI, are discussed with a view to their potential role in the pathogenesis of emphysema.

1.5.5.1 ALPHA-1-ANTIPROTEASE (α_1 -PI)

 α_1 -PI is a small 52 kDa glycoprotein which is predominantly produced by the liver, but lesser amounts are produced by a variety of other tissues. The primary function of α_1 -PI is to protect body tissue, particularly that of lung, against the action of HLE. The importance of α_1 -PI is emphasized by the fact that it is one of the few antiproteases that can inhibit HLE, the main protease candidate in this hypothesis.

 α_1 -PI reaches most tissues and organs, including the lower respiratory tract, by simple diffusion from the circulation, although there is evidence that locally produced α_1 -PI may have an important protective function at the tissue site of production (Carlson et al., 1988; Koopman et al., 1989). An example, relevant to the pathogenesis of emphysema, is the expression of α_1 -PI by protease containing inflammatory cells (such as neutrophils, monocytes and macrophages) found in the lung (du Bois et al., 1991; Perlmutter, 1991; Perlmutter and Pierce, 1989). Neutrophils can also store the antiprotease and release it together with its target protease, HLE, when activated (Paakko et al., 1996). The most likely reason for the secretion by the neutrophil of α_1 -PI together with its target enzyme is that it may be part of a mechanism for modulating local HLE activity (Paakko et al., 1996). Thus α_1 -PI provides not only a systemic protective function against HLE activity in the lower respiratory tract, through diffusion of α_1 -PI synthesised by the liver, but also a local protective function at the cellular level where it may regulate HLE activity through its co-secretion with HLE.

In addition to its inhibitory activity towards HLE, α_1 -PI exhibits some activity against most other serine proteases, including trypsin, chymotrypsin, proteinase 3, pancreatic elastase, cathepsin G, plasmin and thrombin. However, its primary physiological function is thought to be the inhibition of HLE (Travis and Salvesen, 1983) as demonstrated, for example, by the fact that α_1 -PI inhibits HLE at least 10 times more rapidly than any other protease tested (Beatty et al., 1980). α_1 -PI functions as a pseudosubstrate by forming a tight equimolar complex with specific target proteases, such as HLE. The reactive site of α_1 -PI, the so called PI residue, is centred on the aminoacid residues Met³⁵⁸ - Ser³⁵⁹ and is part of an external loop protruding from the molecule (Figure 1.1). This Met³⁵⁸ - Ser³⁵⁹ tip fits closely into the reactive pocket of HLE (Figure 1.1). A disadvantage of exposure of the reactive centre on the stressed loop is that the α_1 -PI molecule is liable to inactivation, either by cleavage of the loop or by oxidation of the reactive Met³⁵⁸ to methionine sulphoxide. The latter derivative of the oxidised α_1 -PI molecule is now too large to fit the reactive centre of HLE (Crystal et al., 1989; Carrell, 1986) resulting in a significant reduction of its inhibitory function. For example, the binding association constant (k_{ass}) which describes the binding affinity of α_1 -PI for HLE is reduced after oxidation from 10^7 M^{-1} sec ⁻¹ to $10^4 M^{-1}$ sec ⁻¹ (Beatty et al., 1980).



Figure 1.1 A diagramatic representation of the reactive site of α_1 -PI centred at the stressed loop containing Met³⁵⁸ - Ser³⁵⁹ which is susceptible to cleavage on interaction with the reactive pocket of HLE.

(α_1 -PI can be inactivated by oxidation of the Met³⁵⁸ to methionine sulphoxide.)

As of 1988 (Brantly et al.), more than 75 different PI alleles had been identified, of which the common M allele (frequency greater than 95%) is associated with normal α_1 -PI levels. Only a few alleles result in decreased serum levels of α_1 -PI. The S and Z alleles are associated with slightly and markedly reduced α_1 -PI levels, respectively, while the null alleles are associated with the complete absence of α_1 -PI (Brantly et al., 1988; Crystal et al., 1989). The most studied is the PI Z mutation, which is responsible for more than 95% of all α_1 -PI deficiencies. The PI Z form is also slightly less effective as an inhibitor due to a slower association rate with human leukocyte elastase (HLE) than the normal PI M form of α_1 -PI (Lomas et al., 1993).

 α_1 -PI is the most abundant of the antiproteases in human serum (Carrell, 1986), as well as in most tissues and organs, including the lung. Sampling of the fluid on the epithelial surface of

the lower respiratory tract of normal persons has demonstrated average α_1 -PI levels of 3 to 4 μ M, which represents about 10% of normal serum levels (Wewers et al., 1987). However, even at this concentration, α_1 -PI has been reported to constitute the majority of the anti-elastase activity in the normal lower respiratory tract. For example, more than 80% of the anti-elastase activity in BAL has been attributed to α_1 -PI in some studies (Wewers et al., 1987; Wewers, 1989) and 50% or more in some other studies (Afford et al., 1988; Walsh et al., 1992).

It has been estimated that only about 1% of all emphysema is due to the inherited homozygous α_1 -PI deficiency while the majority of cases (95%) occur in smokers with normal α_1 -PI levels (Tetley, 1993). To include subjects with this more common form of emphysema in the protease-antiprotease hypothesis for emphysema, a "two hit concept" has been proposed in which smoking leads to emphysema by inducing both an increase in elastase activity (see Section 1.5.2) and by decreasing α_1 -PI activity (Wewers, 1989; Hunninghake and Crystal, 1983). It has been proposed that α_1 -PI activity is reduced, along with other susceptible proteases in the lung of smokers, as a result of inactivation by oxygen radicals present in the inhaled tobacco smoke and by the release of oxygen radicals from macrophages and leukocytes (Johnson and Travis, 1979; Carp and Janoff, 1978; Carp et al., 1982).

However the evidence supporting the oxidative inactivation of α_1 -PI in smokers is mixed, with not all reports being able to demonstrate the presence of functionally inactivate or oxidised α_1 -PI. As this is one of areas investigated in this thesis, the evidence will be reviewed briefly. In support of the oxidative proposal is a report that BAL from smokers contained α_1 -PI that was 40% less effective as an inhibitor of HLE than α_1 -PI from non-smokers when compared to the total amount of α_1 -PI present (Gadek et al., 1979). Similar findings were seen in rats after short term exposure to cigarette smoke (Janoff et al., 1979). It has also been shown that the rate of inhibition of HLE by α_1 -PI in BAL fluid from smokers is reduced, as demonstrated by an approximately two fold reduction in the k_{ass} constant of α_1 -PI for HLE in cigarette smokers compared to non-smokers (Wewers et al., 1989). More direct evidence for compromised α_1 -PI function in the lungs of smokers is seen where the α_1 -PI purified from BAL fluid of smokers took twice as long to inhibit HLE as did α_1 -PI purified from BAL fluid of non-smokers (Ogushi et al., 1991). Oxidative damage to the α_1 -PI molecule as a result of smoking has also been demonstrated by the isolation of α_1 -PI containing methionine sulphoxide from lung secretions of cigarette smokers but not from the lung secretions of non smokers (Carp et al., 1982). While oxidative inactivation of α_1 -PI is most likely to be detected in the lung lining fluid which comes into contact with tobacco smoke, some workers have been able to detect small amounts of α_1 -PI inactive towards elastase in serum or plasma from smokers (Beatty et al., 1982; Cox et al., 1984; Bridges et al., 1985).

While the above reports support the proposition that α_1 -PI activity in smokers is functionally compromised, other studies using similar methods, have failed to find any noticeable differences in the elastase inhibitory capacity of α_1 -PI obtained by BAL from smokers and nonsmokers (Stone et al., 1983; Boudier et al.; 1983; Stockley and Afford, 1984; Abboud et al., 1985; Walsh et al., 1992). Even the use of specific monoclonal antibodies to differentiate the native and oxidised forms of α_1 -PI in BAL fluid, failed to find any differences between smokers or non-smokers in one study (Campbell et al., 1987). Two other reports have failed to detect the presence of inactivated α_1 -PI in serum or plasma from smokers (Lellouch et al., 1985; Walsh et al., 1992).

While the evidence for increased or free HLE activity in the pathogenesis of emphysema is strong (section 1.5.4), the evidence, as described above, for compromised α_1 -PI activity in smoking associated emphysema is inconsistent. The variable results from these studies may be explained by differences in sensitivity and specificity in being able to detect the small amounts of inactivated α_1 -PI present in BAL fluid and serum or plasma from smokers. In comparison, the degree of inactivation of α_1 -PI is much greater and more easily detected in an acute disease,

such as ARDS, than the more subtle inactivation which is proposed to occur in a life-long chronic lung disease, such as smoking associated emphysema. In addition, inactivation of α_1 -PI is likely to occur only in those parts of the lung exposed to tobacco smoke, as suggested by the pattern of emphysema that is seen in the lungs of smokers (Thurlbeck, 1988) and, as proposed elsewhere (Campbell and Campbell, 1988; Weiss and Regiani, 1984), inactivation of α_1 -PI may be localised to the pericellular junction between the inflammatory cell and the extracellular matrix surface of the lung. In conclusion, the possibility that there are only small amounts of inactivated α_1 -PI localised to sites of inflammatory cell degranulation in the emphysematous lung may explain, when combined with methodological differences, the difficulty in detecting the presence of inactivated α_1 -PI in BAL and serum from smokers.

1.5.5.2 ALPHA-2-MACROGLOBULIN (α_2 -M)

 α_2 -M is unique in its ability to inhibit all four classes of proteases (serine, cysteine, aspartate and metallo- proteinases) suspected of having a role in emphysema (Travis, 1988). While α_2 -M appears to be a non-specific protease inhibitor it may have an important regulatory role towards proteases, as suggested by the fact that it binds preferentially and more rapidly to collagenase than the specific tissue inhibitor, TIMPS (Cawston and Mercer, 1986). While α_2 -M function is primarily restricted to the vascular system and interstitial fluid (Tollefsen and Saltvedt, 1980) because of its size (M_r 725,000), it is also found in small amounts in the lung due to diffusion from the vascular system and local synthesis by macrophages (Stockley et al., 1979; White et al., 1981). Thus, the presence of α_2 -M in the lung suggests that it may have a role as part of the lung antiprotease system.

 α_2 -M functions as an antiprotease by unwinding its strands to open a cavity for entrance of the protease and then retwisting the strands to encapsulate the protease (Qazi et al, 1998). However, while having a broad specificity, α_2 -M is not a very effective antiprotease as it does not

completely inhibit most protease activity. For example, when complexed to α_2 -M, proteases typically retain 80-100% of their hydrolytic activity against low molecular weight substrates and 10% or less against large proteins (Barrett, 1981). This has led to the proposal that α_2 -M-HLE complexes may play a role in the development of pulmonary emphysema by allowing the encapsulated HLE limited access to lung tissue (Galdston et al., 1979; Stone et al., 1979). This proposal is supported by the detection of α_2 -M-PPE complexes in BAL from the lungs of hamsters with PPE-induced emphysema (Stone et al., 1982).

Because of its broad specificity as a protease inhibitor, α_2 -M is thought to act as an important backup inhibitor to the more specific inhibitors, such as α_1 -PI and TIMPS. Some support for this role is suggested by the fact that α_2 -M inhibitory function appears to compensate for compromised α_1 -PI function, as suggested by the finding of significantly increased levels of α_2 -M in all types of α_1 -PI deficient individuals (Fryksmark et al., 1983; Brissenden and Cox, 1983). Another report has also described an apparent increased elastase binding capacity of α_2 -M in individuals with COPD and normal α_1 -PI levels (Kilroe-Smith et al, 1989). The above reports would suggest that α_2 -M has a role as a high-affinity broad spectrum proteinase inhibitor but there is little evidence, even from subjects with α_2 -M deficiency (Poller et al, 1989; Kruger, 1993), to support α_2 -M having a major role in the development of emphysema.

1.5.5.3 ALPHA-1-ANTICHYMOTRYPSIN (α_1 -ACH)

 α_1 -ACH is a major plasma acute phase protein that can increase 5-7 fold in concentration after an inflammatory insult (Aronsen et al., 1972) suggesting that it has an important role in controlling certain enzyme(s) associated with inflammation. It is synthesised and secreted by hepatocytes and by alveolar macrophages and is detectable in lung lavage fluid. α_1 -ACH has been shown to inhibit chymotrypsin-like proteases (Beatty et al., 1980; Travis et al., 1978) and mast cell chymases (Travis et al., 1978; Reilly et al., 1982). The most rapid inhibition is seen towards cathepsin G (Beatty et al., 1980; Travis et al., 1978) which suggests that its function is to regulate cathepsin G activity. Although cathepsin G is a relatively poor protease, it can degrade some components of connective tissue and is much more effective in this capacity when it can work synergistically with HLE (Boudier et al., 1991). As both enzymes are coreleased from leukocytes *in vivo*, this is a likely mode of action in an inflammatory response. However, the presence of HLE specifically inactivates α_1 -ACH allowing cathepsin G unrestricted activity (Potempa et al., 1991), which suggests that the regulation of cathepsin G activity may not be the prime function of α_1 -ACH as earlier studies have suggested.

However, a role for α_1 -ACH in defending the lung against proteolytic attack is suggested by reports of an association between α_1 -ACH deficiency and disturbances of lung (Eriksson et al., 1986; Weidinger et al., 1992; Poller et al., 1993). At this stage, the mechanism and the extent of involvement of α_1 -ACH deficiency with the development of COPD is uncertain. Nevertheless, it is possible that a deficiency in α_1 -ACH function may be one of the familial factors that could explain why only some individuals with an α_1 -PI deficiency and only some smokers develop emphysema.

1.5.5.4 SECRETORY LEUKOPROTEASE INHIBITOR

SLPI (secretory leukoproteinase inhibitor; also called antileukoprotease or bronchial mucus proteinase inhibitor) is a low molecular weight boomerang-shaped protein (M_r 11,700) (Hochstrasser et al., 1972) responsible for more than 80% of antiprotease activity in bronchitic sputum (Stockley et al., 1986). SLPI is a reversible inhibitor of trypsin, cathepsin G and chymotrypsin (Schiessler et al., 1978) as well as a fast acting inhibitor of HLE (Boudier and Bieth, 1989). It is also an effective inhibitor of mast cell chymase in the presence of heparin, a glycosaminoglycan which is found as part of the lung tissue surface (Walter, 1996).

Early studies indicated that SLPI appeared to have two reactive sites, but a later study has shown that all enzymes bind to a single site on the inhibitor (Eisenberg et al., 1990). The binding site on SLPI is susceptible to oxidation, with a resultant loss of inhibitory activity against HLE (Kramps et al., 1989). This oxidative inactivation of SLPI is similar to that which occurs with α_1 -PI suggesting a methionine residue at the reactive site as found in α_1 -PI (see previous section 1.5.5.1). SLPI is localised in secretory granules of serous cells in the upper respiratory tract (Kramps et al., 1981), is also found in Clara and goblet cells of the small peripheral airways of the lung (DeWater et al., 1986) and is synthesised by epithelial cells (Abe et al., 1991). SLPI has also been reported to be the major inhibitor that is found in the neutrophil cytosol, where it has been proposed that it may act as a protective screen against proteinases spilling from the azurophilic granules into the cytosol (Sallenave et al, 1997a). Based on the recovery of SLPI and α_1 -PI in lavage samples from different parts of the lung, SLPI has been suggested to be the main inhibitor in the larger airways, whereas α_1 -PI is predominant in the peripheral airways (Kramps et al., 1988).

No inherited deficiency of SLPI or a role in the development in emphysema, similar to that seen for α_1 -PI, has been reported as yet. However some of its functions are similar to that of α_1 -PI. For example, both SLPI and α_1 -PI concentrations are increased in BAL from emphysema patients, presumably to balance the increased elastolytic load in the lungs, although the mechanism is unknown (Willems et al., 1989; Trefz et al., 1992). Like α_1 -PI, SLPI is inactivated by oxidants *in vivo*, as has been demonstrated in a patient with cystic fibrosis (Vogelmeier et al., 1991), although it has not been reported in emphysema. As is the case for α_1 -PI inhibition of induced emphysema in animal studies, SLPI is also able to significantly diminish the development of Escherichia coli lipopolysaccharide - mediated pulmonary emphysema (Rudolphus et al., 1993) and to prevent the development of HLE induced emphysema in hamsters (Rudolphus et al., 1994). What is unique about SLPI is that it is able to inhibit the contact-dependent proteolysis that occurs between HLE and a substrate surface; a property not possessed by any other known endogenous inhibitor. Campbell et al. (1982, 1988) have shown that large inhibitors such as α_1 -PI and α_2 -M are partially excluded from the leukocyte-substrate interface and are thus ineffective as proteolytic inhibitors once the leukocyte is bound to the substrate. However, the low molecular weight inhibitor, SLPI, is able to inhibit HLE after it is bound to elastin (Morrison et al., 1990) and inhibits ongoing proteolysis by leukocytes already bound to substrate (Rice and Weiss, 1990).

In conclusion, SLPI has been shown to be one of the major serine proteinase inhibitors in the lung and the fact that it has been found to be associated with elastic fibres (Willems et al., 1986; Rudolphus et 1994) suggests that this antiprotease may have a role in protecting the lung against elastolytic proteinases.

1.5.5.5 ELAFIN

More recently, another low molecular weight inhibitor, elafin, has been found to be secreted by Clara cell and type II pneumocyte cell lines. It had been found previously only in skin (Sallenave et al., 1993). Elafin has also been shown to be present in expectorated sputum (Sallenave et al., 1993), in BAL fluid from normal subjects and in elevated amounts in BAL fluid from individuals with farmer's lung and lymphocytic alveolitis (Tremblay et al, 1996). Elafin appears to be synthesised as a 12-14,000 M_r precursor and then processed to the secreted 6,000 M_r protein. The importance of elafin as an antiprotease in the lung and determination as to whether it has a role in the development of emphysema awaits further investigation.

1.5.5.6 MONOCYTE/NEUTROPHIL ELASTASE INHIBITOR

Monocyte/neutrophil elastase inhibitor (M/NEI) is a protease inhibitor that has been found in high levels in human neutrophils, monocytes and macrophages (Remold-O'Donnel, 1989). This inhibitor can regulate the activity of the neutrophil proteases: HLE, cathepsin G and proteinase 3. M/NEI is a nonglycosylated single polypeptide of M_r approximately 42,000 with an essential cysteine at the reactive centre of the molecule (Renold-O'Donnel, 1992). Initial studies suggest that it is an oxidation-sensitive molecule that would only be able to inhibit serine proteases in the immediate vicinity of the carrier cells. Its function in the regulation of proteolytic activity from neutrophils, monocytes and macrophages is at present unknown. Further studies may determine whether it has is any role in protecting the lung against elastolytic proteinases.

1.5.5.7 OTHER ENDOGENOUS INHIBITORS

A number of endogenous compounds, other than the known proteases, have been reported to inhibit HLE activity to varying degrees. The most potent of these compounds are the sulphated GAGs, such as heparin, heparan sulphate, chondroitin sulphate and dermatan sulphate (Redini et al., 1988a; Walsh et al., 1991a; Drag and Petersen, 1994). However it has not been determined whether endogenous sulphated GAGs have an anti-elastase role *in vivo*. Although the occurrence and location of sulphated GAGs *in vivo* on cell surfaces and as components of proteoglycans in cell and basement membranes (Höök et al., 1984; Marcum and Rosenberg, 1984) lends some support for this role. The finding that sulphated GAG's, such as heparan sulphate, chondroitin sulphate and dermatan sulphate, are synthesised and secreted by neutrophils (Bartold et al, 1989) would suggest that sulphated GAG's may have a role in regulating the activity of HLE secreted by neutrophils.

Elastin peptides have also been reported to be potent inhibitors of HLE (Bonnaure-Mallet et al., 1995) with the formation of an unusually stable complex between HLE and elastin peptides rich

in the alanine/glycine and hydrophobic domains of elastin (Tyagi and Simon, 1993). Tyagi and Simon (1993) have suggested that the elastin peptides, released during elastinolysis, have a regulatory role towards HLE activity at local sites of proteolysis.

Other compounds in the lung may also be able to influence HLE activity. One such compound is pulmonary surfactant, which is found as a thin layer on the alveolar surface where it may be able to partially protect the alveolar surface against HLE activity. This proposal is suggested by the report that the administration of pulmonary surfactant resulted in a significant inhibition of elastase-induced emphysema in mice (Otto-Verberne et al., 1992). The above examples demonstrate that HLE activity may also be regulated by other components *in vivo* other than the specific anti-proteases.



Figure 1.2: Diagram outlining the protease-antiprotease hypothesis of emphysema.

Note that not all the proteases have elastolyic activity but may have a role, by working in parallel or synergisticly with the major elastases such as HLE, in damaging the connective tissue of the lung. α_1 -PI and SLPI appear to be the only antiproteases that are inactivated by oxidants. Oxidant activity is normally regulated by the presence of antioxidants such as superoxide dismutase, glutathione, and catalase (not shown).

1.6 THE PROTEASE-ANTIPROTEASE HYPOTHESIS: REDRESSING THE IMBALANCE

Despite the fact that the protease-antiprotease hypothesis is only supported by indirect evidence, it has stimulated much study of the proposed mechanisms in the development of emphysema (see Figure 1.2), in the course of which a variety of new methods for preventing emphysema have been tested. Most of the procedures have involved reducing the protease levels or increasing the antiprotease levels within the lungs.

1.6.1 REDUCING THE PROTEASE LEVELS

If the protease-antiprotease hypothesis is accepted as the basis of the pathogenesis of emphysema, one would predict that the development of emphysema would be prevented or reduced by lowering the protease load, particularly of elastolytic enzymes in the lungs of susceptible individuals. One way of achieving this would be to prevent exposure to the agent responsible for the influx of inflammatory cells into the lungs, thereby preventing the release of proteases and oxidants responsible, in this model, for the development of emphysema.

As discussed previously, there is a wide range of indirect evidence from population and animal model studies to support the view that smoking is the major risk factor in the pathogenesis of emphysema. The influence of smoking is such that it overrides every other risk factor, including the only clinically proven genetic factor, α_1 -PI deficiency. Thus smoking appears to be able to alter the balance of the protease-antiproteases in the lung resulting in elastolytic lung damage. One likely mechanism for the protease excess is the release of elastases from macrophages and neutrophils attracted to the cigarette smoke particles together with reduced antiproteases due to inactivation by smoking and phagocyte derived oxidants.

Numerous population studies have shown that mortality and incidence rates increase with increasing exposure to cigarette smoke and that there is an improvement in mortality and

morbidity rates in smokers who quit compared with smokers who continue to smoke. Smokers who are able to stop smoking reduce their rate of decline of FEV₁ from about 70 ml a year to about 30 ml a year (Cohen et al., 1991). Despite quitting, ex-smokers still have an excessive age-adjusted overall mortality when compared to lifetime non-smokers, which is most likely to be due to the fact that the lungs and some other organs have already been permanently damaged (Cohen et al., 1991). However most smokers cannot give up the habit, as it is so highly addictive (U.S. Surgeon General, 1988) that even the most intensive smoking cessation programs have not been very successful. In one program, for example, about 75% of individuals were still smoking one year after completing the program (Ockene et al., 1990). One strategy to reduce the development of COPD in smokers, who cannot give up the habit, has been to promote the use of low tar, low nicotine cigarettes. However, results with this type of cigarette have been disappointing or absent especially in COPD, although there appears to be a reduction in the prevalence of cough and the risk of lung cancer (U.S. Department of Health and Human Services, 1989).

Another strategy for influencing the balance of protease-antiprotease activity in smokers is to prevent neutrophils from degranulating and releasing enzymes and oxidants into the lung. One such drug that has this potential is colchicine, which has been shown to reduce HLE secretion *in vitro*. However, when smokers with COPD were treated with colchicine in a controlled study, there was no change in the HLE load or elastase degradation products (desmosine and elastin peptide levels) in the smokers' BAL (Cohen et al., 1990). A later controlled study from the same group showed that colchicine was partially effective in that it was able to reduce HLE levels by 51.5% in BAL fluid from ex-cigarette smokers with COPD (Cohen et al., 1991). The authors attempt to explain this difference by suggesting that the difference between the two studies is most likely to be due to the fact that colchicine is too mild a drug to overcome the ongoing stimulation of leukocytes in current smokers, but is able to modify degranulation in the less stimulated leukocytes of ex-smokers. However, on the above evidence colchicine is ineffective in smokers. Nevertheless, the use of a drug, with a similar mode of action, that could

significantly alter the protease-antiprotease balance in the lung, holds much promise, particularly where it could be included in cigarettes of smokers who are unable to give up smoking.

1.6.2 INCREASING THE ANTIPROTEASE LEVELS

Most research into the effects on emphysema of altering the protease-antiprotease balance has been directed at supplementing the antiprotease system using either purified or recombinant α_1 -PI, other endogenous inhibitors or one of the many synthetic inhibitors that have been developed. Replacement therapy with α_1 -PI, purified from plasma, has been used in clinical trials in α_1 -PI deficient patients (PiZ) and has been shown to maintain adequate plasma and lung concentrations without inducing appreciable side effects when given either parentally or by inhalation (Gadek et al., 1981(a); Hubbard et al., 1988; Hubbard et al., 1989). This therapy has been used in most of the 1000 PiZ patients enrolled since 1989 in the US National Heart, Lung, and Blood Institute α_1 -PI deficiency registry (Luisetti and Travis, 1996). So far, preliminary reports of replacement therapy show little or no decline in lung function over time, but in most subjects the period of therapy is still short. For example, in a recent study, the authors have concluded from the first year of a program of weekly infusions of α_1 -PI to PiZ individuals that there may be a slowing in the rate of decline in FEV1 after 1 year (Seersholm et al., 1997). To date, the most comprehensive study has monitored 14 patients for periods of from 12 months to four years, where it was concluded that replacement therapy 'may be associated with decreased hospitalisation and possibly with stabilisation of lung function' (Barker et al., 1994). At this stage of the program, it is unproven that replacement therapy with purified α_1 -PI can prevent the development of emphysema in α_1 -PI deficient patients.

To prove definitively that replacement therapy can prevent the development of emphysema in α_1 -PI deficient patients requires the use of large amounts of α_1 -PI in large population studies
over many years. A weakness of this type of study is to obtain sufficient quantities of purified α_1 -PI. An alternate source of α_1 -PI is to use that produced by recombinant technology where it may be possible to synthesise sufficient amounts for large scale replacement studies. However studies have shown that the non-glycosylated form produced by recombinant technology is more rapidly excreted in urine when compared to the glycosylated native form of α_1 -PI (Casalaro et al., 1987; Schnebli, 1991). The importance of the carbohydrate component is suggested by the fact that it constitutes nearly 20% of the molecular mass of native α_1 -PI (Schnebli, 1991). At this stage, the recombinant form of glycosylated α_1 -PI has only been produced in small amounts (Garver et al., 1987) but it is obvious from the preliminary studies that for it to be as effective as the purified form, the recombinant form may require the carbohydrate component of the molecule to be equally as effective.

A number of other recombinant forms of α_1 -PI have been produced in which there are alterations in the P₁ residue, which is thought to be responsible for the specificity of the inhibitor. Two examples are the recombinant α_1 -PI MET³⁵⁸ \rightarrow VAL (Travis et al., 1985) and MET³⁵⁸ \rightarrow LEU (Jallat et al., 1986), both of which are better inhibitors of HLE than the native α_1 -PI as well as being oxidation resistant, unlike the native form. The improved recombinant forms of α_1 -PI offer the potential to be more effective and longer lasting inhibitors of HLE, particularly for use, in the short term in acute conditions such as ARDS. However, the long term administration of oxidation-resistant forms of α_1 -PI for the prevention of the development of emphysema would need to be carefully evaluated so that it did not interfere with the normal physiological role that HLE has in inflammatory reactions, such as chemotaxis, phagocytosis, and acute phase protein synthesis (Travis and Fritz, 1991). In addition, there is also the question as to whether the presence of oxidation-resistant forms of α_1 -PI would interfere in the regulation of the function of the oxidation susceptible native form of α_1 -PI by oxidants released by macrophages and nutrophils. So far none of these modified forms of α_1 -PI has been clinically evaluated. As α_1 -PI is not the only inhibitor found in BAL, it has been proposed that replacement therapy with a combination of α_1 -PI and other inhibitors found in BAL could provide a more balanced and effective form of therapy for the prevention of the development of emphysema. To date, the only inhibitor investigated to any extent for this role has been SLPI. Initial studies have demonstrated that recombinant SLPI can be delivered to the lung successfully as an aerosol, where it has been shown to move from the epithelial surface to the lung interstitium, providing not only a potential increase in the anti-HLE screen but also a potential increase in antioxidant levels by increasing glutathione levels (Gillissen et al, 1993; Stolk et al, 1995; Vogelmeier et al, 1996).

Because of the difficulties in obtaining sufficient amounts of α_1 -PI for large scale clinical replacement studies, many synthetic protease inhibitors have now been evaluated as inhibitors of HLE in vitro (Powers and Harper, 1986; Groutas, 1987). Some of the more promising candidates have also been investigated for their ability to prevent elastase-induced emphysema in laboratory animals. These include:- suc-ala-ala-pro-val-chloromethylketone in hamsters (Stone et al., 1981; Fletcher et al., 1990); a peptide boronic inhibitor in hamsters (Soskel et al., 1986; Stone et al.; 1990); a peptide aldehyde reversible inhibitor in hamsters (Kennedy et al., 1987); specific β-lactam inhibitors in hamsters (Bonney et al., 1989; Fletcher et al., 1990); SR 26831, a tetrahydrothieno pyridine derivative in rats and rabbits (Herbert et al., 1991); ICI 186756, a peptide aldehyde inhibitor in hamsters (Williams et al., 1991); ICI 200355, a peptide trifluoromethyl ketone inhibitor in hamsters (Williams et al., 1991); ONO-5046, a low molecular weight inhibitor in guinea pigs (Sakamaki et al., 1996). In general, all the above inhibitors are very effective in preventing the development of HLE-induced emphysema in laboratory animals but because of the extensive evaluation studies and clinical trials that are required to detect possible clinical side effects none has, as yet, been evaluated for use in humans.

As well as synthetic inhibitors, a number of natural compounds, such as polypeptidic antielastases and sulphated GAG's and their derivatives have been evaluated as anti-elastase compounds. The more promising candidates have also been investigated for their ability to prevent elastase-induced emphysema in laboratory animals. These include:- CY222, a low molecular weight heparin fraction of mean M_r 4,500 in hamsters and mice (Redini et al., 1988b; Lafuma et al., 1991); Arteparon, an oversulphated derivative of chondroitin sulphate in hamsters (Rao et al., 1990); sheep lung surfactant in mice (Otto-Verberne et al., 1992); and an elastin derived peptide preparation of mean M_r 57,000 in rats (Bonnaure-Mallet et al., 1995). None of these preparations has been evaluated in clinical studies. However, they are attractive as therapeutic agents because, being derived from endogenous compounds, they are likely to produce less clinical side effects and so more rapidly gain approval for clinical use.

In general, inhibitors of HLE are effective in preventing or reducing the HLE-induced acute tissue damage that initially occurs prior to the development of the emphysematous lesions that are seen in experimental animals. The evidence that HLE inhibitors are as effective in preventing the more chronic lesions found in human emphysema is at present not available, primarily because of the difficulty in meeting ethical requirements for the use of most of these compounds in human studies.

Emphysema develops slowly over a number of years, even in α_1 -PI deficient patients, so that a long time period and many patients will be needed in trials to assess the benefits and side effects of any type of replacement therapy. A native α_1 -PI preparation (Prolastin[®]), which is already being used in clinical studies, and recombinant α_1 -PI and SLPI, and perhaps some derivatives of other endogenous compounds (eg. heparin derivatives) appear likely to offer the highest degree of safety for long term clinical use. Of these agents, the low molecular weight, synthetic inhibitors, will require more development initially, but seem likely to be cheaper and technically easier to apply by an aerosol than are protein preparations. Some of the low molecular weight, synthetic inhibitors may even be active when given orally. The impetus to develop synthetic inhibitors is also driven by the fact that they are likely to have wider applications for the treatment of other conditions in which HLE has been implicated in the pathogenesis. Examples range from acute conditions such as septic shock and ARDS, to chronic conditions such as cystic fibrosis and rheumatoid arthritis, and include infectious diseases like AIDS.

Gene therapy is the newest approach to the treatment of lung disease. The lung is an attractive target organ for gene therapy because of its accessibility through the airways or the vasculature. Therapeutic gene delivery has largely focused on introducing corrective genes in lung diseases arising from single gene defects, such as cystic fibrosis. More recently interest has been centred on gene therapy as a tool in the treatment of more complex pulmonary conditions, such as emphysema (Sallenave et al., 1997b). It has been proposed that modification of critical components of the inflammatory process, so as to increase the expression of anti-elastase genes, could circumvent elastase mediated lung damage in emphysema. In the case of α_1 -PI deficiency, one approach would be to replace the PiZ gene with the PiM gene so as to restore normal α_1 -PI levels in the lung of deficient individuals and to prevent the development of the inherited deficiency form of emphysema.

1.6 SUMMARY

From this review of the literature, it can be seen that the pathogenesis of emphysema is complex and that the critical mechanisms still remain to be elucidated. The protease-antiprotease hypothesis has been very useful in that it has provided a framework for many directions in emphysema research, the results of which, in general, are held to support the hypothesis. However, it is important to note that almost all of the evidence that supports this theory is still indirect. In the following paragraphs the major evidence supporting the theory is noted with comment on what further research is required to prove, modify or disprove the theory.

Evidence for HLE being the major protease part of the hypothesis

HLE has been shown in animal studies, to induce lesions typical of emphysema, including increased airspace size and lung volumes. These have been induced in the lungs of the animal by sequestering leukocytes into the lung or more directly by introducing HLE, or other elastases, intratracheally.

HLE has been shown to have a broad substrate specificity, so that it is capable of degrading a wide range of extracellular matrix proteins, including elastin. It can potentially be secreted in millimolar amounts, in response to various inflammatory stimuli, from the azurophil granules of leukocytes and also to some extent from monocytes and macrophages. No other elastolytic enzyme has been found in the lung in such large amounts, nor does any other elastase have such broad substrate specificity. There appears to be sufficient evidence to conclude that this enzyme, on its own, could account for the bulk of the tissue damage observed histologically in emphysema.

The finding of elevated elastin degradation products (such as elastin derived peptides in plasma, urine and BAL and in recent studies, desmosine in urine) supports the presence of increased elastase activity, most likely HLE, in the lungs of emphysema patients. More specific evidence for a role of HLE in the pathogenesis of emphysema can be inferred from two reports that HLE has not only been found to accumulate on interstitial elastin in areas of the human lung affected by emphysema, but also that the amount of HLE present strongly correlated with the local severity of emphysema. Markedly elevated plasma elastase specific fibrinopeptides, which are only derived from HLE activity on fibrinogen, have been reported in individuals with α_1 -PI deficiency where the fibrinopeptide level correlates with the degree of α_1 -PI deficiency. As

those individuals with α_1 -PI deficiency have a greater risk of developing emphysema, the presence of increased HLE activity supports a role for HLE in the pathogenesis of the inherited form of emphysema. Thus, the overall evidence strongly supports a role for HLE in the pathogenesis of emphysema, if not the primary causal agent. Nevertheless, further research into the development of, and use of more specific markers of lung destruction would be desirable, as it would allow the identification of those individuals with a high rate of lung degradation and the monitoring of the effectiveness of treatment with anti-HLE inhibitor therapy. These studies would also further clarify the extent of the involvement of HLE in the pathogenesis of emphysema.

The variable pathogenesis and different forms of emphysema suggests that there may be other factors causing an imbalance in the protease - antiprotease balance allowing increased HLE activity in the lung. Such factors could include a local variation in the lungs of concentrations of other elastolytic enzymes (eg. metalloproteinases, cathepsin G, proteinase 3), proteinase inhibitors (eg. α_1 -PI, SLPI, elafin) and oxidants (eg. neutrophil oxidants, external oxidants such as tobacco smoke and other occupational and environmental agents). However, at this stage, this is only speculation as there is no evidence in the literature to support the possible role of other factors in human studies.

Despite an abundance of information on the activity of HLE in the lung, definitive proof will require prospective studies with specific HLE inhibitors or control of HLE expression at the gene level. The elimination of free HLE from the lungs of patients with the inherited PiZ form of α_1 -PI deficiency and in smokers with augmentation therapy (using α_1 -PI, SLPI and the new range of specific HLE inhibitors) should give more definitive proof that HLE is the primary causative agent in the pathogenesis of human emphysema and whether the protease - antiprotease hypothesis is still valid.

Evidence for α_1 -PI being the major antiprotease part of the hypothesis

 α_1 -PI inhibits HLE at least 10 times more rapidly than any other protease tested (Beatty et al., 1980) and therefore it is claimed that the primary function of α_1 -PI is to inhibit HLE. As discussed previously, most of the evidence in the literature supports the view that HLE is the prime candidate responsible for the degradation of lung tissue in emphysema. If the evidence is accepted that the primary inhibitory role of α_1 -PI is towards HLE, together with the evidence that reduced α_1 -PI levels are linked with the development of the inherited form of emphysema, then α_1 -PI must be strongly considered to be the major antiprotease in the protease - antiprotease hypothesis.

If α_1 -PI is the critical protease in the pathogenesis of emphysema, one would expect it to be the major anti-elastase in the lower respiratory tract where the emphysematous lesions occur. This is indeed the case where a number of studies have reported that α_1 -PI constitutes the majority of the anti-elastase activity in the normal lower respiratory tract, unlike many of the other major protease inhibitors, which are only present in minor amounts (eg. SLPI, α_2 -M).

While adequate levels of α_1 -PI have been shown to prevent emphysema, the absence of α_1 -PI does not necessarily lead to the development of emphysema. This is found in individuals with the homozygous form of α_1 -PI deficiency (PI Z), in which a recent study has estimated that only 10-15% of such individuals develop emphysema (Knight et al., 1997). This suggests that factors in addition to α_1 -PI deficiency play a role in the development of emphysema. One of the other factors could be a deficiency in one of a number of other HLE inhibitors, such as SLPI, elafin, α_2 -M, α_1 -ACH or M/NEI. However, only a deficiency in either α_1 -ACH or α_2 -M has been reported to be associated with the development of emphysema in some individuals, but neither have been linked with α_1 -PI deficiency. While there is no reported association between an inherited deficiency of α_1 -PI and the other known HLE inhibitors, there is more of a

likelihood of association between α_1 -PI and certain inhibitors, which are subject to oxidative inactivation in the more common smoking induced form of emphysema. For example, SLPI, elafin and M/NEI are inactivated by oxidants similar to that reported for α_1 -PI. It is possible that variable inactivation of these other inhibitors by smoking and/or oxygen free radicals together with oxidative inactivation that has been reported for α_1 -PI could explain the different susceptibilities of individual smokers to develop emphysema. In conclusion, the contribution, if any, of these other inhibitors and their relative oxidative status in smokers to the development of emphysema remains to be determined.

Another factor in the variability of the development of emphysema between individuals could be explained by differences in the susceptibility of the lung matrix to HLE activity due to subtle genetic differences in the composition of the lung matrix. For example, variations in the sulphated GAGs of the extracellular matrix of the lung could conceivably contribute to variable HLE digestion of the lung as has been reported *in vitro* where sulphated GAGs have shown variable inhibitory activity towards HLE. Similarly, a change of sequence or structure of some of the proteins in the lung extracellular matrix, such as has been shown with the different isoforms of elastin, collagen and fibronectin (Boyd et al, 1993), may also contribute to an alteration in the susceptibility of the lung matrix to HLE activity. At present the extent and the significance of variations in the composition of the extracellular matrix of the lung and the effect that this variation has on HLE digestion of the lung is not known.

Role of cigarette smoking in emphysema

Although smoking is the major risk factor for the development of emphysema, with a direct relationship between the amount of cigarettes smoked and respiratory damage, there is a wide variation in the susceptibility to develop progressive emphysema, as demonstrated by the fact that only 10-20% of tobacco smokers develop clinically evident emphysema. One genetic factor

that has been reported to explain some of this variability is the finding that an α_1 -PI deficiency (PI Z) in a smoker results in the much earlier development of emphysema (approximately 20 years), which probably reflects the severely limited antiprotease reserves (less than 20% of normal) in the smoker's lungs (Tetley, 1993). However, no other host or familial factors that make individual smokers more susceptible to emphysema have, at present, been identified. Until these other variables are identified, it is likely that mechanisms involved in the development of emphysema will not be fully understood.

Understanding the mechanism of emphysema generated by smoking has been hampered by the failure to cause true emphysema lesions in animals by prolonged exposure to cigarette smoke. While the obvious species difference is the major factor, there is also the problem that most of these studies have been of a short duration in comparison to the lifetime habit in humans. Another factor is that, as in human studies, there is likely to be variable susceptibility of individual animals to the effects of tobacco smoke which is exacerbated by the small number of animals used in some of the reported studies. In human studies, the most likely hypothesis to explain the development of smoking induced emphysema is that increased numbers of neutrophils are attracted into the lungs by the tobacco smoke and tar. The neutrophils are activated at the site of smoke particles on the lung surface, releasing HLE and oxidants, which combined with the oxidants in tobacco smoke, inactivate α_1 -PI and possibly other inhibitors of HLE. The result is an imbalance between HLE and its inhibitors localised at the site of the smoke particles on the lung surface resulting in conditions favourable for HLE activity. However, there are large gaps in the available evidence to support this hypothesis. In particular, the evidence for the inactivation of α_1 -PI by tobacco smoke and leukocyte generated oxidants and free radicals in the lungs of smokers is inconclusive and awaits further research.

Conclusion

Despite the weakness in a lot of the available evidence, together with the complexity of the problem and, in particular, the difficulty in getting data from human studies, the proteaseantiprotease hypothesis still provides a useful framework for the investigation of the pathogenesis of emphysema. This is despite the fact that the identities of the candidate protease(s) and antiprotease(s) are still unclear. Considerable evidence implicates HLE as the main protease candidate in the pathogenesis of injury to elastic fibres in the lung parenchyma that leads to the subsequent development of pulmonary emphysema. Other elastolytic enzymes have been shown be released from leukocytes (eg. proteinase 3, cathepsin G, metalloproteinases) or secreted by monocytes (leukocyte type enzymes) and alveolar macrophages (metalloproteinases, ingested leukocyte enzymes) and possibly by other resident cells in the lung. It is possible that they may play a role in the pathogenesis of emphysema. After examining the literature, I consider that the above elastolytic enzymes play a secondary role to HLE as the primary elastolytic enzyme in the development of emphysema. For example the different subclasses of emphysema, such as centrilobular, panacinar, etc., is due to HLE activity but that the relative protease concentration imbalance in that part of the lung may be initially contributed to by other elastolytic enzyme (eg. Cathepsin G, which works synergistically with HLE). I consider that HLE is still the main protease part of the proteaseantiprotease hypothesis and for this reason a study of its activity forms the major part of this thesis.

The increased incidence of emphysema in individuals with an inherited α_1 -PI deficiency (e.g. PI Z) can be partially explained by an inadequate level of functional α_1 -PI, as the main antiprotease, although the fact that emphysema does not occur in all individuals suggests that other, as yet unexplained factors, are involved. However, in the more common smoking induced form of emphysema there is even less evidence to explain how the disease arises. It is still unclear as to whether the disease arises from either the smoke components and the

inflammatory cell products inactivating α_1 -PI and other inhibitors (eg. SLPI) or simply from the increased elastolytic load on the lung overwhelming the antiproteases or a combination of both.

From the above literature review it is concluded that HLE is the best candidate for the protease and α_1 -PI is the most favoured or key antiprotease in the protease-antiprotease hypothesis. For this reason, the primary aim of the work in this thesis was to examine *in vitro* the proteaseantiprotease hypothesis with HLE as the main protease component and α_1 -PI as the main antiprotease component. Initial studies examined the functional activity of α_1 -PI in smokers and in patients with emphysema so as to detect the presence of inactivated α_1 -PI. If the protease-antiprotease theory holds true, a reduction in antiprotease levels in smokers and patients with emphysema, as a result of inactivation of α_1 -PI, should lead to a protease imbalance which should be detected by sensitive assays developed as part of this thesis project.

During the course of measuring enzymatic activity of HLE, it was noted that a number of components in the assay systems influenced the outcome of the assays. A detailed study using the HLE assay has shown that these components included proteins (eg. elastin peptides), surfactants (eg. pulmonary surfactant), GAGs (eg. heparin), and a number of endogenous and exogenous inhibitors. As HLE is known to have a broad substrate specificity that allows it to bind and interact with numerous structural protein components and metabolites in the lung, it was not surprising that a number of lung components have an inhibitory effect towards HLE which supports the proposal that components, in addition to the defined lung antiproteases, may play a significant regulatory function on HLE activity in the lung and could partly explain the variable development of emphysema in α_1 -PI deficient individuals and smokers. In particular, it has been shown that sulphated GAGs, particularly heparin, are potent inhibitors of HLE. A detailed study of the binding of, and modifying effects of heparin on, HLE

has been completed, resulting in the finding that there are high-affinity binding sites for HLE along the heparin chain. This finding raises the possibility that GAGs, such as heparin, may have an important regulatory influence on HLE activity in the lungs. It is postulated that they function as a component of the antiprotease screen on the lung surface.

Hypothesis and Aims of this study.

- 1. To demonstrate that the protease antiprotease hypothesis holds true under conditions in which HLE is the principal protease and α_1 -PI is the principal antiprotease.
- 2. To determine whether other specific and non-specific inhibitors have a functional role in the protease antiprotease hypothesis.
- 3. To examine the relative specificity and effectiveness of a variety of other specific and nonspecific inhibitors towards HLE.
- 4. To apply these data to the proposal that HLE is the major candidate in the protease antiprotease hypothesis.

CHAPTER TWO

MATERIALS AND METHODS

2.1 INTRODUCTION

The studies reported in the following chapters describe a number of the aspects of the interaction of HLE with a range of endogenous and synthetic inhibitors so as to further investigate the central role of this enzyme in the elastase - anti-elastase model of a number of inflammatory diseases such as pulmonary emphysema. A large part of this study relied on assays for HLE activity under a variety of assay conditions. Variations of assays for this enzyme together with other methods and the materials used are described in this chapter.

All chemicals and biochemicals described throughout this thesis were of analytical grade or better if available.

2.2 MATERIALS

2.2.1 ENZYMES

HLE (EC 3.4.21.37; 875 units/mg of protein) was from Elastin Products Co., USA. HLE was demonstrated to be at least 90% active using previously published kinetic constants with Suc-A-A-NA as substrate (Nakajima et al., 1979) as described in Section 4.2.

Other human leukocyte enzymes used were cathepsin G (EC3.4.21.20; Elastin Products) and myeloperoxidase (EC 1.11.1.7; Calbiochem Corp, CA. USA).

Pancreatic enzymes used were PPE (EC 3.4.21.36; Calbiochem), trypsin (EC 3.4.21.4; porcine type IX; Sigma Chemical., USA) and chymotrypsin (EC 3.4.21.1; bovine type 11; Sigma).

Heparin lyase I (heparinase, EC 4.2.2.7 and heparin lyase II (heparitinase II, no assigned EC number) were from Seikagaku Kogyo, Tokyo, Japan.

2.2.2 NON-ENZYME PROTEINS AND PEPTIDES

 α_1 -PI (Cat# A9024)

 α_2 -M

α -Benzoyl-DL-arginine-p-nitroanilide (trypsin substrate)

Aprotinin (Trasylol, 10,000 KIU/mL)

Elastin peptides (a -elastin) preparation

Human albumin

Human antithrombin III

Human lung elastin

Lung surfactant protein

MeO-Suc-A-A-P-V-NA (elastase substrate) Sigma. n-Acetyl-L-alanyl-L-alanyl-å-azanorvaline p-nitrophenyl ester n-Succinyl-(L-alanine)2-proline-L-phenylalanine-p-nitroanilide Sigma. (chymotrypsin and cathepsin G substrate) Suc-A-A-NA (elastase substrate)

VPGVG & VGVAPG (custom synthesised)

Sigma, USA

Sigma.

Sigma.

Bayer AG, Germany.

"In house" preparation. (Section 2.3.6)

Behringwerke AG, Germany.

Calbiochem, USA

Elastin Products, USA

Gift from Dr. J. Powers Flinders University, Australia.

Auspep, Australia.

Calbiochem.

Auspep.

2.2.3 **GLYCOSAMINOGLYCANS**

| Chondroitin sulphate A (bovine trachea) | Sigma. |
|--|----------------------------|
| Chondroitin sulphate B (dermatan sulphate, pig skin) | Seikagaku Kogyo, Japan. |
| Chondroitin sulphate C (shark cartilage) | Seikagaku Kogyo. |
| Heparan sulphate (bovine kidney) | Sigma. |
| Heparan sulphate (bovine intestinal) | Sigma |

Heparan sulphate (fast form, bovine intestinal)

Heparin (porcine)

Heparin (porcine)

Heparin (ovine)

Heparin (bovine, Cat# H-5765)

Heparin (source not stated)

Heparin (OP 381/2, Mr 850-1000)

Heparin (OP 381/1, M_r 2000)

Heparin (OP 2123, Mr 4500)

Heparin (Kabi 2165, *M*_r 4000-6000)

Heparin tetra-, hexa-, and octa-saccharide gel marker

[³H]-heparin & [³H]-heparan sulphate

Hyaluronic acid (rooster combs)

2.2.4 OTHER REAGENTS

Alcian Blue Antiserum to human α_1 -PI (Cat. No. 003-11) USA. 2, 2'-Azinobis(3-ethylbenzthiazoline)-6-sulphonic acid **Biogel P2** Brij 35 (polyoxyethylene monolauryl ethers) Sigma **Bromphenol Blue** Sigma. Chlorophenol Red

Sigma

David Bull Laboratories, Australia.

Fisons, U.K.

David Bull Laboratories.

Sigma.

Commonwealth Serum Laboratories, Australia.

Opocrin S.P.A., Italy

Opocrin S.P.A.

Opocrin S.P.A.

Kabi Vitrum AB, Sweden

Gift from Dr. Turnbull, University of Manchester, U.K.

"In house" preparation. (Section 2.3.3)

Pharmacia, AB, Sweden.

Sigma.

Atlantic Antibodies.

Sigma.

Bio-Rad, USA

Aldrich, USA.

| CNBr-activated Sepharose 4B | Pharmacia. |
|--|--|
| Complete and incomplete Freund's adjuvant. | Commonwealth Serum Labs |
| D-Glucuronic acid lactone | Sigma |
| 1,9-Dimethylmethylene blue | Koch-Light, U.K. |
| Dimethylsulphoxide | Sigma. |
| EAH-Sepharose 4B & ECH-Sepharose 4B | Pharmacia. |
| 1-Ethyl-3-(3-dimethylaminopropyl) carbodiimide hydrochloride | Calbiochem. |
| ICI 186756 (low molecular weight aldehyde inhibitor of HLE) | ICI Americas, Wilmington,USA. |
| ICI 200355 (a peptide trifluoromethyl ketone inhibitor of HLE) | ICI Americas |
| Lung surfactant | "In house" preparation. (Section 2.3.7). |
| Meta-hydroxydiphenyl | K & K Laboratories, USA |
| NaB ³ H ₄ | Amersham, Australia. |
| Peroxidase-conjugated swine antibody to rabbit immunoglobulin | Dako A/S,Glostrup, Denmark. |
| Phenylmethanesulphonyl fluoride | Sigma. |
| Protein Standard Plasma (Human, Cat. # OTF1-03, stated α_1 -PI value 4.2 g/l) | Calbiochem. |
| Rabbit antiserum to elastin peptides (α -elastin) preparation | "In house" preparation. (Section 2.3.8) |
| Sodium dodecyl sulphate (SDS) | BDH Chemical, Poole, U.K. |
| Triton X-100 (octylphenol-polyethyleneglycolether-formaldehyde polymer) | BDH |
| Tween 80 (polyoxyethylene sorbitan mono-oleate) | BDH |

2.3 METHODS

2.3.1 BRONCHOALVEOLAR LAVAGE

BAL was performed as described previously (Bell et al., 1981; Walsh et al., 1992). Briefly, subjects were premedicated with atropine 0.6 mg intramuscularly. A 0.5% (w/v) solution of lignocaine was used for topical anaesthesia of the airways. An Olympus IT bronchoscope (Olympus, Japan) was wedged into a subsegment of the right middle lobe. Three 50 mL aliquots of sterile 140 mM physiological saline were instilled and recovered by gentle aspiration into separate containers. A portion of the second recovered specimen was used for differential and total cell counts. Aliquots 2 and 3 were then pooled and centrifuged at 2000 x g for 10 min to remove cells and debris. Half the supernatant was concentrated to one-fifth of the original volume for normal BAL and one-fifth for ARDS BAL (see Assay Sample Handling, Section 2.4.1.4) in a Minicon B15 concentrator (Amicon Corp., USA) and frozen with the unconcentrated supernatant and a matched serum (collected within 20 min after the lavage) at - 70 °C until analysed.

2.3.2 PREPARATION OF HUMAN LEUKOCYTE ELASTASE-SEPHAROSE GELS

HLE-Sepharose gels were prepared by adding 2 mg of HLE to a 3 ml suspension of EAH-Sepharose 4B or ECH-Sepharose 4B gel with 1-ethyl-3-(3-dimethylaminopropyl) carbodiimide hydrochloride as condensation agent according to a standard procedure provided by Pharmacia. All solutions were adjusted to pH 4.5 prior to mixing. The mixture was mixed end-over-end for 24h at room temperature. Approximately 50 - 70% of HLE was estimated to be bound to the gels (approx. range of 0.3 to 0.4 mg of HLE / ml of gel) by estimation of residual HLE activity using the assay described in Section 2.4.2.2.

2.3.3 PREPARATION OF [³H]-HEPARIN AND [³H]-HEPARAN SULPHATE.

[³H]-heparin and [³H]-heparan sulphate were prepared by reduction of the reducing-end anhydromannose of heparin (porcine, David Bull) or heparan sulphate (bovine kidney, Sigma) chains by adding 0.1 ml of NaB³H₄ (10 mCi) in 0.1 M NaOH to 4 mg of Heparin (porcine, David Bull) in 0.9 ml of 0.05 M sodium phosphate buffer (pH 7.4) and adjusting the pH to 7.4 with 0.1 M HCl. The solution was allowed to stand for 2 hr. in ice and then 18 hr. at 4°C. Excess NaB³H₄ was destroyed by acidification in a fume hood followed by dialysis against the phosphate buffer. Specific activity of [³H]-heparin and [³H]-heparan sulphate was typically in the range of $4 - 6 \ge 10^6$ cpm / mg. The specific activity of the end-labelled heparin was typical for the procedure used in this study, although higher activity could have been achieved with changes to the procedure. Nevertheless the activity was satisfactory for this study. CPM was used as the units throughout this thesis as all values were comparisons between data from the same β -counter using the same assay conditions and instrument settings

2.3.4 PREPARATION OF ANTITHROMBIN – SEPHAROSE.

Human antithrombin III (2 mg) was coupled via amino groups to approximately 5 ml of CNBractivated Sepharose 4B with the recommended procedure provided by Pharmacia with the activated Sepharose except that the reaction was performed in the presence of excess amounts of acetylated heparin as described by Hook et al (1976). The acetylation of any free amino groups present on heparin should prevent any coupling of heparin to the Sepharose beads. Acetylated heparin was prepared by treating heparin with acetic anhydride (Danishefsky and Steiner, 1965). Excess acetylated heparin was added to the antithrombin III to saturate the heparin binding site on the antithrombin molecules and thus prevent this site from binding to the Sepharose beads.

2.3.5 PREPARATION OF 12-22 % ACRYLAMIDE GELS.

Discontinuous PAGE gels with a linear acrylamide and cross-linker gradient (12-22% total acrylamide, pH 8.3) were prepared as previously described (Rice et al., 1987) but with a number of modifications which are described as follows. The gels (20x40x0.15 cm) were prepared by adding 60 mL of 12% gel into the reservoir and 60 ml of 22% gel into the mixing chamber of the linear gradient maker (Pharmacia). The acrylamide solutions were then pumped at 10 ml/min with a peristaltic pump (model CH4A-302, Amicon, USA) into the top of the gel casting unit (essentially two offset glass plates and spacers sealed with tape and held with clamps; Owl Scientific, U.S.A) forming a linear gradient from the bottom to the top. The top of the gel was layered with approximately 0.5 ml of distilled water, sealed and allowed to stand overnight before use.

2.3.6 ELASTIN PEPTIDES (α-ELASTIN) PREPARATION.

The soluble elastin peptides (α -elastin, approx. 70,000 molecular weight as determined by SDS PAGE) were prepared from insoluble human lung elastin by hydrolytic cleavage with 0.25 M oxalic acid for 1 h at 100 °C according to the method of Partridge et al (1985). This hydrolysis of the elastin sample was repeated five times. The protein concentration of the dialysed soluble elastin peptides was determined by the Folin-Lowry method (Lowry et al., 1952) using human albumin as standard.

2.3.7 ISOLATION OF PULMONARY SURFACTANT.

Surfactant was isolated from washings of sheep lung according to the procedure described by Hawgood et al (1985). Briefly, lungs were lavaged with 5 mM Tris/HCl, 100 mM NaCl, pH 7.4. The lavage was centrifuged at low speed (150 g) to remove cellular material and then at high speed (20,000 g) to pellet the surfactant which was then washed and stored at -70° C.

All procedures, including the lavage, were performed at 4° C. Phospholipid content was calculated according to the method used by Bartlett (1959).

2.3.8 RABBIT ANTISERUM TO ELASTIN PEPTIDES (α -ELASTIN) PREPARATION.

New Zealand white rabbits were purchased from the South Australian Government Department of Agriculture (Gilles Plains, South Australia). Antisera of high titre that was successfully used in the ELISA for elastin peptides (see Section 2.4.4) were raised in two rabbits as reported previously (Dillon et al., 1992) and is described briefly as follows. The primary inoculation was administered subcutaneously with multisite injections. Each rabbit received 0.5 mg of an elastin peptides (α -elastin) preparation (see Section 2.3.6) in a total volume of 0.5 ml of saline emulsified in an equal volume of complete Freund's adjuvant. Subsequent immunisations at weekly intervals were given as 0.25 mg of elastin peptides in 0.5 ml of saline with an equal volume of incomplete Freund's adjuvant for at least 3 months. When the antiserum was of sufficient titre, the rabbits were bled by venipuncture and the antiserum was stored at 4°C with 0.02 % sodium azide.

2.3.9 AFFINITY CHROMATOGRAPHY OF HEPARIN ON HUMAN LEUKOCYTE ELASTASE-SEPHAROSE

Heparin (4 mg; porcine, David Bull) with [3 H]-heparin (approx. 3 x 10⁶ cpm, see Section 2.3.3) as tracer in a volume of 0.75 ml was loaded onto a column of either HLE-EAH- or HLE-ECH-Sepharose gel (see Section 2.3.2) in 50 mM/l Tris/HCl, 50 mM/l NaCl, 0.01 % Brij 35, 0.02% NaN₃, pH 7.5 buffer and washed with 10 ml of the same buffer with a flow rate of 0.3 ml/min. Bound heparin fractions were eluted with a 70 ml linear 0.05 – 2 M/l NaCl gradient in the same buffer. Fractions of 1 ml were collected and radioactivity in each was determined. Fractions containing bound heparin peaks were pooled, dialysed (Spectra/por dialysis tubing,

MWCO:3500, Spectrum, USA) against distilled water, lyophilised and reconstituted with distilled water to original column load volume. Heparan sulphate (bovine kidney, Sigma) was also fractionated on HLE-Sepharose by essentially the same procedure.

2.3.10 AFFINITY CHROMATOGRAPHY OF HEPARIN ON ANTITHROMBIN-SEPHAROSE.

The affinity gel was prepared (see Section 2.3.4) and heparin was separated on the gel into high and low affinity fractions following the principles of a previously described method (Hook et al., 1976). Affinity chromatography was performed with 100 μ g of heparin with [³H]-heparin as tracer with approximately 70% (low affinity fraction) not being adsorbed in the presence of 150 mM NaCl. The remaining bound heparin (high affinity fraction) was eluted with 3M NaCl. The low affinity fraction was recycled on the column to remove any remaining high affinity heparin. The high and low affinity pools were then dialysed (Spectra/por, MWCO:3500) against distilled water, lyophilised and reconstituted with distilled water to original column load volume.

2.3.11 DIGESTION OF HEPARIN WITH HEPARIN LYASES.

2.3.11.1 HEPARIN BOUND TO HUMAN LEUKOCYTE ELASTASE-SEPHAROSE.

Heparin (1 mg; porcine, David Bull) with [³H]-heparin as tracer (approx. 0.7 x 10⁶ cpm) was bound to HLE-EAH-Sepharose in 5 mM phosphate, 100 mM NaCl, pH 7.15 buffer and then incubated with heparin lyases (Seikagaku Kogyo) for 30 hours at 30^o C. Heparin lyase I (heparinase, EC 4.2.2.7); 5 miu and heparin lyase II (heparitinase II, no assigned EC number); 5 miu were added at the start of the incubation with further two lots of 5 miu of heparin lyase I added at approximately 10 and 20 hours because of the instability of this enzyme (see Lohse and Linhardt, 1992). Digested heparin and heparinases are then eluted off the gel with the same buffer. The digestion procedure was then repeated to ensure complete digestion of unbound or loosely bound heparin chains. The remaining tightly bound fraction was then eluted off the gel with 2 M NaCl, desalted on a 360 x 20 cm Biogel P2 column (Bio-Rad) and concentrated by lyophilisation. All fractions were boiled for 1 min at 100°C to inactivate any residual enzyme. Unbound heparin was also digested as a control.

2.3.11.2 HEPARIN BOUND TO HUMAN LEUKOCYTE ELASTASE

The digestion procedure is as described above in 2.3.11.1 except that free HLE was used in place of HLE-EAH-Sepharose and that the digestion procedure was not repeated. At the end of the 30 hr. digestion period, the heparin/HLE/heparinase mixture was dialysed against 50 mM Tris/HCl, 100 mM NaCl, pH 8.0 using Spectrapor dialysis tubing (MWCO:3500, Spectrum) to remove digested and unbound heparin fragments. To isolate the remaining bound heparin fragments from HLE and the heparinases, an apoprotin-Sepharose affinity column was used in a procedure reported by Martodam et al., 1979. The apoprotin-Sepharose was prepared by coupling aprotinin (20 mg) via amino groups to approximately 10 ml of CNBr-activated Sepharose 4B with the recommended procedure provided by Pharmacia. Prior to chromatography, NaCl was added to the dialysed heparin/HLE/heparinase mixture to bring the concentration to 1M so as to disassociate the undigested heparin from HLE. The mixture was added to the affinity column and then washed with 1M NaCl in the above Tris buffer to elute the heparin fractions which were desalted on a 360 x 20 cm Biogel P2 column (Bio-Rad) and concentrated by lyophilisation. The HLE was eluted from the apoprotin-Sepharose column with 50 mM sodium acetate, 1M NaCl, pH 8.0.

2.3.12 ELECTROPHORETIC SEPARATION OF HEPARIN FRACTIONS WITH 12-22 % POLYACRYLAMIDE GEL ELECTROPHORESIS

Electrophoresis was performed with two resolving gel buffers as previously described (Rice et al., 1987) but with a number of modifications described, as follows. A comb of 20 teeth of 20 x 5 x 1.5 mm was used as a well former in the stacking gel which allowed 40 μ l samples added

together with 40 μ l of 50% (w/v) sucrose. An oligosaccharide marker consisting of a tetrasaccharide, a hexasaccharide and an octasaccharide fraction prepared from porcine heparin was used to estimate the size of the heparin fragments. Electrophoresis was run on an adjustable vertical electrophoresis system (Owl Scientific) at 600 V in a 4 °C room to provide cooling for 16 hours or until the gel markers Bromphenol Blue (1 mg/ml) and Chlorophenol Red (1 mg /ml) had moved approximately 35 cm and 38 cm respectively into the resolving gel. The gel was removed from the plate and stained and fixed in 0.5% (w/v) Alcian Blue in 2% (v/v) acetic acid for 30 min. followed by destaining with several washes of 5% acetic acid. The bands were made visible with silver staining using a method described previously (Al-Hakim and Linhardt, 1991).

2.3.13 HIGH PRESSURE LIQUID CHROMATOGRAPHY ESTIMATION OF THE SIZE OF ELASTIN PEPTIDES IN THE PRESENCE OF HEPARIN.

Human lung elastin (10 mg) in 1 ml of 50 mM Tris/HCI, 140 mM NaCI and 0.02% sodium azide pH 7.8 containing 100 μ g of HLE was incubated at 37°C for 60 h. An identical preparation, but with 8 mg of heparin added, was also used. HLE activity was stopped with 50 μ l of 5 mg of phenylmethanesulphonyl fluoride per ml of dimethylsulphoxide. Samples were centrifuged and the protein concentration was determined in the supernatants. The weight of residual elastin was determined by weighing the dried pellets. The supernatants were freeze-dried and reconstituted in 200 μ l of distilled H₂O. Samples (10, 20 or 50 μ l) were loaded on to a Bio-Sil gel filtration column (SEC-400; 300 mm x 7.5 mm; Bio-Rad Laboratories, Richmond, USA) with a guard column (75 mm x 7.5 mm, Bio-Rad). The HPLC instrument (Varian 5000; Varian Associates Inc., Palo Alto, USA) was used with an on-line ultraviolet detector and a Varian 4270 integrator. Sodium phosphate buffer (100 mM, pH 6.8) was used with a flow rate of 1.0 ml/min. The column was calibrated with a mixture of proteins of known Mr (Bio-Rad gel filtration standard).

2.3.14 HIGH PRESSURE LIQUID CHROMATOGRAPHY OF HEPARIN FRACTIONS

Size exclusion chromatography was used to estimate size differences between heparin fractions isolated by HLE-Sepharose chromatography (see Section 2.3.9). The HPLC Bio-Sil gel filtration column and procedure are the same as described in Section 2.3.13 except that the column was fitted to a Bio-Rad HPLC 7000 system with a 1706 UV monitor set at 205 nm (Bio-Rad).

2.4 ASSAYS

2.4.1 α_1 -PI AND ELASTASE INHIBITORY FUNCTION OF PLASMA, SERUM, AND BRONCHOALVEOLAR LAVAGE.

2.4.1.1 INSTRUMENTATION.

A Cobas Bio centrifugal analyser with a DENS (Data Evaluation of Nonlinear Standards) curve option from Hoffman-LaRoche, Basle, Switzerland. Analyser settings used are listed in Table 1 and have been previously reported (Walsh et al., 1992).

2.4.1.2 ASSAY STANDARDS.

Protein Standard Plasma (human, stated α_1 -PI value 4.2 g/l) was our routine laboratory standard which had previously been used to determine serum reference intervals. The use of this commercial standard allowed a comparison of the assay values obtained for the volunteer and patient groups in this study with the previously derived reference range (1.6-3.5 g/l). The α_1 -PI in this standard was shown to be fully active in the functional assays by reducing the standard with 2-mercaptoethanol (Johnson and Travis, 1978). No inhibition was seen in the functional assay if the α_1 -PI was inactivated by oxidising the standard with nchlorosuccinimide (Jori et al., 1968), demonstrating that in plasma the inhibition is due to α_1 -PI alone. This standard was used as the assay calibrator in the assays described below in this section.

Purified α_1 -PI was functionally titrated against an active site-titrated preparation of PPE. The azapeptide, N-acetyl-L-alanyl-L-alanyl-Å-azanorvaline p-nitrophenyl ester was used as the active site titrant as described by Powers et al (1984). The purified α_1 -PI demonstrated that the commercial standard had a value 20-30% too high, which is consistent with other commercial preparations as discussed by Wewers et al (1987). However, since our routine laboratory assay and reference intervals are derived using this commercial standard value, we have for convenience used the stated commercial concentration in this study.

2.4.1.3 SUBJECTS

Serum samples were collected from 17 normal children (age <15 years). Normal adult serum samples (35 non-smokers, 20 smokers) were from a voluntary blood donor service. Serum samples were also collected from patients with established ARDS (n = 5) admitted to the Intensive Care Unit of the Royal Adelaide Hospital, and from pulmonary emphysema patients. Patients with emphysema (n = 19) were defined clinically by the presence of bullous disease on chest x-ray, and abnormal results for lung function tests. The Mann-Whitney U-test was used to test the significance of differences between subject groups.

BAL was performed in the following subjects. One group consisted of eight volunteers with no history of pulmonary disease, normal clinical examination and normal pulmonary function as demonstrated by spirometry, single breath diffusion capacity for carbon monoxide, and pulmonary elastic recoil using exponential analysis of the pressure volume curve (Colebatch et al., 1979b). Four subjects were life-long non-smokers, and the other four were current smokers with a history of at least 30 pack-years. The age range was 30-50 years. The smokers were allowed to smoke up to the time of the pre-medication injection, this being approximately 30 minutes prior to the BAL. The second group consisted of 5 ARDS patients who were clinically defined by a partial pressure of oxygen of less than 75 mm Hg with an inspired oxygen of 0.5 or greater, new diffuse bilateral infiltrates on x-ray, and a pulmonary artery wedge pressure of less than 18 mm Hg. Patients with clinical evidence of primary bacterial pneumonia or bronchoscopic evidence of bronchitis were excluded.

All subjects who participated in this study gave informed consent.

2.4.1.4 ASSAY SAMPLE HANDLING.

A preliminary study to establish whether anticoagulants interfered in the assays was performed on blood samples from 7 normal volunteers collected into heparin (12.5 International units/ml), sodium citrate (12 mM), dipotassium EDTA (5 mM) and sodium fluoride/potassium oxalate (23 mM/15 mM). All assays were optimised for linearity between 5 and 50 mg/l of α_1 -PI. To bring readings within this range all plasma and serum samples were diluted 1 in 100 with saline prior to assay. Because of the wide range of α_1 -PI concentrations found in normal volunteer BAL (n = 8, 0.22-3.75 mg/l) and particularly in ARDS patients BAL (n = 5, 9.0-400 mg/l), both unconcentrated and concentrated BAL samples were used to prepare serial dilutions to obtain 3 assay readings where possible within the linear portion of the assay standard plots (5-50 mg/l).

2.4.1.5 ESTIMATION OF α_1 -PROTEASE INHIBITOR WITH PORCINE PANCREATIC ELASTASE INHIBITION ASSAY.

This assay has been previously reported (Walsh et al., 1992). The assay buffer was 50 mM Tris HCl, 50 mM NaCl, 0.01% (v/v) Brij 35, pH 8.0. PPE was prepared in the assay buffer as a stock 1 g/l reagent and stored at -70° C in 50 ul aliquots. The stock HLE was diluted 1 in 250 in the assay buffer to give the working enzyme concentration of 4 mg / l such that 1.0 μ g is added in the "reagent volume" (Table 2.1). Suc-A-A-NA was used as the elastase substrate. It was prepared as a 0.2 mM stock solution in dimethyl sulfoxide and was diluted in the assay buffer 1 in 50 to give a concentration of 4 mM/l for the "start reagent".

2.4.1.6 ESTIMATION OF α_1 -PROTEASE INHIBITOR WITH HUMAN LEUKOCYTE ELASTASE INHIBITION ASSAY.

This assay has been previously reported (Walsh et al., 1992). HLE was prepared in the assay buffer (see in Section 2.4.1.5) as a stock 1 g/l reagent and stored at -70° C in 50 μ l aliquots. The stock HLE was diluted 1 in 500 in the assay buffer to give the working enzyme concentration of 2 mg/l such that 0.5 μ g is added in the "reagent volume" (Table 2.1). MeO-Suc-A-A-P-V-NA was used as the elastase substrate. It was prepared as a 0.2 M stock solution in dimethyl sulfoxide and was diluted in the assay buffer 1 in 50 to give a concentration of 4 mM for the "start reagent".

2.4.1.7 *IMMUNO-TURBIDIMETRIC* α_1 *-PROTEASE INHIBITOR ASSAY.*

This assay has been previously reported (Walsh et al., 1992). The assay buffer was 15 mM phosphate, 140 mM NaCl, 15 mM NaN₃ pH 7.4 containing 4% (w/v) polyethylene glycol (PEG) Mr 6000. Antiserum to human α_1 -PI was diluted 1 in 50 in the buffer/ PEG solution to give the working reagent that is added as the "reagent volume" (Table 2.1).

2.4.2 MEASUREMENT OF ELASTASE INHIBITION.

2.4.2.1 HUMAN LEUKOCYTE ELASTASE (2.5 μg) WITH SUC-A-A-A-NA AS SUBSTRATE.

This assay has been previously reported (Walsh et al., 1991a) and was performed on a Cobas-Bio centrifugal analyser as follows. Eighty μ l of sample (e.g. a HLE inhibitor) was mixed with 250 μ l of the assay buffer (see in Section 2.4.1.5), containing 2.5 μ g of HLE. After incubation at 25°C for 60 s, the residual HLE activity was measured by the addition of 20 μ l of 5 mM of Suc-A-A-NA in the same buffer. The rate of hydrolysis was monitored at 405 nm for 30 min. Residual HLE activity in the presence of an inhibitor was expressed as a percentage of the activity found in the absence of the inhibitor.

2.4.2.2 HUMAN LEUKOCYTE ELASTASE (0.5 μg) WITH MeO-SUC-A-A-P-V-NA AS SUBSTRATE (15 μl sample)

This assay was performed on the Cobas-Bio centrifugal analyser with as follows. Briefly, 15 μ l of sample (eg, a HLE inhibitor) was mixed with 250 μ l of the assay buffer (see in Section 2.4.1.5) containing 0.5 μ g of HLE. After incubation at 25°C for 60 s, 20 μ L of 5 mM of MeO-Suc-A-A-P-V-NA in the same buffer was added. The rate of hydrolysis was monitored at 405 nm for 5 min. Residual HLE activity in the presence of an inhibitor was expressed as a percentage of the activity found in the absence of the inhibitor.

2.4.2.3 HUMAN LEUKOCYTE ELASTASE (0.5 μg) WITH MeO-SUC-A-A-P-V-NA AS SUBSTRATE (75 μl sample).

This assay is a modification of that described in 2.4.2.2 but with sample, enzyme and substrate volumes being of equal volume to allow variations in the order of addition of reagents. Briefly the assay used 75 μ l of sample (eg, a HLE inhibitor), 75 μ L of assay buffer

(see in Section 2.4.1.5) containing 0.5 μ g of HLE and 75 μ l of 0.75 mM of MeO-Suc-A-A-P-V-NA in the same buffer.

2.4.2.4 HUMAN LEUKOCYTE ELASTASE (150 ng) WITH MeO-SUC-A-A-P-V-NA AS SUBSTRATE

This assay is a more sensitive variant of the above assay (Section 2.4.2.2) and was used for monitoring the inhibitory activity of heparin fractions from affinity column separations (Section 2.3.9) and heparinase digests of heparin (Section 2.3.11). Briefly, 80 μ l of a suitably diluted heparin fraction was mixed with 150 μ l of the assay buffer (see in Section 2.4.1.5), containing 150 ng of HLE and then the residual HLE activity was measured with 20 μ l of 2.5 mM of MeO-Suc-A-A-P-V-NA in the same buffer. The rate of hydrolysis was monitored at 405 nm for 15 min.

2.4.2.5 HUMAN LEUKOCYTE ELASTASE WITH HUMAN LUNG ELASTIN AS SUBSTRATE.

This assay has been previously reported (Walsh et al., 1991a). Elastin was added in triplicate to a microtitre tray as a 5 μ g suspension (particle size <250 μ m diameter) in 100 μ L of 50 mM Tris, 140 mM NaCl and 0.02% NaN₃, pH 8.0. HLE was added as 5 ng in 100 μ l of the Tris buffer. Assay samples (eg, HLE inhibitors) were added in triplicate in 100 μ l of the Tris buffer. After 24 h at 25°C in a humid chamber, 100 μ l of the supernatant was removed and soluble elastin peptides were measured with an ELISA method (Section 2.4.4).

83

2.4.2.6 PORCINE PANCREATIC ELASTASE WITH SUC-A-A-NA AS SUBSTRATE.

This assay has been previously reported (Walsh et al., 1991a) and was performed as described for the HLE assay in Section 2.4.2.1 except that PPE was used in place of HLE and that the assay was monitored for 10 min rather than 30 min.

2.4.3 MEASUREMENT OF CATHEPSIN G, CHYMOTRYPSIN AND TRYPSIN INHIBITION.

These assays have been previously reported (Walsh et al., 1991a) and are described briefly below.

2.4.3.1 CATHEPSIN G AND CHYMOTRYPSIN ASSAY.

This assay was performed as described for the HLE assay in Section 2.4.2.1 except that cathepsin G (0.1 μ g) or chymotrypsin was used in place of HLE with n-Succinyl-(L-alanine)₂-proline-L-phenylalanine-p-nitroanilide replacing the elastase substrate. The assay was monitored for 15 min. Assay variations included performing the assay at 37°C with a preincubation step of 30 min also at 37°C to detect any inhibition of this enzyme by heparin at this temperature.

2.4.3.2 TRYPSIN ASSAY.

This assay was performed as described for the HLE assay in Section 2.4.2.1 except that trypsin was used in place of HLE with α -Benzoyl-DL-arginine-p-nitroanilide replacing the elastase substrate. The assay was monitored for 15 min.

2.4.4 ELISA FOR ELASTIN PEPTIDES.

This assay has been reported previously (Dillon et al., 1990; Walsh et al., 1991a; Dillon et al., 1992). Briefly, an elastin peptide (α -elastin) preparation (see Section 2.3.6) was used to prepare standards. These standards together with unknowns were added to U-bottom microtitre plates in triplicate with a known amount of antisera (see Section 2.3.8) present in excess; the plates were then incubated overnight at 4°C. The assay mixture was then transferred to ELISA plates, which had been coated previously with 0.01 μ g of elastin peptides. The ELISA plates were incubated for 1 h at room temperature.

The presence of bound rabbit antibody on the ELISA plates was detected by the addition of peroxidase-conjugated goat antibody to rabbit immunoglobulin and subsequent colour development with 0.08% hydrogen peroxide and 2, 2'-azinobis(3-ethylbenzthiazoline)-6-sulphonic acid/mL. After sufficient colour development (usually 60 min.) the absorbance was read at 405 nm.

The antigen concentration of the unknowns was related to the standard curve (assay range 0.5-50 ng/ml), which was included on each plate to compensate for plate-to-plate variation. The inter-batch coefficient of variation was 5.2 % and the intra-batch coefficient of variation was 4.3 %, both at 0.4 ng/mL (n= 10).

2.4.5 URONIC ACID CONTENT OF GLYCOSAMINOGLYCANS.

This assay is a modification of that reported by Blumenkrantz and Asboe-Hansen, 1973. Briefly, 50 μ l of standards (0-200 μ g/ml of uronic acid prepared from D-glucuronic acid lactone) and unknown samples in duplicate were heated in a boiling water bath for 5 min in sealed glass digestion tubes together with 300 μ L of the sulphuric acid/tetraborate reagent. The samples were cooled on ice and each assay sample was split into two 140 μ l aliquots to which was then added 2 μ l of meta-hydroxydiphenyl. All samples were read within 5 min. at 520 nm on the Cobas-Bio centrifugal analyser. Thus four readings were taken on each assay sample. The Cobas-Bio cuvette rotor and sample tip were immediately discarded after the assay so that the sulphuric acid/tetraborate reagent did not damage the analyser.

2.4.6 TOTAL SULPHATE CONTENT OF GLYCOSAMINOGLYCANS.

Heparin fractions and heparin standards in duplicate were boiled in 0.75 M HCl for 1 hr. in sealed glass digestion tubes and then lyophilised in a Speedvac Concentrator (Savant Instruments, USA). The lyophilised samples and standards were reconstituted in distilled water and the sulphate content was determined with a turbidimetric method (Lundquist et al., 1980) adapted to the Cobas-Bio analyser. The method is described briefly as follows:- 8 μ L samples of hydrolysed heparin fractions and heparin standards together with sulphate standards (0-50 mM/l sulphate in the form of sulphuric acid) were diluted with 80 μ L of distilled water, then acidified with 215 μ L of 0.2 mM/l HCl followed by 75 μ L of BaCl₂/polyethylene glycol/Na₂SO₄ reagent (as in Lundquist et al., 1980). The increase in turbidity due to BaSO₄ was determined at 600 nm after 5 min.

2.4.7 QUANTITATION OF SULPHATED GLYCOSAMINOGLYCANS.

This dye binding method was primarily used to monitor eluted heparin and heparan sulphate fractions from affinity column separations (Section 2.3.9). The assay used is a modification of the 1,9-dimethylmethylene blue method reported by Farndale et al (1986) which, in this study, was adapted to the Cobas-Bio analyser. The method is described briefly as follows:- 16 mg of DMB dye was dissolved in 50 ml of 20% ethanol (AR grade) and stored as 5 ml aliquots at - 20° C. Before assay a thawed 5 ml aliquot of the dye was added to 100 ml of 0.2 M glycine/HCl buffer and the pH adjusted to pH 3.0 to make the working dye reagent. The assay uses 25 μ l of sample or heparin standards (0-50 ug/ml) to which was added 150 μ l of the working dye reagent and the absorbance was read at 525 nm after 1 minute.

2.4.8 ASSAY FOR THE DETERMINATION OF THE KINETIC CONSTANTS OF HLE.

The assay used by Nakajima et al (1979) to determine the kinetic constants for HLE, with Suc-A-A-NA as substrate was adapted to the Cobas-Bio analyser. Briefly, 25 μ l of a variable amount of substrate (0.0156-8 mmol/L of Suc-A-A-NA) and 2.5 μ g of HLE in 250 μ l, both in 0.1M HEPES buffer (pH 7.5) containing 0.5 M NaCl and 10% dimethylsulfoxide, were incubated at 25° C for 120 s. The rate of hydrolysis of the substrate was monitored at 405 nm for 30 min. after the substrate and enzyme were mixed. Assays were also run in the presence of 0.01% Brij 35 in the assay buffer.

| Parameter | Immunoturbidimetric | PPE Inhibition | HLE Inhibition |
|------------------------------------|---------------------|----------------|----------------|
| Listing | a1-PI Assay | Assay | Assay |
| 1 Units | mg/l | mg/l | mg/l |
| 2 Calculation factor | 1000 | 1000 | 1000 |
| 3 Standard 1 conc | 5 | 5 | 5 |
| 3 Standard 2 conc | 10 | 10 | 10 |
| 3 Standard 3 cone | 20 | 20 | 20 |
| 3 Standard 4 conc | 30 | 30 | 30 |
| 3 Standard 5 conc | 40 | 40 | 40 |
| 3 Standard 6 conc | 50 | 50 | 50 |
| 6 Limit | 0 | 0 | 0 |
| 7 Temperature (°C) | 25.0 | 25.0 | 25.0 |
| 8 Type of analysis | 7.5 | 7.5 | 7.5 |
| 9 Wavelength (nm) | 340 | 405 | 405 |
| 10 Sample volume (μ l) | 50 | 25 | 10 |
| 11 Diluent volume (μ l) | 00 | 00 | 00 |
| 12 Reagent volume (μ l) | 250 | 250 | 250 |
| 13 Incubation time (sec) | 0 | 600 | 10 |
| 14 Start reagent volume (μ l) | 0 | 20 | 20 |
| 15 Time of first reading (sec) | 1.0 | 30.0 | 30.0 |
| 16 Time interval (sec) | 20 | 30 | 10 |
| 17 Number of readings | 20 | 11 | 15 |
| 18 Blanking mode | 1 | - 1 | 1 |

Table 2.1: Cobas Bio settings

2.5 DATA INTERPRETATION AND STATISTICAL ANALYSIS

Before any assay data were accepted for interpretation, the assay was performed in at least triplicate, within assay, and in quadruplicate between assays. This assessed the reliability of the assay and allowed the data to be scrutinised for outliers before being subjected to statistical analysis. The paired *t* test was used to determine whether there was a significant difference between two sets of assay data. The Mann-Whitney U-test was used to test the significance of differences between subject groups. This non-parametric test was chosen as it makes no assumptions about the underlying distribution and is particularly suited to small data sets, as used in this thesis. A significant difference for both tests was considered where p < 0.05, with p < 0.01 being highly significant.

CHAPTER THREE

ANTI-ELASTASE ACTIVITY OF PLASMA, SERUM AND BAL.

3.1 INTRODUCTION

Although the cause of emphysema remains uncertain, it is currently thought that it results from an imbalance of the protease-anti-protease systems in the lung with HLE being implicated as the main elastolytic enzyme as discussed in chapter one of this thesis. This imbalance leads to digestion of elastin, with structural changes and dilatation of air spaces that are the hallmarks of this disease. The main protease inhibitor that regulates excessive HLE activity is α_1 -PI. As α_1 -PI is responsible for 80-90% of the anti-elastase activity towards HLE in serum (Ohlsson, 1978) and in the peripheral airways (Gadek et al., 1981b, Kramps et al., 1988), a reduction in α_1 -PI levels or function should, according to the theory, lead to the development of emphysema. A lower concentration of α_1 -PI with a reduced inhibitory function towards HLE has been reported in the severe genetic PiZ form of α_1 -PI deficiency which is frequently associated with pulmonary emphysema in adults (Eriksson, 1965, Sharp et al., 1969). However the question as to whether there are reduced levels of α_1 -PI and increased amounts of functionally inactive α_1 -PI present in the majority of emphysema cases, which are due to smoking, remains unresolved. It is known that α_1 -PI can be inactivated by oxidising agents resulting in the loss of binding of α_1 -PI towards HLE. This oxidative inactivation of α_1 -PI can occur during acute pulmonary inflammation due to the release of various oxidants from activated neutrophils (Zaslow et al., 1983), an example being the acute inflammatory lung injury that is seen in ARDS (Lee et al., 1981; McGuire et al., 1982; Cochrane et al., 1983). In this chapter, the total HLE inhibitory capacity and the inhibitory capacity due to α_1 -PI alone was determined in serum and BAL from non-smokers, smokers and patients with emphysema and ARDS. The aim being to determine whether there are any changes in reduced levels of α_1 -PI and increased amounts of functionally inactive α_1 -PI in individuals with emphysema and in smokers compared to non-smokers. The

ARDS group was used as a positive control because of the increased amount of functionally inactive α_1 -PI that has been reported in this acute inflammatory disease of the lung.

The development of the assays to measure total HLE inhibitory capacity and the inhibitory capacity due to α_1 -PI alone is described as follows. As HLE is the protease that is generally accepted to be involved in the lung destruction seen in emphysema, and is reported to bind with α_1 -PI many times faster than any other protease (Beatty et al., 1980; Travis and Salvesen, 1983), it was used to develop a rapid assay for functional α_1 -PI. The rapid binding of α_1 -PI to HLE (10 sec in this assay) was combined with rate-analysis capability of the Cobas Bio centrifugal analyser, the sensitive substrate MeO-Suc-A-A-P-V-NA, and the HLE activator Brij 35 so as to allow a fast assay (180 sec) suitable for rapid processing of a large number of samples. This assay (see Section 2.4.1.6) was used to measure the inhibitory activity of α_1 -PI in plasma and the HLE inhibitory capacity of BAL fluid from normal subjects and individuals with emphysema and ARDS. This assay was compared to two other α_1 -PI assays on the Cobas Bio analyser. One was an immuno-turbidimetric assay which uses a specific antibody towards α_1 -PI (see Section 2.4.1.7) to allow the measurement of all immuno-reactive α_1 -PI present, whether it is functional or not. The other was a PPE inhibitory assay that measures the inhibition due to functional α_1 -PI alone (see Section 2.4.1.5). The assay parameters used on the Cobas Bio centrifugal analyser for the three assays are compared in Table 2.1 (Chapter 2).

3.2 ASSAY EVALUATION

MeO-Suc-A-A-P-V-NA was used as substrate for the HLE inhibition assay on the basis of data reported previously (Nakijima et al., 1979) that it was 200 times more sensitive than the commonly used elastase substrate Suc-A-A-NA. This substrate allowed the development of an automated rapid assay (3 min assay time) on the Cobas Bio analyser for the detection of HLE. α_1 -PI has a high association rate with HLE ($t_{1/2 \text{ ass}} = 0.61$ msec; Travis and Salvesen, 1983) which allowed the use of the minimum incubation time on the analyser of 10 seconds. No
significant difference was observed when either a 10 sec or a 15 min incubation step was used prior to substrate addition for the assay of plasma or BAL (see Section 3.5). While a difference was noted for purified α_1 -PI where 86% inhibition was seen after a 10 sec. compared to a 15 min. incubation period (Figure 3.1), no significant difference was seen with the standard used in the assay, Protein Standard Plasma (Figure 3.2). The use of this sensitive substrate on the Cobas Bio analyser, together with Brij 35, allowed for a rapid assay of 3 min with 10 μ l of a 1 in 100 dilution of plasma or serum (Table 2.1). Brij 35 was added to the assay buffer on the basis of reports that surfactants activate HLE (Starkey and Barrett, 1976; Cook and Ternai, 1988a; Wenzel et al., 1990; Walsh et al., 1991b). In this assay system final concentrations of Brij 35 greater than 0.005% increased sensitivity by increasing HLE activity by as much as 40% (see Section 5.2.1 and Figure 5.1). A Brij 35 concentration of 0.01% was used routinely in the assay. The increased activity of HLE in the presence of Brij 35 allowed the use of a smaller amount of HLE (0.5 μ g) in the assay so that low concentrations of HLE inhibitors can be detected, as in BAL fluid.

Suc-A-A-NA is the most effective (Nakijima et al., 1979) and commonly used substrate for PPE and has been widely used in a number of studies to demonstrate the presence of inactive α_1 -PI in plasma and BAL on the basis that significant inhibition of PPE is only seen with functional α_1 -PI (eg. Klumpp and Bieth, 1979; Boudier et al., 1983; Morrison et al., 1987). Assay conditions and instrument parameters were optimised to obtain the same rate of absorbance change as found for the HLE inhibition assay (Figure 3.3; Table 2.1).

All α_1 -PI inhibition values were obtained from the linear part of the calibration plots (5-50 mg/l, Figure 3.3) using the DENS (Data Evaluation of Nonlinear Standards) curve option on the analyser to allow for any non-linearity. The inhibition assays were compared with an immuno-turbidimetric assay for α_1 -PI on the Cobas Bio Analyser. The settings for this assay are shown in Table 2.1.



Alpha-1-antiprotease ug/ml

Figure 3.1: Effect of incubation time on alpha-1-antiprotease inhibition of HLE with MeO-Suc-A-A-P-V-NA as substrate. Assay as described in Section 2.4.16



Alpha-1-antiprotease ug/ml

Figure 3.2: Effect of incubation time on plasma (Behring standard plasma) inhibition of HLE with MeO-Suc-A-A-P-V-NA as substrate. Assay as described in Section 2.4.1.6

% HLE Activity



Alpha-1-antiprotease ug/ml

Figure 3.3: Calibration plots for HLE and PPE inhibition assays with Behring protein standard plasma as calibrator. MeO-Suc-A-A-P-V-NA and Suc-A-A-NA as substrates for HLE (Section 2.4.1.6) and PPE (Section 2.4.1.5) respectively.



Figure 3.4: Effect of anticoagulants on HLE inhibition assay. Serum and plasma samples from normal volunteers (n=7) were collected into the anticoagulants described in Section 2.4.1.4. HLE inhibition assay is as described in Section 2.4.1.6.

3.3 EFFECT OF ANTI-COAGULANTS

Initial studies with heparinised plasma in the HLE inhibition assay gave apparent α_1 -PI values greater than twice that measured with the PPE inhibition and the immuno-turbidimetric assays. This apparent elevation in α_1 -PI activity was found to be due to the presence of heparin which inhibited HLE activity (see Chapter 4). An initial study was then undertaken to determine what anti-coagulants could be used in the three α_1 -PI methods. It was found that plasma containing commonly used anti-coagulants and serum could be employed in these assays except plasma containing heparin, which gave results significantly higher in the HLE inhibition assay (Figure 3.4). Thus, heparinised samples must be avoided if the HLE inhibition assay is used.

| | Sample | Mean <u>±</u> SD (g/L) | CV (%) |
|------------------------|--------|---------------------------|-----------|
| Intra-Assay $(n = 20)$ | | | |
| Immuno-turbidimetric | А | 1.25 ± 0.02 | 1.8 |
| | В | 2.58 ± 0.0 | 0.9 |
| HLE Inhibition | А | 1.16 ± 0.04 | 3.5 |
| | В | 2.46 ± 0.04 | 1.5 |
| PPE Inhibition | А | 1.26 ± 0.06 | 4.7 |
| | В | 2.42 ± 0.06 | 2.4 |
| Inter-Assay (n = 15) | | | |
| Immuno-turbidimetric | В | 3.10 ± 0.11 | 3.4 |
| HLE Inhibition | В | 2.80 ± 0.15 | 5.3 |
| PPE Inhibition | В | 2.75 ± 0.18 | 6.6 |

Table 3.1: Imprecision of assay

3.4 SERUM α_1 - PROTEASE INHIBITOR CONCENTRATIONS

It is important in comparison studies, where different assays are used to measure the same parameter, that reproducibility of the assays be known so that results can be correctly interpreted allowing for this assay variability. Within run and between run imprecision were determined with one sample below the reference range (Sample A) and one in the middle of the reference range (Sample B). The intra-batch and inter-batch imprecision for both samples (A and B) were satisfactory for the HLE inhibition assay, and fell between the figures found for the immuno-turbidimetric and PPE inhibition assays (Table 3.1).

The immuno-turbidimetric and HLE inhibition assays were used to determine α_1 -PI concentrations in children, normal adults and adults with emphysema or ARDS (Figure 3.5). The reference intervals for normal homozygotes (α_1 -PI phenotype MM) were previously determined in our routine laboratory on blood donor serum samples with the immuno-turbidimetric α_1 -PI assay to be 1.6-3.5 g/l. In the present study the reference intervals were found to be lower. The range (mean ± 2 SD) for the combined adult non-smokers, and smokers (n = 55) were 1.0-3.3 g/l and 1.0-2.8 g/l for the immuno-turbidimetric and HLE inhibition assays respectively. Total (immuno-turbidimetric) and functional (inhibition) α_1 -PI were significantly increased in children (both p < 0.01) and in adults with emphysema (p < 0.05 and p < 0.01 respectively) when compared to those for normal adults (Figure 3.5). The mean values in smokers did not differ significantly from those in non-smokers. Values for ARDS were significantly higher than in any other group (both p < 0.05).

When the functional α_1 -PI concentration (HLE assay) was expressed as a percentage of the immuno-turbidimetric α_1 -PI concentration (Figure 3.6) the only difference noted was that for the ARDS group where the percentage was reduced (p < 0.05). The PPE inhibition assay on the above groups correlated well with the HLE inhibition assay (see also comparison data in Section 3.5).

3.5 COMPARISON OF α_1 - PROTEASE INHIBITOR CONCENTRATION IN SERUM WITH BAL FLUID

Both HLE and PPE inhibition assay values were expressed as a percentage of the immunoturbidimetric α_1 -PI concentrations in a group of normal volunteer smokers (n = 4) and nonsmokers (n = 4) as shown in Figures 3.7 and 3.8. Serum α_1 -PI was 91% active in non-smokers and 97% active in smokers for the PPE assay (Figure 3.7), and 94% active in non-smokers and 96% active in smokers in the HLE assay (Figure 3.8a).

In BAL fluid, using the PPE inhibition assay, the α_1 -PI was 70% active in non-smokers and 85% active in smokers (Figure 3.7). However, in the HLE assay the figures were 138% and 153% respectively (Figure 3.8a), demonstrating the presence of HLE inhibitors other than α_1 -PI in normal BAL fluid. Nevertheless, there was no significant difference in values for BAL fluid between smokers and non-smokers. There was also no significant difference between smokers and non-smokers immuno-turbidimetric α_1 -PI concentrations in serum or BAL samples (Figures 3.7, legend).

In patients with ARDS there was an approximately two-fold increase for immuno-turbidimetric α_1 -PI concentrations in serum (p < 0.05) (Figure 3.9: mean ± SD, 4.50 ± 2.45 g/l) when compared to normal volunteers (Figure 3.7: non-smokers, 2.08 ± 0.39; smokers 2.03 ± 0.30). However, the α_1 -PI was only 76% active in both the PPE and HLE inhibition α_1 -PI assays (p < 0.01) as shown in Figures 3.9 and 3.10. In BAL fluid from ARDS patients there was a greater than 40-fold increase in the immunoturbidimetric α_1 -PI (p < 0.01) (Figure 3.9: 0.101 ± 0.167 g/L) when compared to normal volunteers (Figure 3.7: non-smokers, 0.00112 ± 0.00094, smokers, 0.00235 ± 0.00150). However, the α_1 -PI in BAL fluid from ARDS patients was only 37% active as measured in the PPE inhibition α_1 -PI assay (p < 0.05) (Figure 3.9). As is the case for normal BAL fluid inhibition of HLE (Figure 3.8a), the inhibition of HLE by ARDS BAL fluid was elevated (118%; Figure 3.10a) also indicating the presence of other HLE inhibitors in this fluid.



Figure 3.5: Comparison of serum immuno-turbidimetric (•) and HLE inhibition assays (+) for the estimation of α_1 -PI concentration in children (<15 years, n = 17), normal adults (non-smokers, n = 35; smokers, n = 20) and adults with emphysema (n = 19) and ARDS (n = 5). Each group is shown as mean \pm standard error. Assay procedures are as described in Sections 2.4.1.6 and 2.4.1.7.



Figure 3.6: Comparison of % active α_1 -PI concentration in children (<15 years, n = 17), normal adults (non-smokers, n = 35; smokers, n = 20) and adults with emphysema (n = 19) and ARDS (n = 5). Active α_1 -PI concentration is determined as a ratio of serum HLE inhibition and immuno-turbidimetric assays. Each group is shown as mean \pm standard error. Assay procedures are as described in Sections 2.4.1.6 and 2.4.1.7.





Figure 3.7: Comparison of % active α_1 -PI concentration in serum and BAL from normal volunteers - 4 non-smokers (o), 4 smokers (•) using the PPE inhibition assay expressed as a ratio of the immunoturbidimetric assay. The immuno-turbidimetric α_1 -PI concentration (g/l, mean \pm SD) for non-smokers and smokers respectively were: Serum, 2.08 \pm 0.40, 2.03 \pm 0.31; BAL, 0.00112 \pm 0.00094, 0.00235 \pm 0.00150. The PPE inhibitory capacity values for non-smokers and smokers respectively were: Serum, 1.89 \pm 0.38, 1.97 \pm 0.27; BAL, 0.00072 \pm 0.00079, 0.00202 \pm 0.00130. Assay procedures are as described in Sections 2.4.1.5 and 2.4.1.7.



Figure 3.8 a & b: Comparison of % active α_1 -PI concentration in serum and BAL from normal volunteers - 4 non-smokers (0), 4 smokers (•) using the HLE inhibition assay with 10 sec (a) or 15 min (b) incubation expressed as a ratio of the immuno-turbidimetric assay. The immunoturbidimetric α_1 -PI concentrations are as for Figure 3.7. The HLE inhibitory capacity values for non-smokers and smokers respectively were: Serum, 2.15 ± 0.33, 1.95 ± 0.29; BAL, 0.00141 ± 0.00101, 0.00383 ± 0.00286. Assay procedures are as described in Sections 2.4.1.6 and 2.4.1.7.



Figure 3.9: Comparison of % active α_1 -PI concentration in serum and BAL from ARDS patients (n = 5) using the PPE inhibition assay expressed as a ratio of the immuno-turbidimetric assay. The immuno-turbidimetric α_1 -PI concentration (g/l, mean \pm SD) for ARDS patients in serum and BAL respectively were: 4.50 \pm 2.45 and 0.101 \pm 0.167. The PPE inhibitory capacity values for ARDS patients in serum and BAL respectively were: 3.44 \pm 1.76, 0.072 \pm 0.144. Assay procedures are as described in Sections 2.4.1.5 and 2.4.1.7.



Figure 3.10 a & b: Comparison of % active α_1 -PI concentration in serum and BAL from ARDS patients (n = 5) using the HLE inhibition assay with 10 sec (a) or 15 min (b) incubation expressed as a ratio of the immuno-turbidimetric assay. The immuno-turbidimetric α_1 -PI concentrations are as for Figure 3.9. The HLE inhibitory capacity values for ARDS patients in serum and BAL respectively were: 3.86 ± 2.38, 0.103 ± 0.166. Assay procedures are as described in Sections 2.4.1.6 and 2.4.1.7.

While the incubation period of 10 sec, 5 or 15 min of either purified α_1 -PI or Protein Standard Plasma with HLE did not lead to significant differences in the HLE inhibition assay (Figures 3.1 and 3.2 respectively) there may be significant differences with serum and BAL samples from patients. To examine this possibility, the HLE inhibition assays on normal volunteers and ARDS patients as shown in Figures 3.8a and 3.10a respectively were also repeated with an incubation step of 15 minutes; the results of which are shown in Figures 3.8b and 3.10b respectively. However, no significant differences were seen whether the normal volunteer plasma samples from non-smokers and smokers combined (n = 8) were incubated for 10 sec (Figure 3.8a, 93.8% inhibition) or 15 min (Figure 3.8b, 93.5% inhibition). This was also the case for ARDS plasma samples (n = 5) whether incubated for 10 sec (Figure 3.10a, 76.4%) inhibition) or 15 min (Figure 3.10b, 75.2% inhibition). With BAL fluid, where other HLE inhibitors are present, there could theoretically be differences in inhibition rates. However, no significant differences were seen whether the normal volunteer BAL samples from nonsmokers and smokers combined (n = 8) were incubated for 10 sec (Figure 3.8a, 145.4% inhibition) or 15 min (Figure 3.8b, 133.4% inhibition) and similarly for ARDS plasma samples (n = 5) incubated for 10 sec (Figure 3.10a, 117.8% inhibition) or 15 min (Figure 3.10b, 117.5%) inhibition).

3.6 **DISCUSSION**

In this chapter, a rapid method for the measurement of functional α_1 -PI (HLE enzyme inhibition) which uses as little as 10 μ L of 100-fold diluted plasma or serum is described. This assay has a shorter combined incubation and assay time (180 sec) than the immuno-turbidimetric (401 sec) and the PPE inhibition assay (960 sec) as shown in Table 2.1. The imprecision is acceptable and falls between that of the immuno-turbidimetric and PPE inhibition assays (Table 3.1). The only disadvantage is that heparinised plasma cannot be used (Figure 3.4).

Inhibition of PPE has previously been used to assess the oxidative inactivation of α_1 -PI in serum (Beatty et al., 1982). In the present study we have used HLE in lieu of PPE because it is the enzyme against which α_1 -PI is primarily directed (Beatty et al., 1980; Travis and Salvesen, 1983). The HLE inhibition assay, like the assay using PPE, has an advantage in serum or plasma in that only the protease inhibitory role of α_1 -PI is being measured and not inactive α_1 -PI such as the complexed, oxidised or fragmented forms which may be measured in the immuno-turbidimetric assay. For example, the oxidised form of α_1 -PI is approximately 2000fold less able to inhibit HLE (Johnson and Travis, 1978). This partially explains the data in Figure 3.5 where HLE functional values were always slightly lower although not significantly less than the corresponding immuno-turbidimetric values. However, the difference is clearly shown in Figure 3.6 where the HLE value is expressed as a percentage of the immunoturbidimetric value. All groups except the ARDS group had approximately 90% functional α_1 -PI activity towards HLE whereas the ARDS group had only 76% active α_1 -PI. This is not surprising as ARDS is an acute disease in which an elevated plasma α_1 -PI is expected (Figure 3.5) because of the inflammatory response. It is likely that much of the α_1 -PI is oxidised or complexed as has been shown in previous studies (Lee et al., 1981; McGuire et al., 1982; Cochrane et al., 1983) and this is consistent with the data presented in Figure 3.6.

The mean immuno-turbidimetric and HLE inhibition assay concentrations in the children, the emphysema and the ARDS groups were raised when compared to the means of adult normal non-smokers and smokers, although many individual values in the children and the emphysema group still fell in the reference range of 1.6-3.5 g/L as used by our routine laboratory (see Figure 3.5). The adult normals did not show any significant differences between non-smokers and smokers in α_1 -PI concentrations (Figure 3.5) or percentage α_1 -PI activity (Figure 3.6).

 α_1 -PI functional concentrations in BAL fluid are likely to reflect, more closely than serum, the inflammatory activity that occurs in the lung. This is shown with the PPE inhibition assay by the much higher inactivation of α_1 -PI in BAL fluid from some ARDS individuals when

compared to the patient's plasma (Figure 3.9). However, the HLE inhibition assay in BAL fluid from normal volunteer and ARDS patients reflects not only α_1 -PI inhibitory activity but also the inhibitory activity of other HLE inhibitors in the lung, as demonstrated by the increased HLE inhibitory activity (Figures 3.8 and 3.10). As with the PPE inhibition assay, there was also less functional activity with the HLE inhibition assay in the ARDS fluid (118%) than in the BAL fluid from normal volunteers (145%).

This increased BAL inhibitory activity towards HLE compared to PPE has been reported in another study (Wewers et al., 1989), although the authors did not relate the activity to α_1 -PI standards and thus did not infer that the increased inhibition is due to inhibitors other than α_1 -PI. They also reported that BAL from smokers took approximately 1.5 times longer to inhibit HLE than in non-smokers by using multiple incubation times from 10 to 80 sec. This study did not aim to specifically examine this difference as only three incubation times of 10 sec, 5 min, or 15 min were used, and no significant difference was detected between non-smokers and smokers (see Figure 3.8a and 3.8b). However, the results of short incubation period of 10 seconds demonstrate that α_1 -PI and the other inhibitors in BAL act rapidly.

The PPE inhibition assay, used as a comparison method, demonstrated in a volunteer sub-group (Figure 3.7) that there was no significant difference in functional α_1 -PI activity between either serum or BAL fluid in smokers or non-smokers. This observation supports other reports (Boudier et al., 1983; Stone et al., 1983) which show similar active α_1 -PI levels in BAL from both smokers and non-smokers. These results suggest that the development of emphysema in smokers is not simply due to an acquired deficiency of active α_1 -PI as has been found in individuals with the ZZ phenotype inherited form of α_1 -PI deficiency, as has been proposed by Travis (1988).

In conclusion, both inhibition assays as described here on the Cobas Bio centrifugal analyser measure functional α_1 -PI in serum, but the HLE inhibition assay is preferable as it uses less sample, requires less assay time and is more precise (Tables 2.1 and 3.1). Whilst this assay is suitable for undiluted and diluted samples in the 5-50 mg/L assay range, the assay has also been used to measure unconcentrated BAL from normal volunteers in the 0.05-1 mg/l assay range by increasing the sample volume (x10), decreasing the HLE concentration (x10) and increasing the assay time from 3 to 30 min. It also has the advantage over other α_1 -PI assays in that it is able to detect not only functional α_1 -PI in BAL but also the inhibitory action of other BAL inhibitors towards HLE. This assay may have a role in monitoring the effectiveness of various treatments developed to augment the anti-elastase defences in acute inflammatory disease such as ARDS, and in chronic inflammatory diseases such as emphysema.

CHAPTER FOUR

ELASTASE INHIBITORS

4.1 INTRODUCTION

There is considerable evidence suggesting that HLE is the main protease responsible for the protease-antiprotease imbalance that leads to the development of emphysema. The evidence is based primarily on the plausible role of HLE in generating emphysema in individuals carrying an inherited deficiency of α_1 -PI, the major endogenous serine proteinase inhibitor for HLE, and its ability to mimic emphysema when administered as a purified enzyme in animal lungs (Janoff, 1985, Snider, 1992). This has led to studies with endogenous inhibitors and the development of many low molecular weight elastase inhibitors with the eventual aim of clinically supplementing the naturally occurring antiproteases in the lung.

Despite its very low elastolytic activity, cathepsin G is considered to be the other major inflammatory serine protease to HLE, which is secreted in neutrophil degranulation. Its occurrence in cells essentially mirrors that for HLE but in addition is found in mast cells (Caughey, 1994). However its physiological action and its role in disease states has not been established although a number have been proposed. One proposal is that cathepsin G acts as a mediator for the function of HLE and other elastase-related proteases secreted from the azurophil granules of neutrophils. Some examples are as follows. Cathepsin G could assist HLE function by allowing more efficient elastolysis because of its different substrate specificity to HLE, which allows it to bind to and act on different protein sequences (Lonky and Wohl, 1983). There have been reports of HLE activity toward elastin being enhanced in the presence of cathepsin G (Boudier et al., 1981; Reilly et al., 1984). There is evidence that cathepsin G is required for HLE to be activated (Kubes et al.,

1993). Cathepsin G has also been reported to be able to convert the latent form of neutrophil collagenase to the active form (Capodici et al., 1989).

As well as α_1 -PI, another proteinase inhibitor that may have a role in the development of emphysema is α_2 -M. α_2 -M appears to be a non-specific proteinase inhibitor, which is able to bind to the majority of all proteinases, including HLE. When complexed to α_2 -M, proteinases typically retain 80-100% of their hydrolytic activity against low molecular weight substrates but less than 10% against large proteins (Barrett, 1981). On the basis of this residual proteolytic activity, it has been proposed that α_2 -M-HLE complexes may play a role in the development of pulmonary emphysema (Galdston et al., 1979; Stone et al., 1979), which is supported by the detection of α_2 -M-PPE complexes in BAL from the lungs of hamsters with PPE-induced emphysema (Stone et al., 1982). Clinically, because of its size (MW 725,000), α_2 -M cannot readily diffuse into the lung alveolus, where it is usually at such low concentrations that it is difficult to detect in normal BAL. Large amounts of α_2 -M have been detected in BAL from patients with ARDS in this laboratory (McLennan et al., 1984) and by others (Wewers et al., 1988) and in BAL from patients with cystic fibrosis (Bruce et al., 1985). However, the specific physiological role of α_2 -M as an antiprotease has not been clearly established, let alone a role in the development of pulmonary emphysema.

Central to the proteolytic mechanism in the development of emphysema is the ability of HLE to bind to and degrade lung elastin. Although HLE has a very broad specificity, studies with oligopeptide substrates show that the enzyme preferentially binds to sequences rich in amino acids with short hydrophobic side chains close to lysine derived cross-links (Yasutake et al., 1981) with cleavage occurring at carboxyl side of residues, particularly valine and alanine (Stein et al., 1987). Thus, studies of HLE inhibitors may be dependent on the relative affinity of HLE for the type of substrate and inhibitor used. For example, HLE has been shown to bind so strongly to insoluble elastin that its inhibition by specific inhibitors is markedly impaired (Morrison et al., 1990, Padrines and Bieth, 1991) which is also likely to be influenced by binding of HLE to elastin derived peptides released during elastinolysis (Tyagi and Simon, 1993). An elastin peptide preparation (κ -elastin) has also been reported to inhibit HLE activity (Bonnaure-Mallet et al., 1995).

A number of compounds, such as GAGs, have been shown to interact with HLE and have the potential to modify the effectiveness of HLE inhibitors. GAGs such as heparin and the structurally similar heparan sulphate are components of proteoglycans which are found both in mast cells (Langunoff et al., 1964, Bergqvist et al., 1971), as part of the luminal surface of cell membranes (Hook et al., 1984) and as part of tissue surfaces such as the vascular endothelium (Marcum and Rosenberg, 1984). The most studied form is the commercially available standard clinical heparin, which is extracted from porcine mucosa and consists of a pool of sulphated polysaccharide chains of varying lengths with an average molecular weight of 14 ± 1 kDa. Heparin has been shown to interact with numerous biological systems (Lane and Lindahl, 1989) including a number of PMN functions, where it has been shown to have an anti-inflammatory role such as the inhibition of HLE activity (Redini et al., 1988a; Walsh et al., 1988; Walsh et al., 1991a). Heparin is able to bind to a large number of proteins (Jackson et al., 1991) because of its polyanionic nature and because it contains iduronic acid residues which allows the carbohydrate backbone of heparin sufficient flexibility to fit many relatively fixed chemical configurations that are electropositive (Casu et al., 1988). The binding of heparin to HLE is essentially by electrostatic interactions, which probably involves the negatively charged sulphate groups on heparin and some of the positively charged guanidinium groups of 19 arginine residues located in clusters on the surface of HLE (Bode et al., 1989; Navia et al., 1989).

In this chapter the effectiveness of a number of endogenous and exogenous protease and HLE inhibitors towards HLE and cathepsin G is examined. Inhibitors evaluated include α_1 -PI, α_2 -M, elastin peptides, sulphated GAGs, and the low molecular weight inhibitors, ICI 186756 and 200355.

4.2 DETERMINATION OF HUMAN LEUKOCYTE ELASTASE ACTIVITY

The HLE used throughout this study was a chromatographically pure preparation from human sputum supplied by Elastin Products. It was stated to be free of cathepsin G activity which was confirmed by assay for enzymic activity (see Section 2.4.3.1 for assay details). Elastin Products stated the product to be at least 90% active. This has been confirmed in two reports; one using active site titrants (Williams et al., 1987) and the other using previously published kinetic constants of Nakajima et al (1979) to determine enzyme activity (Burnett et al., 1988). Because the active site titrants were not readily available, the activity of the enzyme was determined using the published kinetic constants. The assay conditions used were exactly those reported by Nakajima et al (1979) and are shown in Section 2.4.8. The kinetic constants K_m and V_{max} were determined by varying the concentration of the substrate, Suc-A-A-NA using the reciprocal form of the Michaelis-Menten equation :-

 $1/V = 1/V_{max} + K_M/V_{max} \cdot 1/[S]$



Figure 4.1: Determination of kinetic constants for HLE with 0.0156-8 mM of the substrate, Suc-A-A-NA. Assay conditions as described in Section 2.4.8.



Figure 4.2: Determination of kinetic constants for HLE with 0.5-8 mM of the substrate, Suc-A-A-A-NA. Assay conditions as described in Section 2.4.8.

Linear regression equations:

No Brij 35:
$$y = 0.498 + 1.774x$$
 $r^2 = 0.998$
+ Brij 35: $y = 0.212 + 0.980x$ $r^2 = 0.999$

The overall catalytic rate constant k_{cat} is obtained by dividing the maximal rate V_{max} by the total concentration of active enzyme [E⁰]. Initial studies with a substrate concentration range of 0.0156-8 mmol/L of Suc-A-A-NA (Figure 4.1) showed that the reciprocal of the substrate concentration (ie. 1/[S]) only gave a linear plot in the presence of Brij 35 which is routinely included in assays used in this thesis for HLE activity (see Section 2.4). The kinetic constants derived by Nakajima et al (1979) were obtained with a substrate concentration range of 0.4-7.6 mmol/l in the absence of Brij 35. Replotting Figure 4.1 using the substrate range of 0.5-8 mmol/l gave reproducible linear plots (Figure 4.2) in the absence of Brij 35. The kinetic constants found in the absence and presence of Brij 35 and those obtained by Nakajima et al (1979) are shown in Table 4.1.

| Ta | b | le 4 | 4.1 | Kinetic | constants | for | HLE | with | Suc-A- | - A -A | A-NA | as s | ubstrate. |
|----|---|------|-----|---------|-----------|-----|-----|------|--------|---------------|------|------|-----------|
|----|---|------|-----|---------|-----------|-----|-----|------|--------|---------------|------|------|-----------|

| | K _M (mmol) | $\frac{\mathbf{k}_{cat}}{(s^{-1})}$ | $\frac{\mathbf{k_{cat}} / \mathbf{K_M}}{(mol^{-1} s^{-1})}$ |
|--------------------------------------|--------------------------|-------------------------------------|---|
| This study (no Brij 35) | 3.6 | 1.9 | 501 |
| This study (with 0.01% Brij 35 v/v) | 4.6 | 4.5 | 987 |
| Nakajima et al (1979, no surfactant) | 3.7 | 2.1 | 570 |

The kinetic constants obtained in the absence of surfactant are very close to those obtained by Nakajima et al (1979) and would support the manufacturer (Elastin Products) statement that the product is at least 90% active enzyme as also found by Williams et al (1987) using active site titrants. In the presence of Brij 35 the plot is essentially linear to a very low substrate

concentration of 0.0156 mmol/L (ie. 1/S of 64) of Suc-A-A-NA (Figure 4.1), which is not the case in the absence of Brij 35 where non-linearity is seen below 0.125 mmol/L (ie. 1/S of 8). What is most interesting is the approximately 2 fold activation of HLE by 0.01% Brij 35 with an increase (27%) in the K_m . An increase in k_{cat} of 40% was reported by Cook and Ternai (1988a) using the substrate MeO-Suc-A-A-P-V-NA which is consistent with the findings using this substrate as reported and discussed further in Chapter 5.

4.3 **PROTEIN INHIBITION**

4.3.1 INHIBITION OF HUMAN LEUKOCYTE ELASTASE

In this chapter and chapter 5 the actual mass of the inhibitor added to the reaction cuvette was chosen in preference to mass/volume so as to emphasise the mass ratio of inhibitor to enzyme in the cuvette. This allowed a direct comparison of the effectiveness of inhibitors towards an enzyme and overcame the problem of comparing different assay systems using different sample and assay volumes. A number of proteins were evaluated for their ability to inhibit HLE (0.5 μ g) with MeO-Suc-A-A-P-V-NA as substrate (see Section 2.4.2.2 for assay conditions; note that instead of Suc-A-A-P-V-NA, the higher affinity substrate MeO-Suc-A-A-P-V-NA was used here, as it only became available prior to the commencement of this part of the study). Albumin, included as a control, showed almost no inhibitory effect compared with the mild inhibitory effect of α_2 -M and the strong inactivation by α_1 -PI (Figure 4.3). However, if sufficient albumin is present there is a non-specific inhibition of HLE activity with 50% inhibition seen with 6 g/l and almost 90% inhibition seen at normal serum albumin concentrations (routine laboratory reference range; 39-48 g/l), as shown in Figure 4.4. This would indicate that any free HLE in blood would be bound by the albumin present in addition to the protease inhibitors present such

as α_1 -PI and α_2 -M. However, this would not be the case in the lung where the albumin concentration is much lower (eg. alveolar lining fluid, 0.08-0.25 g/l; Baughman et al., 1983).

The complete inhibition of HLE activity (0.5 μ g present in assay cuvette) required 1.0 μ g of α_1 -PI which is consistent with the fact that HLE (MW 26,000) and α_1 -PI (MW 52,000) bind in equimolar amounts. This inhibition was independent of pH within the pH range tested, of 6-9 (data not shown; The assay conditions were as used for heparin inhibition in Section 4.5.1.4). This also supports the data that the HLE used in this study is at least 90% active (see Section 4.2). However, the weak inhibition of HLE by α_2 -M is surprising in that other studies have shown little if any inhibitory effect when using low molecular weight substrates, with one report showing activation of HLE activity (Twumasi et al., 1977).

The effect of elastin peptides on HLE may be dependent on molecular weight. A soluble high molecular weight elastin peptide preparation (α -elastin, approx. MW 70,000) strongly inhibited HLE (50% inhibition with approximately 2.6 μ g and 92% inhibition with 19 μ g) whereas, a synthetic elastin peptide (VGVAPG, MW 588) showed no significant inhibitory activity (Figure 4.3). VGVAPG is a sequence that is repeated multiple times in the elastin molecule (Sandberg et al., 1981) and has been shown to be a chemoattractant for monocytes, elastin-producing fibroblasts (Senior et al., 1984) and tumour cells (Blood et al., 1988). Another repeating sequence in the elastin molecule is VPGVG, which appears to be an essential sequence for the autoregulation of elastin synthesis (Wachi et al., 1995). This pentapeptide was also found not to inhibit HLE activity (data not shown).



Protein ug

Figure 4.3: The inhibitory effect of proteins on HLE (0.5 ug) activity with MeO-Suc-A-A-P-NA as substrate. Assay conditions as described in Section 2.4.2.2.



Albumin g/l

Figure 4.4: The inhibitory effect of albumin on HLE (0.5 ug) activity with MeO-Suc-A-A-P-V-NA as substrate. Assay conditions as described in Section 2.4.2.2



Alpha-1-antiprotease (ug)

Figure 4.5: The inhibitory effect of alpha-1-antiprotease on cathepsin G (0.1 ug) activity. Assay conditions as described in Section 2.4.3.1.



Alpha-2-macroglobulin (ug)

Figure 4.6: The inhibitory effect of alpha-2-macroglobulin on cathepsin G (0.1 ug) activity. Assay conditions as described in Section 2.4.3.1.

4.3.2 INHIBITION OF CATHEPSIN G

 α_1 -PI and α_2 -M were evaluated for their ability to inhibit cathepsin G (0.1 µg), with n-Succinyl-(L-alanine)₂-proline-L-phenylalanine-p-nitroanilide as substrate, with an assay described in Section 2.4.3.1. α_1 -PI is a very effective inhibitor of cathepsin G with 50% inhibition seen with approximately 0.075 µg (Figure 4.5). However, unlike HLE (Figure 4.3), complete inhibition of cathepsin G is not seen (10% residual activity found, Figure 4.5).

 α_2 -M is also a very effective inhibitor towards cathepsin G (50% inhibition seen with approximately 6.0 µg, Figure 4.6), which contrasts with its relative ineffectiveness towards HLE (Figure 4.3). As is the case for α_1 -PI inhibition of cathepsin G, α_2 -M did not completely inhibit cathepsin G (20% residual activity found, Figure 4.6).

4.4 HUMAN LEUKOCYTE ELASTASE SPECIFIC SYNTHETIC INHIBITORS

4.4.1 INHIBITION OF HUMAN LEUKOCYTE ELASTASE WITH MeO-Suc-A-A-P-V-NA AS SUBSTRATE.

The effectiveness of two potent synthetic inhibitors (ICI 186756 & 200355) which selectively inhibit HLE was evaluated (assay as in Section 2.4.2.2). ICI 186756, a low molecular weight aldehyde inhibitor, and ICI 200355, a peptide trifluoromethyl ketone inhibitor were found to be very effective inhibitors of HLE (Figure 4.7, 50% inhibition of 0.5 μ g HLE was achieved with approx. 0.065 μ g and 0.035 μ g respectively). ICI 200355 was the more effective inhibitor with complete inhibition of HLE activity seen with less than 0.1 μ g whereas greater than 1.5 μ g was required in the case of ICI 186756.



ICI inhibitor concentration (ug)

Figure 4.7: The inhibition of HLE with the ICI inhibitors 186756 and 200355. Assay conditions with MeO-Suc-A-A-P-V-NA as substrate as described in Section 2.4.2.2.

Table 4.2: Complimentary inhibitory effects of ICI 200355 towards α_1 -PI and heparin inhibition of HLE. Data were shown as actual % inhibition found towards HLE with calculated inhibition that would expected to be seen shown in brackets. HLE inhibition assay with MeO-Suc-A-A-V-P-NA is as described in Section 2.4.2.2.

| | ICI 200355 (μg) | | | | | | |
|-------------------------|-----------------|-------------|-------------|-------------|--|--|--|
| | <u>0</u> | 0.006 | 0.012 | 0.024 | | | |
| No inhibitor | 0 | 6.0 | 16.7 | 33.0 | | | |
| Heparin (μg) | | | | | | | |
| 0.0365 | 12.3 | 17.7 (18.3) | 21.5 (29.0) | 52.3 (45.3) | | | |
| 0.073 | 21.1 | 26.4 (26.1) | 36.3 (37.8) | 54.6 (54.1) | | | |
| 0.156 | 41.6 | 46.8 (47.6) | 52.2 (58.3) | 70.3 (74.6) | | | |
| α ₁ -ΡΙ (μg) | | | | | | | |
| 0.085 | 7.4 | 16.2 (13.1) | 26.9 (24.1) | 52.4 (40.4) | | | |
| 0.17 | 15.9 | 23.7 (21.9) | 36.6 (32.6) | 57.0 (48.9) | | | |
| 0.34 | 31.0 | 40.2 (37.0) | 51.3 (47.7) | 72.0 (64.0) | | | |
| | | | | | | | |

For a synthetic inhibitor to be fully effective *in vivo* it must not interfere with the activity of other inhibitors towards HLE and its inhibitory effect should be additive to other inhibitors. In particular, a synthetic inhibitor should be completely compatible with the activity of α_1 -PI, the main endogenous inhibitor of HLE. For example, ICI 200355 is found, in general, to be additive in inhibitory effect towards the inhibitors α_1 -PI and heparin (Table 4.2). This was also the case for ICI 186756 (data not shown).

4.4.2 INHIBITION OF HUMAN LEUKOCYTE ELASTASE WITH HUMAN LUNG ELASTIN AS SUBSTRATE.

The more effective ICI inhibitor, 200355, was also tested for its ability to block HLE hydrolysis of a more typical endogenous substrate, human lung elastin where the released elastin peptides were measured with an ELISA method for soluble elastin peptides (see Section 2.4.2.5 for assay procedure). As shown in Figure 4.8, 0.02 μ g /ml (ie. 2 ng added) of the inhibitor was required to obtain 50% inhibition of 5 ng of HLE in this assay system. Inhibition was essentially complete at concentrations greater than 1.25 μ g/ml. This demonstrates that this inhibitor is not as effective when an endogenous substrate such as human lung elastin is used.

4.4.3 INHIBITION OF CATHEPSIN G

Two potent synthetic inhibitors of HLE, ICI 186756 & 200355, are reported to be highly specific for this enzyme. This is supported by the finding that they show almost no inhibitory activity towards cathepsin G (Figure 4.9).



Figure 4.8: ICI 200355 inhibition of HLE hydrolysis of human lung elastin (n=9). Assay as described in Section 2.4.2.4.


Figure 4.9: Effect of ICI inhibitors 186756 and 200355 on Cathepsin G activity. Assay procedure as in Section 2.4.3.1.

4.5 GLYCOSAMINOGLYCAN INHIBITION

4.5.1 INHIBITION OF HUMAN LEUKOCYTE ELASTASE WITH Suc-A-A-NA AS SUBSTRATE

The HLE inhibition assay in this Section used the method described in Section 2.4.2.1. This is an early study where Suc-A-A-NA was used as substrate before the more sensitive MeO-Suc-A-A-P-V-NA became available. Some of this material has already been published (Walsh et al., 1988; Walsh et al., 1991a). Preliminary studies showed that heparin strongly inhibited HLE but that the other leucocyte enzymes tested, cathepsin G and myeloperoxidase, and the pancreatic enzymes, pancreatic elastase, trypsin and chymotrypsin, were not inhibited by heparin concentrations to at least 250 μ g/ml (data not shown). This inhibition of HLE and cathepsin G by heparin and other GAGs was investigated further in this Section.

4.5.1.1 TYPE OF GLYCOSAMINOGLYCAN

The effect of a range of glycosaminoglycans on the hydrolysis of Suc-A-A-A-NA by HLE is shown in Fig. 4.10. The most effective inhibitor is heparin (0.22 μ g or 80 ul of 2.8 μ g/mL gave 50 % inhibition of 2.5 μ g of HLE) followed by heparan sulphate (0.64 μ g), chondroitin sulphate B (2.4 μ g), chondroitin sulphate C (>2.4 μ g), chondroitin sulphate A (>2.4 μ g), with hyaluronic acid having almost no inhibitory activity. Complete inhibition of HLE activity is not seen with any of the glycosaminoglycans. For example, the maximum inhibition for heparin and heparan sulphate of 85.5% and 70% is achieved with concentrations of 1.2 μ g and 2.5 μ g respectively.



Figure 4.10: Glycosaminoglycan inhibition of HLE activity with Suc-A-A-A-NA as substrate. Assay conditions as in Section 2.4.2.1.



Heparin (ug)





HLE activity

%

Heparan sulphate (ug)

Figure 4.12: Effect of Heparan sulphate preparations from different sources on HLE activity with Suc-A-A-NA as substrate. Method as described in Section 2.4.2.1.

4.5.1.2 SOURCE OF HEPARIN AND HEPARAN SULPHATE

Heparin and heparan sulphate are the most effective GAG inhibitors of HLE (Figure 4.10). A number of heparin and heparan sulphate preparations were assessed for any variation in their ability to inhibit HLE activity. However, with heparin preparations from different commercial and animal sources, no variation was seen in the inhibition of HLE (Figure 4.11). This was also the case with heparan sulphate preparations where little variation in inhibition was seen (Figure 4.12). The inhibition of HLE by sulphated GAGs was completely removed by pretreating the assay sample with Heparasorb (cationic triethylaminoethyl cellulose) or protamine sulphate as demonstrated by treating heparin (Fig. 4.11). These reagents interfere in the assay by effectively binding heparin or HLE (ie. The cationic HLE).

4.5.1.3. MOLECULAR WEIGHT OF HEPARIN

The effects of various low molecular weight heparins on the hydrolysis of Suc-A-A-A-NA by HLE are shown in Fig. 4.13. Opocrin 2133 (MW 4500) and Kabi 2165 (MW 4000-6000) inhibited HLE equally as well as heparin (typical average MW 15000). However Opocrin 381/1 (MW 2000) did not inhibit as well as heparin, with Opocrin 381/2 (MW 800-1000) showing little inhibition. This suggests that a molecular weight of greater than 2000 is required for maximum inhibition of HLE.



Heparin fraction (ug)

Figure 4.13: Effect of low molecular weight heparin fractions on HLE activity with Suc-A-A-NA as substrate. Assay conditions as described in Section 2.4.2.1.



Heparin (ug)

Figure 4.14: Effect of pH on heparin inhibition of HLE activity with Suc-A-A-NA as substrate. Procedure as described in Section 2.4.2.1.



NaCl Concentration (M)

Figure 4.15: Effect of NaCl concentration on heparin inhibition of HLE activity with MeO-Suc-A-A-P-V-NA as substrate. Procedure is as described in Section 2.4.2.2.

135

4.5.1.4. ASSAY CONDITIONS

The binding of heparin to HLE is essentially by electrostatic interactions, which probably involves the negatively charged sulphate groups on heparin and some of the positively charged arginine residues on the surface of HLE. The relative charge on these groups may be altered by changing the pH of the assay buffer. It was found that while the rate of hydrolysis of Suc-A-A-A-NA by HLE changed with pH (optimum at pH 8.0), the degree of inhibition of HLE by heparin was found to be independent of pH within the 6-9 range of pH studied (4.14). Thus the binding of heparin to HLE appears to be unaffected by pH in the physiological pH range, as was also found for α_1 -PI (see Section 4.2).

However, it is likely that the ionic strength of the assay buffer will influence the degree of binding of heparin to HLE. This was examined by increasing the concentration of NaCl in the assay buffer above the 0.05M used as the standard concentration (see Section 2.4.1.5). NaCl interference in the inhibition of HLE activity by heparin increases progressively at NaCl concentrations above 0.05 M until 0.35 M and above, where NaCl completely blocks the inhibition of HLE activity by heparin is a strong inhibitor of HLE activity at the physiological NaCl concentrations of 0.14 M.

Other variations in assay conditions were examined and are reported elsewhere: ie. incubation time of GAG with HLE (Section 4.6); order of addition GAG, HLE and substrate (Section 4.7); presence of surfactant (chapter 5).

4.5.2 INHIBITION OF HUMAN LEUKOCYTE ELASTASE WITH HUMAN LUNG ELASTIN AS SUBSTRATE

Heparin and heparan sulphate, the most effective GAG inhibitors of the HLE hydrolysis of Suc-A-A-A-NA, also inhibited HLE activity towards human lung elastin where the released elastin peptides were measured with an ELISA method for soluble elastin peptides (Figure 4.16; see Section 2.4.2.4 for assay procedure). Whereas heparin was the more effective inhibitor in the assay with Suc-A-A-A-NA as substrate, heparan sulphate was more effective in the insoluble lung elastin assay. This is shown in Figure 4.16 where 50% inhibition of 5 ng of HLE was obtained with 4.5 μ g of heparin/ml (ie. 0.45 μ g added) or 0.8 μ g of heparan sulphate/ml (ie. 0.08 μ g added). This demonstrates that GAGs are much less effective inhibitors when an endogenous substrate such as human lung elastin is used than when the soluble peptide substrate, Suc-A-A-NA, is used (see Section 4.5.1.1). Like the inhibition seen using Suc-A-A-A-NA as substrate, complete inhibition of HLE activity was not seen; the highest GAG concentration used, 200 μ g/ml, resulted in 85-95% inhibition which is similar to that obtained for heparin using Suc-A-A-NA as substrate (Figure 4.7).

To exclude the possibility that glycosaminoglycans may be directly interfering in the ELISA, standard curves were established in the presence and absence of heparin and heparan sulphate. As can be seen in Fig. 4.17, heparin did not cause a direct interference in the ELISA despite the fact that heparin at concentrations of 2 and 8 μ g/ml significantly inhibited HLE activity towards lung elastin (Fig. 4.16). This was also the case for heparan sulphate (data not shown).



Glycosaminoglycans (ug/ml)

Figure 4.16: Effect of heparin and heparan sulphate on HLE hydrolysis of insoluble human lung elastin (n=12). Procedure is as described in Section 2.4.2.4.



Elastin derived peptides (ng/ml)

Figure 4.17: Effect of addition of heparin to the ELISA standard curve for elastin derived peptides (n=3). Procedure is as described in Section 2.4.4.



Figure 4.18: Effect of heparin and heparan sulphate on cathepsin G activity. Assay conditions as described in 2.4.3.1.

There was also the possibility that the size of the elastin-derived peptides generated by HLE activity towards lung elastin could change in the presence of heparin or heparan sulphate and that the different size peptides might not be recognized by the antibody in the ELISA. To examine this possibility, elastin peptides were prepared by HLE activity on elastin, both in the presence and absence of heparin, and were then separated according to size on a HPLC gel filtration column (see Section 2.3.13). The major peak, both in the presence and absence of heparin, was found to be of MW 4500 with shoulder peaks of MW 3500 and MW 9500 (data not shown). There was no evidence of other peaks or a shift in the molecular weight of the major and shoulder peaks. The peaks prepared with heparin were smaller than that without heparin, as would be expected from the heparin inhibition of HLE activity. Similar data was found for heparan sulphate.

4.5.3 INHIBITION OF CATHEPSIN G

While sulphated GAGs, such as heparin and heparan sulphate, are effective inhibitors of HLE (Figure 4.10), they are ineffective against cathepsin G, even at concentrations as high as 1 mg/ml (Figure 4.18).

4.6 INCUBATION TIME OF INHIBITORS

4.6.1 INHIBITION OF HUMAN LEUKOCYTE ELASTASE.

The evaluation of inhibitors that bind to an enzyme usually requires that the inhibitor be incubated with the enzyme for a period to achieve maximal binding between the inhibitor and the enzyme before measuring the reduced enzyme activity. The importance of this incubation period was evaluated for five different types of inhibitors, α_1 -PI, α_2 -M, ICI 200355, ICI 186756 and heparin. Each inhibitor was incubated at 25° C for 10 seconds (the shortest incubation time available on the Cobas-Bio centrifugal analyser), 5 minutes and 15 minutes with HLE (0.5 µg) prior to the addition of the substrate, MeO-Suc-A-A-P-V-NA (see Section 2.4.2.2 for assay conditions). The inhibition plots show that the inhibitors, ICI 200355, ICI 186756 and heparin are fast acting with approximately 84%, 75% and 93% inhibition achieved after 10 seconds when compared to the same concentration of inhibitors respectively that gave 50% inhibition on each plot with 15 minutes incubation (Figures 4.19, 4.20). However, α_1 -PI was not as rapid with 42% inhibition point (Figure 4.21). α_1 -PI required at least 15 minutes for maximum inhibition (figure 4.21) unlike the other inhibitors tested, where maximum inhibition was achieved after 5 minutes (Figures 4.19, 4.20). However, in the presence of an excess of α_1 -PI, ICI 200355, ICI 186756 or heparin, incubation time was not a factor with complete inhibition seen in every case. α_2 -M, like α_1 -PI, required 15 minutes for maximum inhibition seen in every case. α_2 -M, like inhibitor with only 65% inhibition of HLE (0.5 µg) achieved with 30 µg of α_2 -M (Figure 4.22).



Inhibitor (ug)

Figure 4.19: Effect of incubation time on ICI 186756 and 200355 inhibition of HLE with MeO-Suc-A-A-P-V-NA as substrate. Assay conditions as described in Section 2.4.2.2.



Glycosaminoglycan (ug)

Figure 4.20: Effect of incubation time on heparin inhibition of HLE with MeO-Suc-A-A-V-P-NA as substrate. Assay conditions as described in Section 2.4.2.2.



Alpha-1-antitrypsin (ug)

Figure 4.21: Effect of incubation time on alpha-1-antiprotease inhibition of HLE with MeO-Suc-A-A-P-V-NA as substrate. Assay as described in Section 2.4.1.6.



Alpha-2-macroglobulin (ug)

Figure 4.22: Effect of incubation time on alpha-2-macroglobulin of HLE with MeO-Suc-A-A-P-V-NA as substrate. Assay conditions as described in Section 2.4.2.2.



Alpha-1-antiprotease ug

Figure 4.23: Effect of incubation time on alpha-1-antitrypsin inhibition of cathepsin G. Assay conditions as described in Section 2.4.3.1.



% Cathepsin G activity

Alpha-2-macroglobulin (ug)

Figure 4.24: Effect of incubation time on alpha-2-macroglobulin inhibition of cathepsin G. Assay conditions as described in Section 2.4.3.1.

4.6.2 INHIBITION OF CATHEPSIN G

 α_1 -PI and α_2 -M were the only two inhibitors evaluated that were effective against this enzyme (Section 4.3.2). The inhibition plots show that α_1 -PI is fast acting with approximately 75% inhibition achieved after 10 seconds, when compared to the same concentration of α_1 -PI that gave 50% inhibition with 15 minutes incubation (Figure 4.23). α_2 -M, on the other hand, was a slower acting inhibitor towards cathepsin G achieving only 33% inhibition after 10 seconds, when compared to the same concentration of α_2 -M that gave 50% inhibition with 15 minutes incubation of α_2 -M that gave 50% inhibition after 10 seconds, when compared to the same concentration of α_2 -M that gave 50% inhibition with 15 minutes incubation (Figure 4.24). In the presence of an excess of α_1 -PI, incubation time was not a factor with complete incubation seen. This was not the case with α_2 -M until amounts greater than 2 mg/ml were added.

4.7 ORDER OF ADDITION OF INHIBITOR, ENZYME AND SUBSTRATE

To investigate the interaction of the inhibitors, α_1 -PI, ICI 200355 and heparin, with both HLE and the substrate, MeO-Suc-A-A-P-V-NA, the order of addition of reagents in the assay was varied, using a modified assay with equal volumes of inhibitor, enzyme, and substrate as described in Section 2.4.2.3. The assay was run three ways with the order of addition as follows:-Type 1: enzyme + inhibitor + substrate; Type 2: inhibitor + substrate + enzyme; Type 3: enzyme + substrate + inhibitor. The inhibition plots using the three assay variations are shown as Figures 4.25, 4.26 and 4.27 respectively for α_1 -PI, ICI 200355 and heparin. Both α_1 -PI and heparin are relatively unaffected by the order of addition of reagents, whereas maximum inhibition with ICI 200355 is only seen where the inhibitor is preincubated with HLE. Thus unlike α_1 -PI and heparin, ICI 200355 is not as an effective inhibitor as it is unable to effectively bind to and inhibit HLE, once HLE is bound to the substrate (Type 3 assay, Figure 4.26). ICI 200355 also appears to have a lower affinity for binding to HLE than the substrate, MeO-Suc-A-A-P-V-NA, when the inhibitor and the substrate are added simultaneously to the enzyme (Type 2 assay, Figure 4.26).

4.8 INTERACTION OF HUMAN LEUKOCYTE ELASTASE AND CATHEPSIN G.

Previous reports have suggested HLE activity toward elastin was enhanced in the presence of cathepsin G (Boudier et al., 1981; Reilly et al.,1984). In this Section the possible interaction of HLE and cathepsin G was investigated using soluble peptide substrates with MeO-Suc-A-A-P-V-NA as the substrate in the assay for HLE activity (described in Section 2.4.2.1) and n-Succinyl-(L-alanine)₂-proline-L-phenylalanine-p-nitroanilide as substrate in the assay for cathepsin G activity (described in Section 2.4.3.1). In either assay the other enzyme was added in varying amounts as the sample. No activation or inhibition of either HLE or cathepsin G was observed in the presence of the other enzyme, as shown in Figures 4.28 and 4.29.



Alpha-1-antitrypsin (ug)

Figure 4.25: Effect of the order of addition of assay reagents on alpha-1-antitrypsin inhibition of HLE. Type 1: enzyme(E) + inhibitor(I) + Substrate (S); type 2: I + S + E; type 3: E + S + I. Assay conditions as described in Section 2.4.2.3.



HLE activity

%

ICI 200355 (ug)

Figure 4.26: Effect of the order of addition of assay reagents on ICI 200355 inhibition of HLE. Type 1: enzyme (E) + inhibitor (I) + substrate (S); type 2: I + S + E; type 3: E + S + I. Assay conditions as described in Section 2.4.2.3.

152



Heparin (ug)

Figure 4.27: Effect of order of addition of assay reagents on heparin inhibition of HLE. Type1: enzyme (E) + inhibitor (I) + substrate (S); type 2: I + S + E; type 3: E + S + I. Assay conditions as described in Section 2.4.2.3.

4.9 COMPARISON OF EFFECTIVENESS OF INHIBITORS

A number of inhibitors were compared with the endogenous inhibitors, α_1 -PI and α_2 -M, for their ability to inhibit HLE activity with the synthetic low molecular weight substrate, MeO-Suc-A-A-P-V-NA. At inhibitor concentrations less than 8.8 μ g /ml (2.5 μ g added, Figure 4.30), α_2 -M did not significantly inhibit HLE and was no more effective than albumin as an inhibitor. Elastin peptides were also effective at these low concentrations although not to the same extent as the specific endogenous HLE inhibitor, α_1 -PI. The synthetic inhibitor, ICI 200355 is the most effective inhibitor, with heparin being equally as effective to approximately 25% residual HLE activity. However if the inhibition is expressed as the number of moles of inhibitor that are required to inhibit 50% of the activity (IC50) of one mole of HLE (Table 4.3), then heparin can be considered to be the most effective inhibitor, although with the proviso that complete inhibition is not found.



Figure 4.28: Effect of cathepsin G on HLE activity with MeO-Suc-A-A-P-V-NA as substrate. Procedure is as described in Section 2.4.2.2.



Figure 4.29: Effect of HLE on cathepsin G activity. Assay procedure as in Section 2.4.3.1.



activity

HLE

%

- → Heparin
- → ICI 200355

Figure 4.30: Comparison of inhibitors towards HLE (0.5 ug) activity with MeO-Suc-A-A-P-V-NA as substrate. Assay conditions as described in Section 2.4.2.2.

Table 4.3: COMPARISON OF INHIBITORS OF HLE (0.5 ug of HLE added, av. MW of 30,000)

| Inhibitor | MW | IC50 (ug) | Inhibitor/HLE (molar ratio) |
|-------------------|-------------|--------------|--------------------------------|
| ICI 200355 | 732 | 0.035 | 2.87 |
| Heparin | 14,000 (av) | 0.040 | 0.14 |
| α_1 -PI | 52,000 | 0.5 | 0.58 |
| Elastin peptides | 70,000 (av) | 2.6 | 2.23 |
| α ₂ -Μ | 725,000 | 23.0 | 1.90 |
| Albumin | 65,000 | 90.0 | 83.11 |

4.10 **DISCUSSION**

The primary function of α_1 -PI is considered to be to control the activity of HLE, since the k_{ass} is more than tenfold higher for this enzyme than that for other proteinases (Beatty et al., 1980). As reported elsewhere (Travis and Salvesen, 1983), the effectiveness of α_1 -PI towards HLE is confirmed in this study where complete inhibition of HLE is found, with equimolar amounts of α_1 -PI (Figure 4.3). α_1 -PI is equally effective even when added after HLE is preincubated with the low molecular weight substrate, MeO-Suc-A-A-P-V-NA (Figure 4.25). However α_1 -PI required at least 15 minutes incubation with HLE, prior to addition of the substrate, to achieve maximal binding between the inhibitor and the enzyme before measuring the reduced enzyme activity (Figure 4.21). This high specificity of α_1 -PI for HLE compared to inhibition of other proteases is demonstrated by the fact that α_1 -PI is not as effective against the other major inflammatory serine protease, cathepsin G, where incomplete inhibition was seen (10% residual activity was found, Figure 4.5). α_1 -PI is also relatively unaffected by the order of the addition of reagents, such as adding α_1 -PI after the peptide substrate, MeO-Suc-A-A-P-V-NA was added to HLE (Figure 4.25) although this is not seen where elastin is used as substrate as reported elsewhere (Morrison et al., 1990, Padrines and Bieth, 1991).

In this study, with the low molecular weight substrate MeO-Suc-A-A-P-V-NA, α_2 -M is a very poor inhibitor of HLE (figure 4.3). This finding is consistent with the general finding that proteinases, when complexed to α_2 -M, typically retain 80-100% of their hydrolytic activity against low molecular weight substrates (Barrett, 1981). α_2 -M also required at least 15 minutes incubation HLE, prior to the addition of the substrate MeO-Suc-A-A-P-V-NA, to achieve maximum inhibition. Together with the poor inhibition achieved (65% inhibition of 0.5 μ g of HLE with 30 μ g of α_2 -M; Figure 4.22) this result suggests that the primary role of α_2 -M may not be to control HLE activity. In one study, α_2 -M was able to activate HLE activity from 1.1 to 15.3 fold the original activity, depending on the peptide substrate used (Twusami et al., 1977). However, α_2 -M is a more effective inhibitor towards cathepsin G, although as with proteases in general using low molecular weight substrates, α_2 -M did not completely inhibit cathepsin G (20% residual activity found, Figure 4.6). α_2 -M was even slower acting towards cathepsin G than towards HLE, as shown in Figure 4.24, where only 33% inhibition of cathepsin G was found after 10 seconds incubation of α_2 -M with HLE. What role α_2 -M has in regulating HLE and cathepsin G activity, if any, is not known. Any α_2 -M - protease complexes that do form in vivo are rapidly cleared by the reticuloendothelial system (Travis and Salvesen, 1983), so that any residual proteolyic activity of the complexes is likely to be minimal. A role in the development of pulmonary emphysema is unlikely as α_2 -M, because of its size, cannot readily diffuse into the lung alveolus where it is usually at such low concentrations that it is difficult to detect in normal BAL (Mclennan and Walsh, 1995).

While cathepsin G is a relatively poor protease, its activity is not completely inhibited by the major protease inhibitors, α_1 -PI (Figure 4.5) or α_2 -M (Figure 4.6). In addition, α_1 - ACH, which appears to have a role in the regulation of cathepsin G activity, is specifically inactivated by the presence of HLE (Potempa et al., 1991). These findings suggest that cathepsin G may play a role in causing tissue damage in chronic inflammatory diseases such as emphysema where there is evidence of increased HLE activity in the lungs of emphysema patients, as discussed in chapter one of this thesis.

In this study, I demonstrate that soluble elastin peptides of high molecular weight can be effective inhibitors of HLE with over 90% inhibition found at 15 μ g/ml (a total of 9.37 μ g added, Figure 4.3) and with over 99.5% achieved at 75 μ g /ml seen with an α -elastin preparation (approx. 70,000 MW). This finding suggests that the elastin peptide preparation appears to be so tightly bound to HLE that the enzyme is essentially unavailable to act on the peptide substrate, MeO-Suc-A-A-P-V-NA. Such high affinity binding between elastin peptides and HLE is also reported to occur during elastinolysis, with the formation of an unusually stable complex between HLE and the released elastin peptides rich in the alanine/glycine and hydrophobic domains of elastin (Tyagi and Simon, 1993). An elastin peptide preparation (k-elastin) of average molecular weight of 57,000 has also been reported to be an effective inhibitor of HLE but that preparations of <10,000 MW did not show any significant inhibitory activity (Bonnaure-Mallet et al., 1995). While the elastin peptide sequences, VAGAPA and VPGVG have been suggested to have a number of biological functions (Sandberg et al., 1981; Senior et al., 1984; Blood et al., 1988; Wachi et al., 1995), the separate peptides did not show any inhibition of HLE activity in this study (Figure 4.3). While no inhibition is seen with the specific peptides used here, it is unlikely that inhibition would be seen with any other elastin peptides of such low

molecular weight as suggested by a report that elastin peptides with MW <10,000 had no major influence on elastase (Bonnaure-Mallet et al., 1995). This high affinity binding between elastin peptides and HLE reported here and by others has been suggested by Tyagi and Simon (1993) to have a regulatory role towards HLE activity at local sites of proteolysis during elastinolysis. How this proposal fits with normal remodelling of elastic tissue as distinct from the degradative mechanisms that occur in diseases such as pulmonary emphysema awaits further studies.

While a non-specific protein, such as albumin, shows almost no inhibitory effect at low concentrations (Figure 4.3), significant inhibition of HLE activity is seen if a relatively large concentration of albumin is present (50% inhibition seen with 6 g/l and almost 90% inhibition seen at normal serum albumin concentrations (routine laboratory reference range; 39-48 g/l) as shown in Figure 4.4). This would indicate that any free HLE in blood would be bound by the albumin present in addition to the protease inhibitors present such as α_1 -PI and α_2 -M. However there would not be any protective effect in the lung where the albumin concentration is much lower (eg. alveolar lining fluid, 0.08-0.25 g/L; Baughman et al., 1983). HLE has a very broad specificity allowing it to degrade a wide variety of proteins including a number of serum proteins (Bieth, 1986). Thus, it is likely that the inhibition of HLE by albumin can be explained by the large concentrations of albumin acting as a substrate and saturating all the available catalytic sites on HLE.

Two synthetic inhibitors (ICI 186756 & 200355) of HLE were found to be very effective with complete inhibition of HLE activity when MeO-Suc-A-A-P-V-NA was used as substrate (Figure 4.7). ICI 186756 & 200355 were also found to be fast acting with no significant differences noted whether the inhibitors where incubated with HLE for 10 seconds or 15 minutes prior to the addition of the peptide substrate (Figure 4.19). However, if the inhibitor (eg. ICI 200355) was not added to the enzyme first it was a less effective inhibitor of HLE (Figure 4.26) which

suggests that HLE has a lower affinity for ICI 200355 than for the substrate, MeO-Suc-A-A-P-V-NA. This type of inhibitor was less effective using human lung elastin as demonstrated with ICI 200355 (Figure 4.8) and as reported elsewhere for ICI 186756 & 200355 with bovine elastin as substrate (Williams et al., 1991; Sommerhoff et al., 1991). This reduced effectiveness of an inhibitor towards HLE having elastin as substrate compared to having a chromogenic peptide as substrate is similar to that found with heparin and heparan sulphate inhibition of HLE with Suc-A-A-NA (Figure 4.10) and human lung elastin (Figure 4.16) as substrates. This is common with a number of inhibitors due to the strong binding of HLE to insoluble elastin as has been reported by others (Morrison et al., 1990; Padrines and Bieth, 1991). Both inhibitors are reported to be highly specific for HLE with almost no activity towards cathepsin G (Williams et al., 1991; Sommerhoff et al., 1991) which was confirmed in this study (Figure 4.9). Importantly, ICI 186756 & 200355 were compatible and additive to the inhibitory activity of the main endogenous inhibitor of HLE, α_1 -PI, as well as another endogenous inhibitor, heparin, as shown for ICI 200355 in Table 4.2.

The binding of sulphated GAGs to HLE is rapid and occurs with a high affinity as shown by the binding of heparin to HLE which is relatively unaffected by incubation time of heparin with HLE prior to assay (Figure, 4.20; at least to the minimum time of 10 seconds studied) and the order of addition of assay reagents (Figure, 4.27). Evidence for ionic binding is demonstrated by the displacement of heparin from a HLE affinity column (see Section 6.2) which begins to occur at a NaCl concentration (approximately 0.4 M). This NaCl concentration is about the point (0.35 M and above) where NaCl completely blocks the inhibition of HLE activity by heparin (Figure 4.15). This agrees with another report that ionic strengths higher than 0.5 (0.4 in this study) completely abolished the inhibitory capacity of heparin towards leucocyte proteinases (Redini et al., 1988a). Alteration of pH could also change the degree of ionic binding between heparin and HLE. However, it was found that the inhibition of HLE by heparin was independent of pH, within the
6-9 pH range studied (Figure 4.14), which agrees with previous reports that inhibition is seen towards a chromogenic substrate at pH 6.0 (Marossy, 1981) and pH 8.0 (Redini et al., 1988a). However, two other studies have reported differing results, the first demonstrating inhibition only below pH 5.8 (Avila and Convit, 1976) and the second demonstrating that heparin stimulates enzyme activity towards native elastin (Lonky et al., 1978). The discrepancy in the data is most probably due to differences in assay conditions. The chromogenic assay data in this study are consistent with the most recently reported data on heparin (Drag and Petersen, 1994) and with the physiologically more relevant data obtained with human lung elastin as the substrate for HLE (Figure 4.16).

The inhibition of HLE activity is thought to occur by wrapping the linear polyanion chain of the sulphated GAG around the cationic enzyme (Baici et al., 1980). In general, maximum inhibition is seen with GAGs having long chain lengths and high sulphate content. The importance of sulphate content is demonstrated in this study where heparin, which has the highest sulphate content overall of GAGs (Lindahl and Hook, 1978), is the most effective inhibitor of HLE. This is in contrast to hyluronic acid, a GAG with no sulphate content, which is ineffective as an inhibitor (Figure 4.10). These data are consistent with other studies using desulphated and oversulphated heparin derivatives (Redini et al., 1988a) and dextran and chitosan sulphates (Lentini et al., 1985). There is evidence that the position of sulphation on the GAG chain is also important (Redini et al., 1988; Lentini et al., 1985; this study, Chapter 5). The importance of chain length is shown with various low molecular weight heparins, where preparations with a MW of 4000-6000 inhibited equally as well as heparin (average MW 15000) but preparations of MW <2000 were not as effective. These data are consistent with other studies using varying chain lengths of heparin derivatives (Redini et al., 1988a) and dextran and chitosan sulphates (Lentini et al., 1985). However, this work, which has been previously reported (Walsh et al., 1991a) is not completely consistent with the later report of Drag & Petersen (1994) where a

gradual increase in HLE inhibition was shown with heparin preparations of MW from 2300 to 12700. The difference is likely to be due to the fact that the low MW fractions used in the Drag & Petersen (1994) study were a series of defined MW fractions prepared from one original heparin preparation whereas in the present study the preparations were obtained commercially from a number of different suppliers (Section 2.2.3). In this study there is the possibility of differences in original heparin source, method of preparation and differences in sulphate content and defined MW of the commercially supplied fractions.

In this study, heparin and heparan sulphate inhibit HLE activity towards insoluble human lung elastin (Figure 4.16) which differs from one earlier report (Lonky et al., 1978) where heparin is claimed to stimulate HLE activity towards elastin. However, the data in this thesis where heparin and heparan sulphate were found to inhibit HLE using elastin as substrate is consistent with a more recent report (Moczar and Hornebeck, 1991), where heparin inhibited rat leucocyte elastase with ³H-bovine elastin as substrate. The heparin and heparan sulphate inhibition of HLE found in this thesis using insoluble human lung elastin as substrate is also consistent with the inhibition seen where chromogenic peptide substrates have been used in this thesis and other studies. The inhibition seen in the assay system with insoluble human lung elastin as substrate is a true inhibition of HLE and is not due to heparin and heparan sulphate interference in the ELISA for elastin peptides (see Section 2.4.2.4 for assay procedure). Heparin and heparan sulphate do not produce a change in the size of the elastin derived peptides, another potential cause for apparent elastin peptide concentration changes.

Heparin and heparan sulphate were found to be ineffective as inhibitors of cathepsin G (Figure 4.18) which differs from the inhibition reported in two other studies (Redini et al., 1988a; Drag and Petersen, 1944). No inhibition of cathepsin G by heparin could be demonstrated even when the assay conditions of Redini et al (1988a) were used which included a 15-30 minute

preincubation step and an assay step at 37°C. This difference could not be readily explained at the time of publication (Walsh et al., 1991a) and has not been investigated further but is most likely to be due to differences in enzyme source.

Sulphated GAGs have been shown here and in other studies to be potent anti-elastase agents, as demonstrated for heparin (Redini et al., 1988a; Walsh et al., 1988; Walsh et al., 1991a; Drag and Petersen, 1994) and for a number of heparin derivatives and other sulphated GAGs (Barg et al., 1979; Baici et al., 1980; Baici et al., 1981; Baici et al., 1984; Lentini et al., 1985; Redini et al., 1988a; Redini et al., 1988b; Moczar and Hornebeck, 1991; Walsh et al., 1991a). However it has not been determined whether endogenous sulphated GAGs have an anti-elastase role in vivo. Although the occurrence and location of sulphated GAGs *in vivo* lends some support to this role. For example, heparin and heparan sulphate are found widely in vivo on cell surfaces and as components of proteoglycans (Staprans and Felts, 1985) in cell and basement membranes, where it is thought that they may have a role in the regulation of passage of substances and cells across membrane surfaces, including the proteolytic mechanisms involved. Some specific examples, include the evidence that heparin can prevent the passage of PMNs through subendothelial basement membrane by inhibiting the selectin family of cell adhesion receptors involved with granulocyte adhesion and activation (Parekh and Edge, 1994; Nelson et al., 1993) as well as inhibiting the hydrolytic enzymes released such as heparanase (Vlodavosky et al., 1992; Bartlett et al., 1995).

Overall heparin and ICI 200355 appear to be the most potent inhibitors of HLE with an IC50 for HLE of 0.040 and 0.035 μ g respectively (Table 4.3), using MeO-Suc-A-A-P-V-NA as substrate, although heparin is only most effective as a partial inhibitor (approx. 25% residual HLE activity, Figure 4.30). Whilst ICI 200355 is a synthetic inhibitor designed to be a potent inhibitor of HLE, it is surprising that heparin appears to be a better partial inhibitor of HLE than the primary

endogenous inhibitor, α_1 -PI, which suggests that heparin may have some regulatory role towards HLE. At the IC50 point, heparin has a molar ratio of inhibitor to HLE of 0.14 in this study (Table 4.3) which indicates multiple HLE binding sites on the linear chain of heparin (approx. 7). The question as to whether there are specific high affinity binding sites for HLE along the heparin chain is examined in chapter 6.

CHAPTER FIVE

INTERACTION OF SURFACTANTS ON ELASTASE AND ELASTASE INHIBITORS

5.1 INTRODUCTION

A number of non-ionic surfactants, such as Brij 35, Triton X-100, Pluriol PE 6800 and Tween 20, have been widely used in the isolation of HLE, in assay systems using HLE and in kinetic assays for HLE (Spitznagel et al., 1974; Engelbrecht et al., 1982; Cook and Ternai, 1988a; Wenzel et al., 1990). Brij 35 has been reported to activate HLE activity towards elastin and oligopeptide substrates (Cook and Ternai, 1988a; Wenzel et al., 1990; Starkey and Barrett, 1976; Walsh et al., 1991b), but it was also found to reduce the inhibitory activity of the hydrophobic inhibitors; alkyl 2-pyrones, oleic acid and alkyl peptides (Cook and Ternai, 1988a). These observations suggest that Brij 35 binds to a hydrophobic site on the enzyme in competition with the hydrophobic inhibitors. I have described in a preliminary report that Brij 35 also modifies sulphated GAG inhibition of HLE (Walsh et al., 1991b). In contrast, the ionic detergents, sodium dodecyl sulphate and sodium linoleate have been shown to inhibit hydrolysis of Suc-A-A-NA by HLE (Finlay et al., 1982; Walsh et al., 1991b), but in one report to stimulate elastolytic activity of α_2 -macroglobulin-protease complexes (Finlay et al., 1982).

Because surfactants, such as Brij 35, are commonly added to *in vitro* HLE assay systems, it was the aim of this study to examine whether these surfactants could alter the effectiveness of a number of endogenous and exogenous HLE inhibitors. This study also examines the effectiveness of HLE inhibitors in the presence of pulmonary surfactant, a study that has relevance to the use of exogenous HLE inhibitors in the possible prevention of the development of pulmonary emphysema.



% Surfactant

Figure 5.1: Effect of surfactants on HLE activity with MeO-Suc-A-A-P-V-NA as substrate. Assay procedure as described in Section 2.4.2.2.



% Surfactant

Figure 5.2: Effect of Brij 35 and lung surfactant on HLE hydrolysis of human lung elastin. Assay as described in Section 2.4.2.4.

The range of inhibitors examined are those evaluated in chapter 4 and include α_1 -PI, α_2 -M, elastin peptides, sulphated GAGs, and the low molecular weight inhibitors, ICI 186756 and 200355.

5.2 EFFECT OF SURFACTANTS ON HUMAN LEUKOCYTE ELASTASE ACTIVITY

5.2.1 MeO-Suc-A-A-P-V-NA AS SUBSTRATE

I have routinely included Brij 35 in my HLE preparations in order to prevent autolytic degradation of the enzyme and to minimise its adsorption to the surfaces of reaction container and cuvettes. Because of reports (Starkey and Barrett, 1976; Cook and Ternai, 1988a; Wenzel et al., 1990; Walsh et al., 1991b) that Brij 35 is able to activate HLE, I investigated its effect in a HLE assay using MeO-Suc-A-A-P-V-NA as substrate and compared its effect with two other commonly used non-ionic surfactants (Triton X100 and Tween 80), an ionic surfactant (SDS) and a pulmonary surfactant preparation. As seen in Figure 5.1, all the non-ionic surfactants activated HLE activity towards MeO-Suc-A-A-P-V-NA by 20-40%. Activation still occurred at a surfactant concentration of 0.001%, the lowest concentration tested in the assay system. HLE, a cationic enzyme (pI > 9), is completely inactivated by the anionic SDS, at a concentration as low as 0.04%. Pulmonary surfactant has a mild inhibitory effect (Figure 5.1) with almost 50% inhibition achieved with a surfactant concentration of 0.5%.

5.2.2 HUMAN LUNG ELASTIN AS SUBSTRATE

The effect of Brij 35 and a lung surfactant preparation on HLE activity, with insoluble human lung elastin as substrate, was investigated as a model of possible surfactant modification of HLE mediated damage of lung elastic tissue. Brij 35 showed a slight HLE activation of 6% in the





Figure 5.3: Effect of Brij 35 on alpha-2-macroglobulin inhibition of HLE activity with MeO-Suc-A-A-P-V-NA as substrate. Assay procedure as described in Section 2.4.2.2. Data interpretation as in Section 2.5.



ICI inhibitor (ug)

Figure 5.4: Effect of Brij 35 on ICI 186756 and 200355 inhibition of HLE activity with MeO-Suc-A-A-P-V-NA as substrate. Assay as described in Section 2.4.2.2.

region of 0.05 % Brij 35 concentration but there was no significant overall activation of HLE at other concentrations (Figure 5.2), as described for the soluble peptide substrate, MeO-Suc-A-A-P-V-NA in Figure 5.1. The slight inhibitory effect of lung surfactant towards HLE activity on insoluble elastin was consistent with the its inhibitory effect towards MeO-Suc-A-A-P-V-NA (Figure 5.1).

5.3 EFFECT OF BRIJ 35 ON HUMAN LEUKOCYTE ELASTASE INHIBITORS

5.3.1 PROTEIN INHIBITION

Because of a previous report (Cook and Ternai, 1988a) of the deleterious effect of Brij 35 on some hydrophobic inhibitors, I investigated its effect on a number of HLE inhibitors and assay components. The modifying effect of Brij 35 on the ability of a range of proteins to inhibit HLE $(0.5 \ \mu g)$ with MeO-Suc-A-A-P-V-NA as substrate was evaluated. The proteins examined were albumin, α_2 -M, α_1 -PI and the elastin peptide preparations (α -elastin, VGVAPG and VPGVG), all of which had been previously examined in the absence of surfactant in Section 4.3.1. Brij 35 had no significant effect on any of these proteins and peptides (data not shown) except that it reduced the ability of α_2 -M, a non-specific protease inhibitor, to inhibit HLE (Figure 5.3), such that a 50% inhibition of HLE (0.5 μ g) required 44 μ g of α_2 -M in the presence of Brij 35 compared with 23 μ g in the absence of Brij 35 (Not all data is shown).

5.3.2 HUMAN LEUKOCYTE ELASTASE SPECIFIC SYNTHETIC INHIBITORS

The effect of Brij 35 on two potent synthetic inhibitors (ICI 186756 & 200355), which selectively inhibit HLE, was evaluated (Figure 5.4). As also shown in section 4.4.1, ICI 186756, a low molecular weight aldehyde inhibitor, and ICI 200355, a peptide trifluoromethyl ketone



Glycosaminoglycan ug

Figure 5.5: Effect of Brij 35 on heparin and chondroitin sulphate inhibition of HLE with Suc-A-A-NA as substrate. Assay procedure as described in Section 2.4.2.1.



Glycosaminoglycan ug

Figure 5.6: Effect of Brij 35 on heparin and chondroitin sulphate inhibition of HLE with MeO-Suc-A-A-P-V-NA as substrate. Assay procedure as described in Section 2.4.2.2.

inhibitor, were very effective inhibitors of HLE (50% inhibition of 0.5 μ g HLE was achieved with 0.065 μ g and 0.035 μ g respectively). The tight binding of these inhibitors was not significantly modified by the presence of Brij 35 (Figure 5.4).

5.3.3 GAG INHIBITION

The most significant effects of Brij 35 on HLE inhibitors were seen with sulphated GAGs. An example of the effects of Brij 35 on the inhibition of HLE by sulphated GAGs is shown with heparin and chondroitin sulphate using Suc-A-A-NA (Figure 5.5) or MeO-Suc-A-A-P-V-NA (Figure 5.6) as substrate. Overall, heparin is a more effective inhibitor with Suc-A-A-NA as substrate (approx. 12% residual HLE activity, Figure 5.5) than with MeO-Suc-A-A-P-V-NA (approx. 23% residual HLE activity, Figure 5.6), but is a less effective inhibitor in the presence of Brij 35 at heparin concentrations below 0.1 μ g with either substrate. Chondroitin sulphate, with MeO-Suc-A-A-P-V-NA as substrate, shows similar inhibition plots to those with heparin, although it is a less effective inhibitor than heparin (approx. 42% residual HLE activity, Figure 5.6). However, with Suc-A-A-NA as substrate, the inhibitory effect of chondroitin sulphate was drastically effected by the presence of Brij 35 (approx. 50% residual HLE activity with Brij 35 compared with 26% in its absence, Figure 5.5). Interestingly, in the absence of Brij 35, chondroitin sulphate at concentrations below 0.05 μ g is a better inhibitor of HLE than heparin at similar concentrations where Suc-A-A-NA was used as substrate in both assays.

Brij 35 was found to reduce the ability of sulphated GAGs to inhibit HLE, the effect being dependent on the type of GAG, the substrate used and the order of addition of reagents (Table 5.1) using heparin, heparan sulphate and chondroitin sulphate as examples. This table only compares the 50% inhibition points for each GAG (ng) of HLE (0.5 μ g) activity and not the maximum inhibition of HLE that can be achieved, as shown in figures 5.5 and 5.6 for heparin

Table 5.1: Summary of the effects of Brij 35 on heparin and chondroitin sulphate inhibition of HLE activity towards the substrates, Suc-Ala-Ala-Ala-NA and MeO-Suc-Ala-Ala-Pro-Val-NA.

Brij 35 was added to either the GAG (sample) or HLE (enzyme) prior to assay and compared to reagents with no Brij 35 present. The amount of GAG added to the assay to achieve a 50% inhibition of HLE activity is shown in the table as ng amounts with the factor (in brackets) indicating the increased amount required to achieve the same 50% inhibition with Brij 35 present compared to its absence. The 50% inhibition points of HLE activity were determined from plots similar to those shown in Figures 5.5 and 5.6. The HLE inhibition assays, with Suc-Ala-Ala-Ala-NA and MeO-Suc-Ala-Ala-Pro-Val-NA as substrate, are as described in Sections 2.4.2.1 and 2.4.2.2 respectively.

| GAG (ng) | Brij 35 | Suc-Ala-Ala-Ala-NA | MeO-Suc-Ala-Ala-Pro-Val-NA |
|-------------------------|---------------------------------------|--------------------------------|-------------------------------|
| Heparin | not present in sample | 45 45 (x1) | 45 45 (x1) |
| | in enzyme | 60 (x1.3) | 67 (x1.5) |
| Heparan sulphate | not present in sample | 15 43 (x2.8) | 110 110 (x1) |
| | in enzyme | 65 (x4.3) | 230 (x2.1) |
| Chondroitin sulphate | not present in sample in enzyme | 15 300 (x20) 460 (x30.6) | 52 75 (x1.4) 110 (x2.1) |



% Lung surfactant



Figure 5.7: Effect of lung surfactant on heparin, heparan sulphate and alpa-1-antiprotease inhibition of HLE with MeO-Suc-A-A-P-V-NA as substrate. Assay procedure as described in Section 2.4.2.2.

and chondroitin sulphate. All the GAGs are effective partial inhibitors of HLE in the absence of Brij 35, with heparan sulphate and chondroitin sulphate being the most effective where Suc-A-A-A-NA was used as substrate. The maximum interference of Brij 35 in the inhibition of HLE by sulphated GAGs was seen only when it was added to the HLE prior to assay. The extent of the interference ranged from a minimal effect on heparin (x1.3) to a large effect on chondroitin sulphate (x30.6) both with Suc-A-A-NA as substrate. The addition of Brij 35 to sulphated GAGs prior to assay had less of an effect, ranging from no effect on heparin (Suc-A-A-NA or MeO-Suc-A-A-P-V-NA as substrate) or heparan sulphate (MeO-Suc-A-A-P-V-NA as substrate) to a 20 fold effect on chondroitin sulphate (Suc-Ala-Ala-Ala-Ala-Ala-NA as substrate).

5.4 EFFECT OF LUNG SURFACTANT ON HUMAN LEUKOCYTE ELASTASE INHIBITORS

As shown in Section 5.2, lung surfactant has an inhibitory effect towards HLE activity. This function of lung surfactant may have a protective function in the lung alveolus to minimise elastolytic lung damage. However, this inhibitory effect of lung surfactant should ideally be additive to, and not interfere with, the main HLE inhibitor, α_1 -PI and other endogenous inhibitors, such as heparin and heparan sulphate. It was found that the inhibitory effect of lung surfactant was additive to the inhibition by α_1 -PI, but appeared to have no additional inhibitory effect on the inhibition seen with heparin and heparan sulphate (Figure 5.7).

5.5 **DISCUSSION**

While the 20-40% activation of HLE by non-ionic surfactants, found in this study, with peptide substrates is consistent with other reports (Cook and Ternai, 1988a; Wenzel et al., 1990), the degree of activation varied depending on the substrate and assay conditions used. For example, I

have found approximately 125% activation with Brij 35 (Figure 5.1) using 350 μ mol/l MeO-Suc-A-A-P-V-NA in 0.05 mol/l NaCl compared with the other reports of 111% with 170 μ mol/l with the same substrate in 0.5 mol/l NaCl (Cook and Ternai, 1988a) and 167% with Suc-Ala-Ala-Ala-NA as substrate (Wenzel et al., 1990). With human lung elastin as substrate (Figure 5.2), only a slight activation of HLE (about 106% with 0.04-0.08% Brij 35) is reported here with no activation seen at Brij 35 concentrations of greater than 0.16%. The only other report in which elastin was used as substrate demonstrated a 14% increase in elastolysis in the presence of 0.1% Brij 35 using a gel plate assay (Starkey and Barrett, 1976). The results from this thesis are consistent with earlier reports that non-ionic surfactants activate HLE activity towards peptide substrates although, as demonstrated with Brij 35, there is little, if any, activation of HLE activity towards elastin as substrate. It has been proposed that certain agents which modify HLE activity, such as surfactants, bind to a regulatory site on HLE (Lestienne and Bieth, 1980). With hydrophobic compounds, such as non-ionic surfactants, the interaction probably occurs at an unusual hydrophobic binding site near the active site of HLE (Cook and Ternai, 1988b; Bode et al., 1989).

I have found that pulmonary surfactant has a mild inhibitory effect on HLE activity towards both a peptide substrate (Figure 5.1) and elastin (Figure 5.2). This inhibitory effect does not interfere with inhibitory activity of the endogenous inhibitors, α_1 -PI, heparin and heparan sulphate. However while the lung surfactant activity is additive to α_1 -PI, the main HLE inhibitor, it appears not to be additive to heparin and heparan sulphate. As pulmonary surfactant consists of 80-90% phospholipid (Hawgood et al., 1985) it is likely that the mild inhibitory effect could be due to electrostatic binding between the anionic phosphate groups on the phospholipid and the cationic arginine groups on HLE. This finding suggests that pulmonary surfactant *in vivo* may have an additional role in that it could partially protect the alveolar surface against elastolytic enzyme induced injuries by direct inhibition of HLE. This proposal is consistent with other reports that the administration of pulmonary surfactant resulted in a significant inhibition of elastase-induced emphysema in mice (Otto-Verberne et al., 1992) and that neltenexine, a pulmonary surfactant inducing drug, showed a significant reduction in alveolar deformation in elastase-induced emphysema, when administered to rats prior to elastase treatment (Braga et al., 1995). However, HLE has also been reported to cleave surfactant proteins, which may alter surfactant function (Pison et al., 1989). Thus it would appear that replacement of the pulmonary surfactant layer on the alveolar surface is essential where HLE comes in contact with pulmonary surfactant so as to prevent elastolytic-induced injuries.

Brij 35 had little effect on the interaction of proteins and elastin peptides with HLE, except with α_2 -M (Figure 5.3) where the presence of Brij 35 required that the α_2 -M concentration be almost doubled to obtain the same inhibition of HLE (eg. 50% inhibition of HLE (0.5 µg) required 44 µg of α_2 -M in the presence of Brij 35 compared with 23µg in the absence of Brij 35). α_2 -M is a non-specific protease inhibitor with poor inhibition shown towards HLE as shown in this thesis with the low molecular weight substrate MeO-Suc-A-A-P-V-NA and in another report where α_2 -M showed little or no inhibitory activity towards a variety of proteases using low molecular weight substrates (Barrett and Starkey, 1973). Thus it is not surprising that the effectiveness of a poor inhibitor, such as α_2 -M in the presence of low molecular weight substrates, is further compromised by the presence of surfactants such as Brij 35. As discussed previously in this chapter, there is evidence that Brij 35 binds to a hydrophobic site on the enzyme. A possible mechanism is that the presence of Brij 35 on the surface of HLE may act by simply blocking or sterically hindering the binding sites on HLE required for binding to α_2 -M.

As discussed in chapter 4, the inhibition of HLE activity by sulphated GAGs is thought to occur by wrapping the linear polyanion chain of the sulphated GAG around the cationic enzyme (Baici et al., 1980). Non-ionic surfactants, such as Brij 35, probably interact with hydrophobic regions on HLE such as the unusual hydrophobic binding site near the active site of the enzyme (Cook and Ternai, 1988b; Bode et al., 1989). Thus Brij 35 binding to one of these nonpolar domains on HLE could sterically hinder the tight (essentially electrostatic) interactions between sulphated GAGs and HLE, with a consequent reduction in the inhibitory activity of sulphated GAGs towards the enzyme. It is likely that sulphated GAGs and substrates with a high affinity for HLE would be less affected by the binding of non-ionic surfactants to HLE than those sulphated GAGs and substrates with a low affinity for HLE. To test this proposal, HLE activity was measured in the presence of Brij 35 and three types of sulphated GAGs with a relatively low affinity substrate (Suc-A-A-NA) in comparison to a relatively high affinity substrate (MeO-Suc-A-A-P-V-NA). Three general conclusions can be made from the results (see Figures 5.5, 5.6 and Table 5.1) obtained from this study. The conclusions are listed as follows.

- Maximum Brij 35 interference in the sulphated GAG inhibition of HLE requires that Brij 35 be bound to HLE prior to assay.
- The HLE assay using the higher affinity substrate, MeO-Suc-A-A-P-V-NA, was less affected by Brij 35 than the assay using the lower affinity substrate, Suc-A-A-NA (a K_m of 0.14 and 3.7 mM respectively has been reported by Nakajima et al., 1979).
- 3. Sulphated GAGs (eg. heparin), which have a high affinity for HLE, generally are less affected by Brij 35 than those with a lower affinity (eg. chondroitin sulphate).

In conclusion, I have demonstrated in this chapter that non-ionic surfactants, such as Brij 35, can modulate the inhibitory effect of sulphated GAGs and α_2 -M towards HLE. However, no measurable effect of the presence of Brij 35 was found on the activity of other inhibitors evaluated which include α_1 -PI, elastin peptides, and the low molecular weight inhibitors, ICI 186756 and 200355.

The other significant finding is that elastin peptides and pulmonary surfactant both inhibit HLE activity. This suggests that they may have a role in modulating elastolytic damage at the alveolar surface and that their influence may need to be taken into account in inhibitor studies using *in vivo* model assay systems.

CHAPTER SIX

ISOLATION OF HIGH AFFINITY HEPARIN FRACTIONS TOWARDS HLE

6.1 INTRODUCTION

Heparin is the commercially purified form of a sulphated GAG that is widely distributed throughout the human body in parallel with mast cell occurrence (Nader and Dietrich, 1989). The standard clinical heparin is extracted from porcine mucosa and consists of a pool of sulphated polysaccharide chains of varying lengths with an average MW of 14 ± 1 kDa. Heparin has been shown to interact with numerous biological systems (Lane and Lindahl, 1989) including a number of PMN functions, where it has been shown to have an anti-inflammatory role. Examples of the anti-inflammatory function of heparin include reduced PMN superoxide anion production, adhesion to endothelial cells, homotypic aggregation, lysosomal enzyme activity and PMN dependent activation of platelets (Bazzoni et al., 1993, Raitjar et al., 1993) as well as the inhibition of HLE activity (Redini et al., 1988a; Walsh et al., 1988; Walsh et al., 1991a). In particular, there is evidence to suggest that heparin can prevent the passage of PMNs through subendothelial basement membrane by inhibiting the selectin family of cell adhesion receptors involved with granulocyte adhesion and activation (Parekh and Edge, 1994; Nelson et al., 1993) and by inhibiting hydrolytic enzymes such as heparanase (Vlodavosky et al., 1992; Bartlett et al., 1995) and HLE (Redini et al., 1988a; Walsh et al., 1988; Walsh et al., 1991a; Drag and Petersen, 1994) which have been implicated in the degradation of the subendothelial basement membrane. These reports suggest that the isolation or synthesis of heparin fractions, particularly with an enhanced or specific anti-inflammatory function may prove to be useful in the prevention of the inflammatory components in the development or emphysema.

While heparin binds to a large number of proteins (Jackson et al., 1991), only one specific binding site on heparin for a protein has been well characterised. This binding site on heparin is a sulphated pentasaccharide sequence that tightly binds to antithrombin resulting in the acceleration of inhibition of thrombin which forms part of the anticoagulant effect of heparin (Casu, 1990). While heparin strongly binds to HLE by essentially electrostatic interactions which probably involves the negatively charged sulphate groups on heparin and some of the positively charged guanidinium groups of 19 arginine residues located in clusters on the surface of HLE (Bode et al, 1989; Navia et al., 1989), no specific binding sequences on heparin for HLE binding have been reported. In this chapter the binding between heparin and HLE is examined to determine if there is any evidence for preferential binding sequences in heparin. In a preliminary report (Walsh et al., 1993) and as explored further in this thesis, it was found that some heparin chains in unfractionated commercial heparin have a higher affinity than others for binding to HLE which is independent of total sulphate content. It was also found that there appears to be oligosaccharide sequences in heparin with high affinity binding for HLE.

6.2 AFFINITY CHROMATOGRAPHY OF HEPARIN ON HLE-SEPHAROSE

HLE-EAH- and HLE-ECH- Sepharose 4B gels were prepared (see Section 2.3.2) to assess the degree binding of sulphated GAGs to each gel. Initial binding studies with heparin showed that it strongly binds to HLE attached through the EAH spacer arm, resisting elution even with high salt concentrations. However, there was no binding of heparin to HLE attached through the ECH spacer arm. The latter finding indicates that the positively charged groups on HLE, such as the guanidinium groups of some of the arginine residues, when bound to the ECH spacer arm, are unavailable for binding to the anionic groups on heparin, either through binding to the carboxylic

group of the ECH spacer arm on the Sepharose gel or through being sterically hindered. Thus only HLE-EAH-Sepharose was used for heparin binding studies.

As Brij 35, a non-ionic surfactant, had been shown to reduce the ability of heparin and heparan sulphate to inhibit HLE elastolytic activity (see Section 5.3.3 and Table 5.1), it was added to the Tris buffer used for affinity chromatography to reduce the high affinity binding of heparin towards HLE and to minimise possible non-specific binding of heparin towards HLE-EAH-Sepharose. Brij 35 enabled the elution of heparin from the HLE-EAH-Sepharose column with a lower salt concentration, such that all the heparin was eluted with less than 1.5 M NaCl.

Affinity chromatography of heparin was performed as described in Section 2.3.9. Four major fractions were eluted with a 0.05-2 M NaCl gradient as shown in the typical profile reproduced in Figure 6.1. The first and the last peaks were eluted with approximately 0.4 and 1.0 M NaCl respectively. A different profile was seen for heparan sulphate. Here too the most tightly bound peak was eluted at approximately 1.0 M NaCl (Figure 6.2). The displacement of heparin begins to occur at a concentration of NaCl (approximately 0.4 M) which is about the point (0.35 M and above) where NaCl completely blocks the inhibition of HLE activity by heparin (see Section 4.5.1.4 and Figure 4.15). However, the salt interference in the inhibition of HLE activity by heparin increases progressively at physiological NaCl concentrations (0.14 M) and above, well before heparin is displaced from HLE (Figure 4.15). There is minimal interference in HLE activity with increasing salt concentration in the absence of heparin, as shown in Fig. 4.15, where 11.4% inhibition of activity was seen on increasing NaCl concentration from 0.05 M to 0.36 M NaCl. No change of inhibition was seen with a further increase to 0.67 M.



Figure 6.1: Affinity chromatography of heparin on HLE-Sepharose with a NaCl gradient in the presence of Brij 35. Preparation of gel and chromatography procedure are as described in Sections 2.3.2 and 2.3.9 respectively.



Figure 6.2: Affinity chromatography of heparan sulphate on HLE-Sepharose with a NaCl gradient in the presence of Brij 35. Preparation of gel and chromatography procedure are as described in Sections 2.3.2 and 2.3.9 respectively.

The ability of the isolated heparin fractions to inhibit HLE elastolytic activity increased progressively from the first to the last eluted fraction as shown in Figure 6.3a, where heparin content of the fraction is based on uronic acid content. This inhibition pattern was also seen, where heparin content of the fraction is based on sulphated GAG content as estimated with a 1,9-dimethylmethylene blue (DMB) dye binding assay (Figure 6.3b). The most tightly bound fractions of heparin, peaks 3 and 4 in Figure 6.1, are better inhibitors of HLE than the original unfractionated commercial heparin preparation which is represented as peak 0 in Figures 6.3a and b. Thus higher affinity binding is consistent with higher enzyme inhibitory activity.

Total sulphate content of the eluted heparin fractions fell with increased binding (Figures 6.4a and b) and inhibition of HLE (Figures 6.3a and b). This is a surprising finding in view of the fact that the presence of sulphate on heparin (an anionic GAG) appears to be essential for its high affinity binding to a cationic enzyme, such as HLE (Redini et al., 1988a; Walsh et al., 1988; Walsh et al., 1991a). This finding demonstrates that maximum binding and inhibition of HLE by heparin is not simply due to the total sulphate content of the heparin.

Using [³H]-end labeled heparin it was shown that the most tightly bound fractions (3 and 4) appear to have a longer average saccharide chain length when expressed on the basis of uronic acid (Figure 6.5a) or sulphated GAG content (Figure 6.5b). Fraction 1, the least tightly bound fraction, appears to have the same average chain length as unfractionated heparin whereas surprisingly fraction 2 appears to consist of shorter chain length heparin. HPLC was performed on unfractionated and affinity separated heparin fractions as described in Section 2.3.14. The HPLC retention times found (eg. unfractionated heparin, 10.6 min.; fraction 3, 10.6 min.; fraction 4, 10.0 min.) give an approximation of the molecular size of the fractions and lend some support to this finding in that while fraction 3 did not show an apparent mean molecular weight difference from unfractionated heparin, fraction 4 was shown to have the largest average molecular



Figure 6.3a: Inhibition of HLE by unfractionated heparin (0) and heparin fractions (1-4) separated by HLE-Sepharose chromatography as shown in Figure 6.1. HLE inhibition and uronic acid assays are as decribed in Sections 2.4.2.3 and 2.4.5 respectively.



Column fraction

Figure 6.3b: Inhibition of HLE by unfractionated heparin (0) and heparin fractions (1-4) separated by HLE-Sepharose chromatography as shown in Figure 6.1. HLE inhibition and DMB dye assays are as described in Sections 2.4.2.3 and 2.4.7.



Figure 6.4a: Total sulphate content of unfractionated heparin (0) and heparin fractions (1-4) separated by HLE-Sepharose chromatography as shown in Figure 6.1. Sulphate and uronic acid assays are as described in Sections 2.4.6 and 2.4.5 respectively.



Column fraction

Figure 6.4b: Total sulphate content of unfractionated heparin and heparin fractions (1-4) separated by HLE-Sepharose chromatography as shown in Figure 6.1. Sulphate and DMB dye assays are as described in Sections 2.4.6 and 2.4.7.



Column Fraction

Figure 6.5a: Tritium content of 'end labelled' unfractionated heparin (0) and heparin fractions (1-4) separated by HLE-Sepharose chromatography as shown in Figure 6.1. Tritium labelling and uronic acid assays are as described in Sections 2.3.3 and 2.4.5.



Figure 6.5b: Tritium content of 'end labelled" unfractionated heparin (0) and heparin fractions (1-4) separated by HLE-Sepharose chromatography as shown in Figure 6.1. Tritium labelling and DMB dye assays are as described in Sections 2.3.3 and 2.4.7. weight. In conclusion, fraction 4 was shown to have a higher mean molecular weight than unfractionated heparin and the other heparin fractions. This is consistent with higher molecular weight fractions having, in general, a greater inhibitory activity towards HLE as discussed previously in chapter 4.

6.3 AFFINITY CHROMATOGRAPHY OF HEPARIN ON ANTITHROMBIN-SEPHAROSE.

The aim of this part of the study was to examine if there was any relationship between the well characterised binding site on heparin for antithrombin and high affinity binding site(s) for HLE. Heparin was separated on an antithrombin-Sepharose column into two fractions, one with no affinity for antithrombin and the remaining fraction (approx. 30%) which contains the sulphated pentasaccharide sequence responsible for the high affinity binding to antithrombin (Figure 6.6). To ensure that no high affinity heparin remained in the unbound fraction because of column overloading, the column was recycled until no heparin remained bound to the column. Overall about 70% of the unfractionated heparin was devoid of affinity for the Sepharose-bound antithrombin III which is consistent with other studies (eg. Andersson et al., 1976). No significant difference was seen between the two fractions in their ability to inhibit HLE activity (HLE inhibition/uronic acid: Low affinity fraction = 4.2 compared to High affinity fraction = 4.6). This demonstrates that the antithrombin binding sequence in heparin has no preferential binding affinity for HLE and would appear to rule out the possibility that this unique sequence might also have an additional role in binding HLE.



CPM (x1000)

Column Fraction

Figure 6.6: Affinity chromatography of heparin on antithrombin III-Sepharose. Preparation of gel and chromatography procedure in Sections 2.3.4 and 2.3.10 respectively.

6.4 ISOLATION OF HLE BINDING FRACTIONS FROM HEPARINASE DIGESTION OF HLE BOUND HEPARIN.

6.4.1 HEPARIN BOUND TO FREE HLE

In an initial series of experiments, heparin bound to HLE was digested with heparinases, following the protocol outlined in Figure 6.7 and described in Section 2.3.11.2. Briefly, at the end of the digestion period, unbound heparin fragments were removed by dialysis, leaving the remaining heparin tightly bound to HLE together with the heparinases. An apoprotin-Sepharose column was used to isolate the bound heparin fractions from HLE; a typical separation is shown in Figure 6.8. The bound heparin fractions were eluted off with 1M NaCl, then the pH was lowered to 5.0 to recover the HLE. While good separations were obtained, variable recoveries were found for heparin fractions and for HLE. This variability was found to occur at the dialysis step. It is most likely to be due to the surface absorption of the cationic HLE and the variable molecular weight cut-off of the dialysis tubing; particularly when dialysing linear molecules such as heparin. Thus further experiments were concentrated on using HLE bound to EAH-Sepharose (see below) which allowed more opportunity to control experimental conditions with good recovery of heparin fractions and the ability to recycle the HLE-EAH-Sepharose for further experiments.

6.4.2 HEPARIN BOUND TO HLE-EAH-SEPHAROSE

The digestion with heparinases of heparin bound to HLE-EAH-Sepharose, is shown diagrammatically in Figure 6.9 and is described in Section 2.3.11.1. Briefly, at the end of the digestion period, unbound heparin fragments and heparinases were eluted from the gel with buffer. The remaining heparin tightly bound to HLE, was eluted off the gel with 2M NaCl. The isolated
Figure 6.7: Outline of procedure for the digestion of heparin bound to HLE with heparin lyases

heparin + HLE

↓ 0.1 M NaCl, 0.005 M PO₄, pH 7.15

heparin-HLE

heparin lyase I & II, 24h at 30 ° C

heparin oligos.-HLE + unbound heparin oligos.

apoprotin-Sepharose

Apoprotin-Sepharose-HLE-heparin oligos. → unbound heparin oligos.

↓ 1 M NaCl

Apoprotin-Sepharose-HLE \rightarrow heparin oligos. Ψ pH 5, 1 M NaCl Ψ

Apoprotin-Sepharose + HLE

Biogel P2 desalting 12-22% PAGE



Column fraction (1 ml)

Figure 6.8: Affinity chromatography separation of heparin from HLE with apoprotin-Sepharose. The chromatography procedure and HLE assay are as described in Sections 2.3.11.2 and 2.4.2.3 respectively.

Figure 6.9: Outline of procedure for the digestion of heparin bound to HLE-Sepharose with heparin lyases

heparin + HLE-Sepharose

↓ 0.1 M NaCl, 0.005 M PO₄, pH 7.15

heparin-HLE-Sepharose

♦ heparin lyase I & II, 24h at 30 ° C

heparin oligos.-HLE-Sepharose → unbound heparin oligos.

J 2 M NaCl

HLE-Sepharose

➔ heparin oligos.

 $\mathbf{\Psi}$

Biogel P2 desalting 12-22% PAGE Figure 6.10: Gradient PAGE (12-22%) of heparinase digested heparin fractions bound to HLE-Sepharose. Lane 1, tetrasaccharide, hexasaccharide and octasacccharide markers prepared from heparin (uronic acid content, 10 μ g); lane 2, unfractionated heparin (4 μ g); lane 3 and 4, digested unbound heparin (4 and 8 μ g respectively); lane 5 and 6, residual bound heparin after digestion (5 and 12 μ g respectively); lane 7, as for lane 1. HLE-Sepharose preparation, heparinase digestion of heparin, gel preparation and electrophoresis were as described in Sections 2.3.2, 2.3.11 and 2.3.12.



a

b

С

a

b

С

Figure 6.11: Gradient PAGE (12-22%) of heparinase digested heparin fractions bound to HLE-Sepharose. Lane 1, low molecular weight heparin (Opocrin 381/1 M_r 2000, 10 µg); lane 2, low molecular weight heparin (Opocrin 381/2 M_r 850-1000, 10 µg); lane 3, tetrasaccharide, hexasaccharide and octasacccharide markers prepared from heparin (uronic acid content, 10 µg); lane 4, residual bound heparin after digestion (10 µg); lane 5 and 6, digested unbound heparin (10 and 15 µg respectively); lane 7, unfractionated heparin (4 µg); lane 8, as for lane 1. Procedure is as for Figure 6.10.



heparin fractions that were bound to HLE after heparinase digestion were then examined with gradient PAGE. The presence of one major and several minor bands of heparin oligosaccharides are shown in the bound heparin fractions from two typical experiments in Figures 6.10 (lanes 5 and 6) and 6.11 (lane 4). This pattern of bands was not seen in heparinase digests of unbound heparin. The chain length of these bands appear to be less than a tetrasaccharide when compared to the tetrasaccharide, hexasaccharide and octasaccharide markers run on the gel, although this may be dependent on their charge. For example, a more highly charged oligosaccharide would run faster and appear to have a slightly lower molecular weight. A pooled fraction of these bands was found to inhibit HLE although not to the degree as seen with intact heparin. For example when the fraction's ability to inhibit HLE activity is expressed as a ratio of uronic acid content a value of 0.23 to 0.51 was found compared to the value seen typically for unfractionated heparin of 4.2 and 0.04 to 0.09 which was seen for the digested unbound heparin control. This finding is consistent with the observations, by myself and others, that very short chain length sulphated oligosaccharides are poor inhibitors of HLE.

6.5 DISCUSSION

Brij 35, was found to reduce the ability of heparin to inhibit elastolytic activity of HLE (see Section 5.3.3 and Table 5.1). It was added to the column buffers to reduce the degree of heparin - HLE interaction, thus allowing heparin to be completely eluted from a HLE-EAH-Sepharose affinity column with a moderate salt concentration of less than 1.5 M NaCl. Brij 35 has also been reported to reduce the inhibitory activity of a number of hydrophobic inhibitors towards HLE (Cook and Ternai, 1988a). The ability of non-ionic surfactants, such as Brij 35, and non-ionic solvents, such as acetonitrile (Frommherz et al., 1991), to lower the binding affinity of heparin towards HLE suggests the involvement of hydrophobic bonds in this binding. It has been proposed that hydrophobic inhibitors of HLE may interact with hydrophobic amino acid side chains of HLE in

the extended substrate binding pocket of HLE (Tyagi and Simon, 1990). Brij 35 binding to one of these nonpolar domains on HLE could sterically hinder the proposed essentially electrostatic interactions between the sulphate groups on sulphated GAGs and some of the positively charged guanidinium groups on HLE (Bode et al., 1989; Navia et al; 1989). The flexibility of the carbohydrate backbone allows GAGs to closely fit electropositive proteins particularly in the case of highly sulphated GAGs such as heparin where the tight ionic binding is less likely to be affected by steric hindrance than less highly sulphated GAGs. This is supported by the fact that sulphated GAGs which have a high affinity for HLE (eg. heparin) generally are less affected by Brij 35 than those with a lower affinity (eg. chondroitin sulphate) as demonstrated in Table 5.1.

With a 0.05-2 M NaCl gradient, heparin was fractionated into low and high affinity fractions for HLE. A similar result was found for heparan sulphate, which is consistent with fact that it has a similar inhibitory profile to heparin in its inhibition towards HLE (Figure 4.10 and Walsh et al., 1991a). The degree of binding affinity, in general, is consistent with higher enzyme inhibitory activity (Figures 6.1 and 6.3). The NaCl interference in the inhibition of HLE activity by heparin occurs at a concentration well before NaCl starts to displace heparin from HLE (see Section 4.5.1.4 and Figure 4.15; Figure 6.1) which suggests that heparin is less tightly bound to areas important for catalytic activity than other areas on HLE.

[³H]- end labelled heparin (Figure 6.5) and HPLC retention times suggested that fraction 4, the most tightly bound fraction, has a longer average chain length compared to the other fractions and unfractionated heparin. It also suggests that longer chain lengths may contain more, high affinity binding sites and thus are more effective at binding to HLE and inhibiting elastolytic activity. This is consistent with earlier studies (Redini et al., 1988a; Walsh et al., 1991a; Fransson and Johansson, 1981; Drag and Petersen, 1994) that, in general, high molecular weight fractions of heparin and other sulphated GAGs are better inhibitors of HLE than low molecular weight fractions.

A well characterised binding sequence on heparin is the pentasaccharide sequence for antithrombin III binding. My data demonstrate that heparin fractions, with and without the antithrombin III binding site, show approximately equal binding to HLE indicating that the proposed high affinity sequences for HLE are distinct from the antithrombin III binding sequence. Whether they are similar to the antithrombin III binding site awaits detailed structural studies of the heparin saccharide sequences bound to HLE.

As demonstrated in a number of studies, the binding between the anionic heparin polymer and the cationic HLE molecule is essentially ionic in nature. However, for maximum binding of the flexible heparin chain, the positioning of sulphate groups along the heparin may be more important than the degree of sulphation. My data demonstrating a relative fall in the total sulphate content of heparin chains with increased high affinity binding to HLE (Fig. 6.4) would suggest that the distribution and spacing of the sulphate groups along the heparin chain may be important for maximum high affinity binding to HLE, implying the occurrence of high affinity binding sequences along the heparin chain. The finding supports the concept, as discussed by Casu (1990), that only a few suitably spaced anionic groups (sulphate) along the heparin chain could achieve a binding to clusters of basic groups (arginine residues) on a protein (HLE) as strong as that obtained with oversulphated polysaccharides. This specific type of binding is not readily apparent, probably because it is swamped by the overall crowding of sulphate groups along the heparin chain randomly binding to cationic groups on HLE. This specificity of binding of heparin to proteins where some sequences are preferred to others, has been reported for the interaction of relatively low-sulphated heparan sulphate species with thrombin and low-density lipoproteins (Fransson and Johansson, 1981; Fransson and Havsmark, 1982).

Preliminary gradient PAGE studies of heparin fractions, obtained from the remaining tightly bound heparin to HLE after heparinase digestion, demonstrate one major and several minor bands of heparin oligosaccharides. These are likely to contain specific binding sequences for HLE of remaining tightly bound fractions to HLE. The number of bands may reflect the likely multiple binding sites on heparin for HLE although this may not be truly representative because of possible steric hindrance or blockage of heparin binding to HLE by the attachment of the EAH spacer arm of Sepharose to HLE. A pooled fraction of these bands was found to inhibit HLE although not to the degree as seen with intact heparin which is not surprising considering their short chain length. However, what is surprising is that such short chain lengths (approx. tetrasaccharide and less) should inhibit HLE at all which could indicate binding at critical sites on HLE that impairs its catalytic activity. The relative inhibition of HLE by each band and their relative importance in the binding of HLE awaits further studies to characterise and sequence these fractions.

A study by Yamada et al. (1999), which was published after this study was completed, has reported the isolation of a number of octasaccharides from the low-sulphated region of porcine intestinal heparin after treatment with heparinase. Whether these fractions demonstrate any higher affinity binding or contain specific binding sites for HLE has at this stage not been investigated.

The isolation in this study of heparin chains with a high specificity for binding to HLE together with the demonstration of heparin oligosaccharides which tightly bind to HLE, suggests the potential for the development of HLE inhibitors for the treatment of HLE mediated inflammatory diseases such as pulmonary emphysema. Heparin has been used commercially for many years, specifically as an anticoagulant, so that the clinical introduction of a heparin derivative should be simpler than for a totally new therapy.

CHAPTER SEVEN

CONCLUDING REMARKS

7.1 SYNOPSIS AND CONCLUSIONS OF THIS THESIS

7.1.1 FUNCTIONAL α₁ - PROTEASE INHIBITOR CONCENTRATIONS IN SERUM AND BRONCHOALVEOLAR LAVAGE FLUID

The evidence from the literature suggests that α_1 -PI is the major antiprotease reacting with proteases that constitute the protease-antiprotease hypothesis for the aetiology of emphysema. The initial part of this thesis work was directed to establishing assays to determine not only whether α_1 -PI is the major antiprotease but whether it is fully functional as an inhibitor towards HLE. α_1 -PI was measured in the serum, plasma and BAL from emphysema patients and compared with those from normal volunteer adult nonsmokers and smokers, children and patients with an acute lung disease, ARDS. This work resulted in the establishment of rapid automated assays for PPE and HLE inhibition as well as an immunochemical assay for α_1 -PI. Improved reference ranges for future population studies were also determined. One of the initial aims at the outset of the project was to confirm the key finding from two reports (Gadek et al., 1979; Janoff et al; 1979) that α_1 -PI was inactivated in the BAL of smokers compared to nonsmokers, which was suggested to explain why smokers develop emphysema while nonsmokers did not. Earlier findings (Walsh et al., 1992) were confirmed in the present study, in which no difference in functional α_1 -PI activity was demonstrated in the two groups. Other workers (Stone et al., 1983; Boudier et al.; 1983; Stockley and Afford, 1984; Abboud et al., 1985) have reported similar findings. While this is the consensus view in the literature, it may not be the case in all smokers, as only 10-20% of smokers develop clinically evident emphysema. In the present study only 4 nonsmokers and 4 smokers were available for the BAL study, partly because of the difficulty in recruiting normal volunteers for the BAL procedure and meeting ethics approval requirements. Larger groups will need to be studied if significant differences are to be detected. There is also the problem that small localised differences in α_1 -PI functional ability, such as at the phagocytic-cell alveolar junction, cannot be detected reliably in the lung washings obtained with the present BAL sampling methods. Direct sampling of alveolar lining fluid from specific areas of the lung combined with more sensitive assays may be needed to resolve this issue.

Of all the subjects studied, only patients with ARDS had significant amounts of functionally inactive α_1 -PI present in the serum and BAL, although this was compensated for by the increase in the total amount of α_1 -PI present. These results are consistent with other studies of ARDS. However the inactivation of α_1 -PI that is seen in the ARDS may not be as readily detectable in emphysema due to the lesser degree of inactivation of α_1 -PI that is likely to occur in a chronic disease such as emphysema compared to the much greater degree that is seen in an acute disease such as the ARDS. The detection of inactivated α_1 -PI is also compounded by the fact that the small rates of change of disease in emphysema often occur at a more local level compared to the large rates of change of disease seen in the ARDS, which involves the whole lung. Nevertheless, an understanding of the mechanism of α_1 -PI inactivation that occurs in ARDS is likely to contribute to an understanding of the mechanism and the detection of α_1 -PI inactivation occurring in emphysema.

In this study, α_1 -PI was found to be responsible for approximately 50% of the anti-HLE activity in BAL from normal nonsmokers, smokers and ARDS. This finding suggests the presence of other inhibitors of HLE in the lower respiratory tract and also that other inhibitors of HLE may have a more important role in the lower respiratory tract, in the development of emphysema, than has been previously suggested. In some early studies, α_1 -PI was thought to account for all of the inhibitory activity in the lower respiratory tract whereas later studies have reported more than 80% (Wewers et al., 1987; Wewers, 1989) and 50% (Afford et al., 1988) α_1 -PI inhibition of HLE. The findings in this study are consistent with other recent studies, which have demonstrated the presence of HLE inhibitors other than α_1 -PI in the lower respiratory tract. In this study as much as 50% inhibition of HLE may be due to the presence of HLE inhibitors, other than α_1 -PI. This suggests that an important next step should be an investigation to determine if these other inhibitors have a role in the development of emphysema. However, because of the difficulty in obtaining volunteers and ethics approval for further BAL studies, the investigation of HLE inhibitors in BAL fluid was not carried further. The major thrust of the project was changed to an *in vitro* study of HLE inhibitors and the investigation of the interesting finding that sulphated GAGs are potent inhibitors of HLE.

7.1.2 ELASTASE INHIBITORS

Because of its broad substrate specificity, HLE can bind to and degrade many protein substrates. In this study HLE activity was found to be inhibited not only by the established antiproteases but also, to varying degrees, by other protein and non-protein compounds, such as albumin, elastin peptides, lung surfactant and heparin. Such is the binding affinity of HLE for protein that it was found that even a nonspecific protein such as albumin can inhibit HLE activity if enough is present. This suggests the possibility that HLE activity in the lung may be regulated not only by the specific antiproteases, such as α_1 -PI, but also by non-specific proteins in the alveolar lining fluid, as well as by the inhibitory effect of lung surfactant. These other protein and non-protein compounds may serve to minimise proteolytic damage to structural proteins, such as elastin. Elastin peptides are also effective inhibitors of HLE and may perform a regulatory role in minimising HLE activity at local sites of elastinolysis. The finding that HLE is strongly inhibited by heparin and heparan sulphate is significant as both components are located on cell surfaces

and are components of proteoglycans in cell and basement membranes, where they may play a regulatory role in HLE activity at the alveolar surface. Sulphated GAG's may also have a more direct role in the regulation of HLE activity as sulphated GAG's, such a heparan sulphate, are synthesised and secreted by neutrophils (Bartold et al, 1989). The role of sulphated GAGs, such as heparin and heparan sulphate, in regulating HLE activity is also suggested in animal model studies of emphysema. For example, Lucey et al. (1991) showed that HLE caused considerably less severe lesions in animal lungs than those induced by PPE with comparable *in vitro* elastolytic activity. On the basis that sulphated GAGs inhibit HLE, but not PPE, *in vitro*, an explanation for the difference in elastolytic activity is suggested by the inhibitory effect of sulphated proteoglycans in the lung towards HLE.

Sulphated GAGs were found to be rapid inhibitors of HLE, in that complete inhibition occurred in less than the minimum time (10 seconds) studied. The essentially ionic nature of the binding was demonstrated by the finding that 0.35 M NaCl and above would completely block the inhibition of HLE activity by heparin. The degree of inhibition obtained was dependent on the type of GAG. For example, the more highly sulphated GAGs, such as heparin and heparan sulphate, were the most effective inhibitors. The importance of chain length was shown by using various low molecular weight heparins where preparations with a M_r of 4000-6000 inhibited equally as well as heparin (average M_r 15000) but preparations of M_r <2000 were not as effective. The effectiveness of a particular sulphated GAG as an inhibitor of HLE also varies with the type of substrate used. For example, whereas heparin was the more effective inhibitor of HLE using a soluble chromogenic substrate, heparan sulphate was more effective with insoluble lung elastin as substrate.

In vitro assay systems often have surfactants, usually non-ionic, added to minimise non-specific interactions such as adsorption of reactants to container surfaces. In this study, as reported

elsewhere, HLE was activated significantly by non-ionic surfactants. HLE activation was confounded by the fact that non-ionic surfactants also interfere with the inhibitory effect of some inhibitors of HLE activity. Non-ionic surfactants did not affect protein inhibitors of HLE, with the exception of α_2 -M, where the relatively poor inhibitory activity was reduced by almost half. The most significant effects were found using non-ionic surfactants with sulphated GAGs. The extent of the effect depended on the order of addition of reagents as well as the type of GAG and the substrate used. In contrast, pulmonary surfactant was found not to interfere with HLE inhibitors. The modifying effects of non-ionic surfactants may be significant in interpreting HLE activity data from *in vitro* assay systems using non-ionic surfactants. In particular, the HLE inhibitory capacity of endogenous and exogenous inhibitors should be assessed in the presence of pulmonary and other surfactants to establish that they are fully functional in the presence of surfactants.

The most significant finding from this part of the study was the effectiveness of heparin as an inhibitor of HLE. While heparin did not completely inhibit HLE activity (approx. 25% residual HLE activity remains) it was more effective as a partial inhibitor of HLE than the primary endogenous inhibitor, α_1 - PI, and ICI 200355, a synthetic inhibitor designed specifically to be a potent inhibitor of HLE. This finding lends support to the proposal that heparin as a component of cell surfaces and tissue in the lung may have some regulatory role towards HLE activity. On the basis of the above findings, further studies of heparin inhibition of HLE activity were undertaken to examine whether there was any specificity in the binding of heparin to HLE.

7.1.3 EVIDENCE FOR SPECIFICITY OF INTERACTION OF HEPARIN WITH HLE

Because of its polyanionic nature, heparin has been reported in the literature to bind to a large number of proteins essentially by relatively nonspecific electrostatic interactions. However in this study, evidence is presented for some specificity of binding of heparin to HLE, a finding which has not been reported elsewhere. Significant results from this part of the study are that heparin can be fractionated into low and high affinity binding fractions for HLE. Not surprisingly, the degree of binding of these fractions for HLE activity. Heparin was fractionated into four major peaks using a salt gradient in the presence of a non-ionic surfactant on a HLE-Sepharose column. A similar elution pattern was obtained using heparan sulphate on the same column. The degree of binding is consistent with studies of low molecular weight heparin preparations in this thesis and elsewhere where inhibition of heparin activity approximately correlates with the molecular weight of the heparin preparations.

Most studies indicate that increased sulphation of GAGs results in a higher affinity binding for cationic proteins such as HLE. Heparin is one of the most highly sulphated GAGs and one of the strongest polyanions in the body, resulting in strong binding to HLE. The surprising finding from this study was that the degree of binding and inhibition of the isolated heparin fractions towards HLE did not correlate with sulphate content but rather with a relative decrease in sulphate content. This demonstrates that the binding does not depend on maximal sulphate content but rather suggests some specificity in the arrangement of the sulphate groups on heparin such as has been shown for specific binding sites on heparin for certain proteins, such as antithrombin III. This study also demonstrates that binding sites on heparin with a high affinity for HLE appear to be independent of the antithrombin III binding site. It has been suggested that the flexible

backbone of iduronic acid containing GAGs, such as heparin, allows tighter binding of the GAG to other surfaces. On the basis of the results obtained in this study, it is proposed that maximal binding can only theoretically be achieved if the sulphate spacing on the backbone is not too crowded but set at intervals defined by the cationic residues on HLE so as allow maximal binding of the flexible backbone of heparin to HLE. Some evidence for higher affinity binding in undersulphated GAGs is suggested in another study by the finding that there is a preferred binding of relatively low-sulphated heparan sulphate species with thrombin and low-density lipoproteins (Fransson and Johansson, 1981; Fransson and Havsmark, 1982).

Gradient PAGE electrophoresis has revealed the presence of one major and several minor bands of heparin oligosaccharides in the heparin fraction that remains bound to HLE after exhaustive heparinase digestion. The number of bands may reflect the likely multiple binding sites on heparin for HLE. Although the saccharide chain length of the heparin oligosaccharides bands appear to be less than a tetrasaccharide, a pool of these oligosaccharide fractions was able to inhibit HLE, albeit much less than did unfractionated intact heparin. It is likely that the heparinase digested fractions that remain bound to HLE contain specific binding sequences to HLE. A knowledge of these specific sequences should allow the synthesis of other specific oligosaccharide derived inhibitors of HLE and could lead to development of even more effective inhibitors of HLE-mediated inflammatory diseases.

7.2 FUTURE STUDIES

The most significant findings from the studies reported in this thesis are those concerning sulphated GAG inhibition of HLE activity. For example, heparan sulphate appears to be a better inhibitor of HLE using insoluble lung elastin as substrate than is heparin, which is the reverse of what is found with insoluble substrates used in this study. A more extensive study using sulphated proteoglycans, particularly those containing heparin and heparan sulphate, may give some insight as to whether heparin and heparan sulphate have a regulatory role towards HLE as components of cell and basement membranes at the lung surface.

It was demonstrated that some heparin chains in unfractionated commercial heparin have a higher affinity than others for binding to HLE. This high affinity binding was found to be independent of the relative total sulphate content of the heparin fraction. Heparinase digestion of heparin bound to HLE revealed the presence of heparin oligosaccharides with a high affinity binding for HLE. The isolation of heparin, with high-affinity fractions for HLE, as described in this thesis offers the potential to provide for a new therapy that could be used clinically to supplement the naturally occurring antiproteases in the treatment of HLE-mediated inflammatory diseases such as pulmonary emphysema. If the specific binding saccharide sequences are known it will allow the synthesis of other specific HLE inhibitors incorporating these sequences, which could lead to development of even more effective inhibitors.

Heparin has been used commercially for many years, specifically as an anticoagulant. The profile of its side effects is well documented so that clinical introduction of any heparin derivative should be simpler than for a totally new therapy using a completely synthetic HLE inhibitor. In addition to its inhibition of HLE activity in inflammation, heparin has also been shown to interrupt diverse cellular events in inflammatory disease processes and thus a heparin derivative may have wider applications as a "natural drug" to treat a range of inflammatory diseases. As the antithrombin III site appears to be separate from the HLE binding sequences it is possible that derivatives could be synthesised without anticoagulant activity or to have dual roles, as both anti-inflammatory and antithrombotic agents.

Further development of the heparin fractions prepared by heparinase digestion, as described in this thesis, has not proceeded beyond the preliminary stage because of funding and time constraints. However, I believe that the exciting findings from this project should be developed further. If I were to continue the project, the plan of investigation for future studies would be as follows. It is essential that further refinement of heparin fraction isolation procedures be undertaken to improve recoveries, as well as scaling up the procedure, so that sufficient material can be recovered for detailed characterisation and sequencing. In addition to the isolation of heparin oligosaccharide fractions after exhaustive heparinase digestion, heparin fractions should also be obtained after partial heparinase digestion. A timed study of heparinase digestion of heparin bound to HLE would give much information on lesser affinity binding sites on heparin for HLE. The heparin oligosaccharide fractions obtained after partial heparinase digestion would be of longer saccharide chain length and most likely would be better inhibitors of HLE activity as suggested by low molecular weight heparin studies in this thesis. The sequencing of the series of heparin oligosaccharide fractions obtained from a timed study of heparinase digestion of heparin bound to HLE would, in conjunction with the known sequence of HLE, allow a binding profile of heparin towards HLE to be built up. This timed study should reveal optimum heparin binding sequences for HLE to be obtained allowing the development of more effective derivatives. This stage of the work should include the lodging of patent applications on the isolated and synthesised sequences. On completion of satisfactory in vitro studies, an investigation of the effectiveness of the inhibitors in animal model studies of emphysema and other HLE associated inflammatory diseases should be undertaken. If the results from animal model studies are successful, the most promising oligosaccharide preparations could then be considered for evaluation as anti-inflammatory agents in clinical trials. The evaluation of these preparations in preventing the development of emphysema in patients at high risk of developing the disease or in slowing the progression of emphysema in patients with the early stages of the

disease should only commence if all the previous evaluations were successful because of the long trial period required.

BIBLIOGRAPHY

Abboud RT, Fera T, Richter A, Tabona MZ, Johal S. Acute effect of smoking on the functional activity of alpha₁-protease inhibitor in bronchoalveolar lavage fluid. Am Rev Respir Dis 1985; **131**: 79-85.

Abe T, Kobayashi N, Yoshimura K, Trapnell BC, Kim H, Hubbard RC, Brewer MT, Thompson RC, Crystal RG. Expression of the secretory leukoprotease inhibitor gene in epithelial cells. J Clin Invest 1991; **87**: 2207-2215.

Abrams WR., Cohen A, Damiano VV, Eliraz A, Kimbel P, Meranze R, Weinbaum G. A model of decreased functional a-1-proteinase inhibitor. Pulmonary pathology of dogs exposed to chloramine-T. J. Clin Invest 1981; **68**: 1132-1139.

Afford SC, Burnett D, Campbell EJ, Cury JD, Stockley RA. The assessment of α_1 -proteinase inhibitor form and function in lung fluid from healthy subjects. Biol Chem Hoppe Seyler 1988; **369:** 1065-1074.

Akers S, Kucich U, Swartz M, Rosen G, Glass M, Rosenbloom J, Kimbel P, Weinbaum G. Specificity and sensitivity of the assay for elastin-derived peptides in chronic obstructive pulmonary disease. Am Rev Respir Dis 1992; **145**: 1077-1081.

Al-Hakim A, Linhardt RJ. Electrophoresis and detection of nanogram quantities of exogenous and endogenous glycosaminoglycans in biological fluids. Appl Theor Electrophor 1991; 1: 305-312.

American Thoracic Society: Chronic bronchitis, asthma, and pulmonary emphysema. A statement by the committee on diagnostic standards for nontuberculous diseases. Am Rev Respir Dis 1962; **85:** 762-768.

American Thoracic Society. Standards for the diagnosis and care of patients with chronic obstructive pulmonary disease (COPD) and asthma. Am Rev Respir Dis 1987; **136**: 225-244.

Andersson LO, Barrowcliffe E, Holmer E, Johnson EA, Sims GEC. Anticoagulant properties of heparin fractionated by affinity chromatography on matrix-bound antithrombin III and by gel filtration. Thrombosis Research 1976; **9:** 575-583.

Aronsen KF, Ekelund G, Kindmark CO, Laurell CB. Sequential changes of plasma proteins after surgical trauma. Scand J Clin Lab Invest 1972; **29:** 127-136.

Avila JL, Convit J. Physicochemical characteristics of the glycosaminoglycan-lysosomal enzyme interaction in vitro. Biochem J 1976; **160**: 129-136.

Auerbach O, Hammond EC, Kirman D, Garfinkel L. Emphysema produced in dogs by cigarette smoking. JAMA 1967; **199:** 241-246.

Baggioloni M, Dewald B, Waltz A. Interlekin-8 and related chemotactic cytokines. In: Gallin, JI, Goldstein, IM, Snyderman, R. eds. Inflammation: basic principles and clinical correlates. New York: Raven Press, Ltd., 1992: 247-263.

Baici A, Salgam P, Fehr K, Boni A. Inhibition of human elastase from polymorphonuclear leucocytes by a glycosaminoglycan polysulfate (Arteparon). Biochem Pharmacol 1980; **29:** 1723-1727.

Baici A, Salgam P, Fehr K, Boni A. Inhibition of elastase polymorphonuclear leucocytes by gold thiomalate and pentosan polysulphate (SP54^R). Biochem Pharmacol 1981; **30**: 703-711.

Baici A, Bradamante P. Interactions between human leucocyte elastase and chondroitin sulphate. Chem Biol Interact 1984; **51:** 1-11.

Bainton DF. Developmental biology of neutrophils and eosinophils. In: Gallin, JI, Goldstein, IM, Snyderman, R. eds. Inflammation: basic principles and clinical correlates. New York: Raven Press, Ltd., 1992: 303-324.

Barg WF, Englert ME, Buermann CW, Oronsky AL, Kerwar SS. Studies on the effect of pentosan polysulphate on proteoglycan degradation by leukocyte neutral proteases. Biochem Pharmacol 1979; **28**: 2639-2643.

Barker AF, Siemsen F, Pasley D, D'Silva R, Buist AS. Replacement therapy for hereditary alpha-1antitrypsin deficiency. A program for long term administration. Chest 1994; **105**: 1406-1410.

Barker DJ, Godfrey KM, Fall C, Osmond C, Winter PD, Shaheen SO. Relation of birth weight and childhood respiratory infection to adult lung function and death from chronic obstructive airways disease. BMJ 1991; **303**: 671-675.

Barrett AJ. Alpha-2-macroglobulin. Methods Enzymol 1981; 80: 737-754.

Barrett AJ, Starkey P M. The interaction of α_2 -macroglobulin with proteinases. Characteristics and specificity of the reaction, and a hypothesis concerning its molecular mechanism. Biochem J 1973; **133**: 709-724.

Bartlett GR. Phosphorous assay in column chromatography. J Biol Chem 1959; 234: 466-471.

Bartlett MR, Cowden WB, Parish CR. Differential effects of the anti-inflammatory compounds heparin, mannose-6-phosphate, and castanospermine on degradation of the vascular basement membrane by leukocytes, endothelial cells, and platelets. J Leukocyte Biol 1995; **57**: 207-213.

Bartold PM, Harkin DG, Bignold LP. Proteoglycans synthesized by human polymorphonuclear leucocytes in vitro. Immunol Cell Biol 1989; **67**: 9-17.

Bascom R. Differential susceptibility to tobacco smoke: possible mechanisms. Pharmacogenetics 1991; 1: 102-106.

Baughman RP, Bosken CH, Loudin RG, Hurtibise P, Wessler T. Quantitation of bronchoalveolar lavage with methylene blue. Am Rev Respir Dis 1983; **128**: 266-270.

Bazzoni G, Beltran-Nunez AB, Mascellani G, Bianchini P, Dejana E, Del Maschio A. Effect of heparin, dermatan sulphate, and related oligo-derivatives on human polymorphonuclear leukocyte function. J Lab Clin Med 1993; **121**: 268-275.

Beatty K, Bieth J, Travis J. Kinetics of association of serine proteinases with native and oxidised α -1-proteinase inhibitor and α -1-antichymotrypsin. J Biol Chem 1980; **255**: 3931-3934.

Beatty K, Robertie P, Senior R M, Travis J. Determination of oxidised alpha-1-proteinase inhibitor in serum. J Lab Clin Med 1982; 100: 186-192.

Bell D, Haseman J, Spock A, McLennan G, Hook G. Plasma proteins of the bronchoalveolar surface of the lungs of smokers and non-smokers. Am Rev Resp Dis 1981; **124:** 72-79.

Berglund DJ, Abbey DE, Lebowitz MD, Knutsen SF, McDonnell WF. Respiratory symptoms and pulmonary function in an elderly nonsmoking population. Chest 1999; **115**: 49-59.

Bergqvist U, Samuelsson. G, Uvnas B. Chemical composition of basophil granules from isolated rat mast cells. Acta Physiol Scand 1971; 83: 362-72.

Betsuyaku T, Yoshioka A, Nishimura M, Miyamoto K, Kondo T, Kawakami Y. Neutrophil elastase associated with alveolar macrophages from older volunteers. Am J Respir Crit Care Med 1994; **151:** 436-442.

Betsuyaku T, Nishimura M, Yoshioka A, Takeyabu K, Miyamoto K, Kawakami Y. Elastin-derived peptides and neutrophil elastase in bronchoalveolar lavage fluid. Am J Respir Crit Care Med 1996; **154:** 720-724.

Bieth JG. Elastase: catalytic and biological properties. In: Mecham RP, ed. Biology of extracellular matrix. Regulation of matrix accumulation. New York: Academic Press, 1986, 217-320.

Birkedal-Hansen H, Moore WGI, Bodden MK, Windsor LJ, Birkedal-Hansen B, DeCarlo A, Engler, JA. Matrix metalloproteinases: A review. Crit Rev Oral Biol Med 1993; **4**: 197-250.

Blackwood CE, Hosannah Y, Perman E, Keller S, Mandl I. Experimental emphysema in rats: elastolytic titer of inducing enzyme as determinant of the Respironse. Proc Soc Exp Biol Med 1973; 144: 450-454.

Blackwood RA, Moret J, Mandl I, Turino GM. Emphysema induced by intravenously administered endotoxin in an alpha₁ -antitrypsin deficent rat model. Am Rev Respir Dis 1984; **130**: 231-236.

Blood CH, Sasse J, Brodt B, Zetter BR. Identification of a tumour cell receptor for VGVAPG, an elastin derived chemotactic peptide. J Cell Biol 1988; **107**: 1987-1993.

Blumenkrantz N, Asboe-Hansen G. A new method for quantitative determination of uronic acids. Anal Biochem 1973; **54**: 484-489.

Bode W, Meyer E Jr., Powers JC. Human leukocyte and porcine pancreatic elastase: X-ray crystal structures, mechanism, substrate specificity, and mechanism-based inhibitors. Biochemistry 1989; **28**: 1951-1963.

Bonnaure-Mallet M, Godeau G, Tixier JM, Lebon-Pelletier P, Chambon Y, Robert L, Hornebeck E. Elastin derived peptides protect elastic fibres degradation by human neutrophil elastase: in vitro and in vivo studies using a mechanically induced rat gingival inflammatory model. J Periodont Res 1995; **30**: 58-65.

Bonney RJ, Ashe B, Maycock A, Dellea P, Hand K, Osinga D, Fletcher D, Mumford R, Davies P, Frankenfield D, Nolan T, Scaeffer L, Hagmann W, Finke P, Shah S, Dorn C, Doherty J. Pharmacological profile of the substituted beta-lactam L-659, 286: A member of a new class of human PMN elastase inhibitors. J Cell Biochem 1989; **39**: 47-53.

Boudier C, Holle C, Bieth JG. Stimulation of elastolytic activity of leukocyte elastase by leukocyte cathepsin G. J Biol Chem 1981; **256**: 10256-10258.

Boudier C, Pelletier A, Pauli, G, Bieth JG. The functional activity of a₁-proteinase inhibitor in broncho-alveolar lavage fluids from healthy human smokers and non-smokers. Clin Chim Acta 1983; **132**: 309-315.

Boudier C, Bieth, JG. Mucus proteinase inhibitor: a fast acting inhibitor of leukocyte elastase. Biochim Biophys Acta 1989; **995**: 36-41.

Boudier C, Godeau G, Hornebeck W, Robert L, Bieth JG. The elastolytic activity of cathepsin G: an ex vivo study with dermal elastin. Am J Respir Cell Mol Biol 1991; **4:** 497-503.

Boyd CD, Pierce RA, Schwarzbauer JE, Doege K, Sandell LJ. Alternate exon usage is a commonly used mechanism for increasing coding diversity within genes coding for extracellular matrix proteins. Matrix 1993; **13**: 457-469.

Braga PC, Piatti G, Limoli A, Dal-Sasso M, Maci S. Neltenexine: morphological investigation of protection against elastase-induced emphysema in rats. Drugs Exp Clin Res 1995; **21:** 51-57.

Brantly M, Nukiwa T, Crystal RG. Molecular basis of alpha-1-antitrypsin deficiency. Am J Med 1988; **84** (suppl 6A): 13-31.

Bridges RB, Kimmel DA, Wyatt RJ, Rehm SR. Serum antiproteases in smokers and nonsmokers. Relationships to smoking status and pulmonary function. Am Rev Respir Dis 1985; **132**: 1162-1169.

Brissenden JE, Cox DW. Alpha-2-macroglobulin in patients with obstructive lung disease, with and without alpha-1antitrypsin deficiency. Clin Chem Acta 1983; **128**: 241-248.

Bruce MC, Poncz L, Klinger JD, Stern RC, Tomashefski JF Jr., Dearborn DG. Biochemical and pathological evidence for proteolytic destruction of lung connective tissue in cystic fibrosis. Am Rev Respir Dis 1985; **132**: 529-535.

Burnett D, Afford SC, Campbell EJ, Rios-Mollineda RA, Buttle DJ, Stockley RA. Evidence for lipid-associated serine proteases and metalloproteases in human bronchoalveolar lavage fluid. Clin Sci 1988; **75:** 601-607.

Burrows, B Knudson, RJ, Cline, MG, et al. Quantitative relationships between cigarette smoking and ventilatory function. Am Rev Respir Dis 1977; **115**: 195-205.

Campbell EJ. Human leukocyte elastase, cathepsin G, and lactoferrin: a family of neutrophil granule glycoproteins which bind to an alveolar macrophage receptor. Proc Natl Acad Sci 1982; **79:** 6941-6945.

Campbell EJ. Preventive therapy of emphysema. Lessons from the elastase model. Am Rev Respir Dis 1986; **134**: 435-437

Campbell EJ, White RR, Senior RM, Rodriguez RJ, Kuhn, C. Receptor-mediated binding and internalization of leukocyte elastase by alveolar macrophages in vitro. J Clin Invest 1979; **64**: 824-833.

Campbell EJ, Senior RM, McDonald JA, Cox DL. Proteolysis by neutrophils. Relative importance of cell-substrate contact and oxidative inactivation of proteinase inhibitors in vitro. J Clin Invest 1982; **70**: 845-852.

Campbell E.J, Endicott SK, Rios-Mollineda RA. Assessment of oxidation of alpha-1-proteinase in broncho-alveolar lining fluid by monoclonal assay: comparison of smokers and non-smokers. Am Rev Respir Dis 1987; **135**: A156.

Campbell EJ, Campbell MA. Pericellular proteolysis by neutrophils in the presence of proteinase inhibitors: effects of substrate opsonization. J Cell Biol 1988; **106**: 667-676.

Capodici C, Muthukumaran G, Amoruso MA, Berg RA. Activation of neutrophil collagenase by cathepsin G. Inflammation 1989; **13:** 245-258.

Carlson JA, Rogers BB, Sifers RN, Hawkins HK, Finegold MJ, Woo SLC. Multiple tissues express α1-antitrypsin in transgenic mice and man. J Clin Invest 1988; **82**: 26-36.

Carp H, Janoff A. Possible mechanisms of emphysema in smokers. In vitro suppression of serum elastase-inhibitory capacity by fresh cigarette smoke and its prevention by anti-oxidants. Am Rev Respir Dis 1978; **118:** 617-621.

Carp H, Miller F, Hoidal J, Janoff A. Potential mechanism of emphysema: alpha-1-proteinase inhibitor purified from lungs of cigarette smokers contains oxidised methionine and has decreased elastase inhibitory capacity. Proc Natl Acad Sci USA. 1982; **79**: 2041-2045.

Carrell RW. α1-Antitrypsin: molecular pathology, leucocytes, and tissue damage. J Clin Invest 1986; **78**: 1427-1431.

Carter HW, Frasca JM, Auerbach O. The morphological changes produced in the distal Respiratory system of smoking dogs: A correlative optical, transmission and scanning electron microscopy study. In: Scanning electron microscopy, Part 5.Proceedings of the workshop on advances in biomedical applications of the SEM. 1976; O. Johari, R.P. Becker, eds. pp 226-232. IIT Research Institute, Chigaco Illinois.

Casolaro MA, Fells G, Wewers M, Pierce JE, Ogushi F, Hubbard R, Sellers S, Forstrom J, Lyons D, Kawasaki G, et al. Aumentation of lung neutrophil elastase capacity with recombinant human alpha-1-antitrypsin, J Appl Physiol 1987; **63**: 2015-2023.

Casu B, Petitou M, Provasoli M, Sinay P. Conformational flexibility : a new concept for explaining binding and biological properties of iduronic acid-containing glycosaminoglycans. Trends Biochem Sci 1988; **13**: 221-225.

Casu B. Heparin structure. Haemostasis 1990; 20 (suppl. 1): 62-73.

Caughey GH. Serine proteinases of mast cells and leukocyte granules. A league of their own. Am J Crit Care Med 1994; **150:** S138-S142.

Cawston TE, Mercer E. Preferential binding of collagenase to α_2 -macroglobulin in the presence of tissue inhibitors of metalloproteinases. FEBS Lett 1986; **209:** 9-12.

Chang, JC, Lesser, M, Yoo, OH, Orlowski, M. Increased cathepsin B-like activity in alveolar macrophages and bronchoalveolar lavage fluid from smokers. Am Rev Respir Dis 1986; **134:** 538-541.

Chang, JC, Yoo, OH, Lesser, M. Cathepsin D activity is increased in alveolar macrophages and bronchoalveolar lavage fluid of smokers. Am Rev Respir Dis 1989; **140**: 958-960.

Chapman Jr. HA, Stone OL. Comparison of live human neutrophil and alveolar macrophage elastolytic activity in vitro. J Clin Invest 1984; 74: 1693-1700.

Chapman Jr. HA, Munger JS, Shi GP. The role of thiol proteases in tissue injury and remodelling. Am J Respir Crit Care Med 1994; **150**: S155-S159.

Chen BH, Hong CJ, Pandey MR, Smith KR. Indoor air pollution in developing countries. World Health Stat Q 1990; **43**: 127-138.

Chitano P, Hosselet JJ, Mapp CE, Fabbri LM. Effect of oxidant air pollutants on the Respiratory system: insights from experimental and animal research. Eur Respir J 1995; 8: 1357-1371.

Chowdhury P, Bone RC, Louria DB, Rayford PL. Effect of cigarette smoke on human serum trypsin inhibitory capacity and antitrypsin concentrations. Am Rev Respir Dis 1982; **126**: 177-179.

Chrzanowski P, Keller S, Cerreta J, Mandl I, Turino GM. Elastin content of normal and emphysematous lung parenchyma. Am J Med 1980; **69:** 351-359.

Ciba Guest Symposium: Terminology, definitions and classification of chronic pulmonary emphysema. Thorax 1959; 14: 286-299.

Cigarette smoking among adults-United States, 1991. MMWR 1993; 42: 230-233. Current Trends.

Cochrane CG, Spragg R, Revak SD. Studies on the pathogenesis of the adult respiratory distress syndrome: evidence of oxidant activity in bronchoalveolar lavage fluid. J Clin Invest 1983; **71**: 754-761.

Cohen AB, Girard W, McLarty J, Starcher B, Stevens M, Fair DS, Davis D, James H, Rosenbloom J, Kucich U. A controlled trial of colchicine to reduce the elastase load in the lungs of cigarette smokers with chronic obstructive pulmonary disease. Am Rev Respir Dis 1990; **142**: 63-72.

Cohen AB, Girard W, McLarty J, Starcher B, Davis D, Stevens M, Rosenbloom J, Kucich U. A controlled trial of colcichine to reduce the elastase load in the lungs of ex-cigarette smokers with chronic obstructive pulmonary disease. Am Rev Respir Dis 1991; **143**: 1038-1043.

(a) Colebatch HJH, Ng CKY, Nikov N. Use of an exponential function for elastic recoil. J Appl Physiol 1979; **46**: 387-393.

(b) Colebatch HJH, Greaves IA, Ng CKY. Exponential analysis of elastic recoil and ageing in healthy males and females. J Appl Physiol 1979; **47:** 683-691.

Colp C, Pappas J, Moran D, Lieberman J. Variants of alpha 1-antitrypsin in Puerto Rican children with asthma. Chest 1993; **103**: 812-815.

Cone F, Lellouch J, Schwartz D. Smoking and leukocyte counts: results of an epidemiological study. Lancet 1971; **2:** 632-634

(a)Cook L, Ternai B. Deleterious effect of Brij 35 on alkyl 2-pyrones and other hydrophobic inhibitors of human sputum and leucocyte elastase. Biochem Intern 1988; 17: 637-646.

(b)Cook L, Ternai B. Similar binding sites for unsaturated fatty acids and alkyl 2-pyrone inhibitors of human sputum elastase. Biol Chem Hoppe-Seyler 1988; **369**: 627-631.

Cox DW, Billingsley GD. Oxidation of plasma alpha 1-antitrypsin in smokers and nonsmokers and by an oxidizing agent. Am Rev Respir Dis 1984; **130**: 594-599.

Crowle AJ. An attempt to produce emphysema in the guinea pig. Am Rev Respir Dis 1959; 80: 153

Crystal RG, Brantly ML, Hubbard RC, Curiel DT, States DJ, Holmes MD. The alpha 1-antitrypsin gene and its mutations: Clinical consequences and strategies for therapy. Chest 1989; **95**: 196-208.

Damiano, VV, Tang, A, Kuchic, U, Abrams, wr, Rosenbloom, J, Kimbel, P, Fallahnejad M, Weinbaum G. Immunolocalisation of elastase in human emphysematous lungs. J Clin Invest 1986; **78:** 482-493.

Danishefsky I, Steiner H. Investigation on the chemistry of heparin V. Disaccharides obtained after partial hydrolysis. Biochim Biophys Acta 1965; **101:** 37-45.

Darnule, TV, McKee, M, Darnule, AT, Turino, GM, Mandl, I. Solid-phase radioimmunoassay for estimation of elastin peptides in human sera. Anal Biochem 1982; **122**: 302-307.

Davies SF, Offord KP, Brown MG, Campe H, Niewoehner D. Urine desmosine is unrelated to cigarette smoking or to spirometric function. Am Rev Respir Dis 1983; **128**: 473-475.

Davis WB, Fells GA, Chernick MS, diSant'Agnese PA, Crystal RG. A role for neutrophils in the derangements of the bronchial wall characteristic of cystic fibrosis. Am Rev Respir Dis 1983; **127**: 207A

Davison AG, Fayers PM, Taylor AJ, Taylor AJ, Darbyshire J, Pickering CA, Chettle DR, Franklin D, Guthrie CJ, Scott MC, et al. Cadmium fume inhalation and emphysema. Lancet 1988; 1: 663-667.

de Santi MM, Martorana PA, Cavarra E, Lungarella, G. Pallid mice with genetic emphysema. Neutrophil elastase burden and elastin loss occur without alteration in the bronchoalveolar population. Lab Invest 1995; **73:** 40-47. DeWater R, Willems LNA, Van Mujen GNP, Franken C, Fransen JAM, Dijkman JH, Kramps JA. Ultrastructural localization of bronchial antileukoprotease in central and peripheral human airways by a gold-labeling technique using monoclonal antibodies. Am Rev Respir Dis 1986; **133**: 882-890.

Dillon TJ, Walsh R., Scicchitano R, Cleary EG, McLennan G. Sensitive ELISA for measuring elastin breakdown products in plasma. Clin Chem 1990; **36:** 392-393.

Dillon TJ, Walsh RL, Scicchitano R, Eckert B, Cleary EG, McLennan, G. Plasma elastin-derived peptide levels in normal adults, children, and emphysematous subjects. Physiologic and computed tomographic scan correlates. Am Rev Respir Dis 1992; **146**: 1143-1148.

Dockery DW, Speizer FE, Ferris BG Jr, Ware JH, Louis TA, Spiro A. Cumulative and reversible effects of lifetime smoking on simple tests of lung function in adults. Am Rev Respir Dis 1988; 137: 286-292.

Drag L, Petersen LC. Activation of Pro-urokinase by cathepsin G in the presence of glucosaminoglycans. Fibrinolysis 1994; 8: 192-199.

du Bois RM, Bernaudin JF, Paako P, Hubbard R, Takahashi H, Ferrans V, Crystal RG. Human neutrophils express the alpha 1-antitrypsin gene and produce alpha 1-antitrypsin. Blood 1991; **77**: 2724-2730.

Dwyer EM Jr, Troncale F. Spontaneous pneumothorax and pulmonary disease in the Marfan syndrome: report of two cases and review of the literature. Ann Intern Med 1965; **62:** 1285-92;

Eden E, Mitchell D, Mehlman B, Khouli H, Nejat M, Grieco MH, Turino GM. Atopy, asthma, and emphysema in patients with severe α -1-antitrypsin deficiency. Am J Respir Crit Care Med 1997; **156:** 68-74.

Edwards DW. A summary of the Third Conference on Research in Emphysema, Air Pollution, and Chronic Pulmonary Insufficiency. Session 5. Am Rev Respir Dis 1961; **38:** 581

Eisenberg SDP, Hale KK, Heimdal P, Thompson RC. Location of the protease-inhibitory region of secretory leukocyte protease inhibitor. J Biol Chem 1990; **265**: 7967-7981.

Engelbrecht S, Pieper E, Macartney HW, Rautenberg W, Wenzel HR, Tschesche H. Hoppe-Seyler's Z Physiol Chem 1982; **363**: 305-315.

Eriksson S. Studies in a1-antitrypsin deficiency. Acta Med Scand 1965; 117: Suppl. 432, 1-85.

Eriksson S, Lindmark B, Lilja H. Familial α_1 -antichymotrypsin deficiency. Acta Med Scand 1986; **220:** 447-453.

Evans MD, Pryor WA. Cigarette smoking emphysema and damage to alpha-1-proteinase inhibitor. Am J Physiol 1994; **266:** L593-L611.

Farndale RW, Buttle DJ, Barret AJ. Improved quantitation of sulphated glycosaminoglycans by use of dimethylmethylene blue. Biochim Biophys Acta 1986; **883**: 173-177.

Feinlieb M, Rosenberg HM, Collins JG, Delozier JE, Pokras R, Chevarley FM. Trends in COPD morbidity and mortality in the United States. Am Rev Respir Dis 1989; **140:** S9-S18.

Fera T, Abbout RT, Richter A, Johal SS. Acute effect of smoking on elastaselike esterase activity and immunologic neutrophil elastase levels in bronchoalveolar lavage fluid. Am Rev Respir Dis 1986; **133**: 568-573.

Fick RB, Reynolds H Y. Pseudomonas Respiratory infection in cystic fibrosis: a possible defect in opsonic IgG antibody. Bull Eur Physiopathol Respir 1983; **19:** 151-161

Finlay TH, Kadner SS, Nathoo SA, LevytskaV, Frazer, JM, Galdston M. Effects of detergents on the activities of α_2 -macroglobulin-protease complexes. Ann NY Acad Sci 1983; **421**: 340-351.

Feinleib M, Rosenberg HM, Collins JG, Delozier JE, Pokras R, Chevarley FM. Trends in COPD morbidity and mortality in the United States. Am Rev Respir Dis 1989; **140:** S9-S18.

Fishbein L. Sources, nature and levels of air pollutants. In: Tomatis, L., ed. Indoor and outdoor air pollution and human cancer. Berlin-Heidelberg, Germany, Springer-Verlag, 1993; pp. 17-66.

Fisk DE, Kuhn C. Emphysema-like changes in the lungs of the blotchy mouse. Am Rev Respir Dis 1976; 113: 787-797.

Fletcher C, Peto R. The natural history of chronic airflow obstructive pulmonary disease. BMJ 1977; 1: 1645-1648.

Fletcher DS, Osinga DG, Hand KM, Dellea PS, Ashe BM, Mumford RA, Davies P, Hagman W, Finke PE, Doherty JB, Bonney RJ. A comparison of α_1 -proteinase inhibitor methoxysuccinyl-alaala-pro-val-chloromethylketone and specific β -lactam inhibitors in an acute model of human polymorphonuclear leukocyte elastase-induced lung hemorrhage in the hamster. Am Rev Respir Dis 1990; **141:** 672-677.

Fowler AA, Walchak S, Giclas PC, Henson PH, Hyers TH. Characterisation of antiproteinase activity in the adult Respiratory distress syndrome. Chest 1982; 81: 50S-51S

Fox B, Bull TB, Guz A, Harris E, Tetley TD. Is neutrophil elastase associated with elastic tissue in emphysema? J Clin Pathol 1988; 41: 435-440.

Fransson LA, Johansson BG. Interaction of heparin and heparan sulphate with thrombin. Int J Biol Macrom 1981; **3:** 25-30.

Fransson LA, Havsmark B. Interaction of heparin and heparan sulphate with low-density lipoproteins. Int J Biol Macrom 1982; **4:** 73-78.

Frasca JM, Auerbach O, Parks V, Jamieson JD. Electron microsopic observations on pulmonary fibrosis and emphysema in smoking dogs. Exp Mol Pathol 1971; **15**: 108-126.

Frasca JM, Auerbach O, Carter HW, Parks VR. Morphologic alterations induced by short-term cigarette smoking. Am J Pathol 1983; **111:** 11-20.

Freeman G, Haydon GB. Emphysema after low-level exposure to NO₂. Arch Environ Health 1964; **8**: 125-128.

Freeman G, Crane SC, Furiosi NJ, Stephens RJ, Evans MJ, Moore WD. Covert reduction in ventilatory surface in rats during prolonged exposure to subacute nitrogen dioxide. Am Rev Respir Dis 1972; **106**: 563-579.

Frette C, Wei SM, Neukirch F, Sesboue R, Martin JP, Kauffman F. Relation of serum elastin peptide concentration to age, FEV1, smoking habits, alcohol consumption, and protease inhibitor phenotype: an epidemiological study in working men. Thorax 1992; **47**: 937-942.

Frette C, Jacob MP, Defouilloy C, Atassi C, Kauffman F, Pham QT, Bignon J. Lack of relationship of elastin peptide levels to emphysema assessed by CT scans. Am J Respir Crit Med 1996; **153**: 1544-1547.

Frette C, Jacob MP, Wei SM, Bertrand JP, Laurent P, Kauffman F, Pham QT. Relationship of serum elastin peptide level to single breath transfer factor for carbon monoxide in French coal miners. Thorax 1997; **52:** 1045-1050.

Frommherz KJ, Faller B, Bieth JG. Heparin strongly decrease the rate of inhibition of neutrophil elastase by α_1 -proteinase inhibitor. J Biol Chem 1991; **266**: 15356-15362.

Fryksmark U, Ohlsson K, Rosengreen M, Tegner H. Studies on the interaction between leukocyte elastase, anti leukoprotease and the plasma proteinase inhibitors, a1-proteinase inhibitor and a2-macroglobulin. Hoppe-Seyler's Z Physiol Chem 1983; **364**: 793-800.

Fujita J, Nelson NL, Daughton DM, Dobry CA, Spurzem JR, Irino S, Rennard SI. Evaluation of elastase and antielastase balance in patients with chronic bronchitis and pulmonary emphysema. Am Rev Respir Dis 1990; **142:** 57-62.

Gabay JE, Scott RW, Campanelli D. Griffith J, Wilde C, Marra MN, Seeger M, Nathan CF. Antibiotic proteins of human polmorphonuclear leukocytes. Proc Natl Acad Sci 1989; **86:** 5610-5614.

Gadek JE, Fells G, Crystal RG. Cigarette smoking induces functional antiprotease deficiency in the lower Respiratory tract of humans. Science 1979; **206**: 1315-1316.

(a) Gadek JE, Klein HG, Holland PV, Crystal RG. Replacement therapy of alpha₁-antitrypsin deficiency: reversal of protease-antiprotease imbalance within the alveolar structures of PiZ subjects. J Clin Invest 1981; **68**: 1158-1165.

(b) Gadek JE, Zimmerman RL, Fells GA, Crystal RG. Anti-elastases of the human alveolar structure: assessment of the protease antiprotease theory of emphysema. J Clin Invest 1981; **68**: 889-898.

Galdston MV, Levystska, V, Liener IE, Twusami DY. Degradation of trophoelastin and elastin substrates by human neutrophil elastase, free and bound to alpha ₂-macroglobulin in serum of the M and Z (Pi) phenotypes for alpha₁-antitrypsin. Am Rev Respir Dis 1979; **119**: 435-441.

Garver RI, Chytil A, Karlsson S, Fells GA, Brantly ML, Courtney M, Kantoff PW, Nienhuis AW, Anderson WF, Crystal RG. Production of glycosylated physiologically 'normal' human α_1 antitrypsin by mouse fibroblasts modified by insertion of human α_1 -antitrypsin cDNA using a retroviral vector. Proc Natl Acad Sci USA 1987; **84:** 1050-1054. Ge YM, Zhu Yj, Lou WC, Gong YH, Zhang XQ. Damaging role of neutrophil elastase in the elastic fiber and basement membrane in human emphysematous lung. Chin Med J [Engl] 1990; **103**: 588-594.

Gelb AF, Hogg JC, Muller NL, Schein MJ, Kuei J, Tashkin DP, Epstein JD, Kollin J, Green RH, Zamel N, Elliott WM, Hadjiaghai L. Contribution of emphysema and small airways in COPD. Chest 1996; **109:** 353-359.

Gillissen A, Birrer P, McElvaney NG, Buhl R, Vogelmeier C, Hoyt RF Jr, Hubbard RC, Crystal RG. Recombinant secretory leukoprotease inhibitor augments glutathione levels in lung epithelial lining fluid. J Appl Physiol 1993; **75:** 825-832.

Glasgow JE, Pietra GG, Abrams WR, Blank J, Oppenheim DM, Weinbaum G. Neutrophil recruitment and degranulation during induction of emphysema in the rat by nitrogen dioxide. Am Rev Respir Dis 1987; **135**: 1129-1136.

Goldstein IM. Polymorphonuclear leukocyte lysosymes and tissue injury. Prog Allergy 1976; 20: 301-340.

Goldstein RA, Starcher, BC. Urinary excretion of elastin peptides containing desmosine after intratracheal injection of elastase in hamsters. J Clin Invest 1978; **61**: 1286-1290.

Gottlieb DJ, Stone PJ, Sparrow D, Gale ME, Weiss ST, Snider GL, O'Connor GT. Urinary demosine excretion in smokers with and without rapid decline of lung function. Am J Crit Care Med 1996; **154**: 1290-1295.

Gould GA, MacNee W, McLean A, Warren PM, Redpath A, Best JJ, Lamb D, Flenly DC. CT measurements of lung density in life can quantitate distal airspace enlargement - an essential defining feature of human emphysema. Am Rev Respir Dis 1988; **137**: 380-392.

Gross P, Babyak MA, Tolker EB, Kaschak M. Enzymatically produced emphysema. A preliminary report. J Occup Med 1964; 6: 481-484.

Gross P, Pfitzner EA, Tolker EB, Babyak MA, Kaschak M. Experimental emphysema. Its production with papain in normal and silicotic rats. Arch Environ Health 1965; 11: 50-58.

Groutas WC. Inhibitors of leukocyte elastase and leukocyte cathepsin G. Agents for the treatment of emphysema and related ailments. Med Res Rev 1987; **7**: 227-241.

Guenter CA, Coalson JJ, Jacques J. Emphysema associated with intravascular leukocyte sequestration. Comparison with papain induced emphysema. Am Rev Respir Dis 1981; **123**: 79-84.

Hammond EC, Auerbach O, Kirman D, Lawrence G. Effects of cigarette smoking on dogs. Arch Environ Health 1970; **21:** 740-753.

Harada RN, Repine JE. Pulmonary host defence mechanisms. Chest 1985; 87: 247-252

Harel S, Janoff A, Yu SY, Hurewitz A, Bergofsky, EH. Desmosine radioimmunoassay for measuring elastin degradation in vitro. Am Rev Respir Dis 1980; **122**: 769-773.

Haslett C. Resolution of acute inflammation and the role of apoptosis in the tissue fate of granulocytes. Clin Sci 1992; 83: 639-648

Haslett C. Granulocyte apoptosis and inflammatory disease. Br Med Bull 1997; 53: 669-683.

Hautamaki Rd, Kobayashi DK, Senior RM, Shapiro SD. Requirement for macrophage elastase for cigarette smoke-induced emphysema in mice. Science 1997; **277**: 2002-2004.

Hawgood S, Benson BJ, Hamilton RL, Jr. Effects of a surfactant-associated protein and calcium ions on the structure and surface activity of lung surfactant lipids. Biochemistry 1985; **24:** 184-190.

Heckman CA, Dalbey WE. Pathogenesis of lesions induced in rat lung by chronic tobacco smoke inhalation. JNCI 1982; **69:** 117-129.

Hendrick DJ. Occupation and chronic obstructive pulmonary disease (COPD). Thorax 1996; **51**: 947-955.

Henson PM, Johnston RB. Tissue injury in inflammation: oxidants, proteinases and cationic proteins. J Clin Invest 1987; **79:** 669-674

Herbert JM, Frehel D, Rosso MP, Seban E, Castet C, Pepin O, Maffrand JP, Le Fur G. Biochemical and pharmacological activities of SR 26831, a potent and selective elastase inhibitor. J Pharmacol Exp Therapeut 1991; **260**: 809-816.

Heremans A, Verschakaelen JA, Fraeyenhoven L, van Demedts M. Measurement of lung density by means of quantitative scanning. A study of correlations with pulmonary function test. Chest 1992; **102**: 805-811.

Higgins MW, Keller JB. An index of risk for obstructive airways disease. Am Rev Respir Dis 1982; **125**: 144-151.

Hirst RN, Perry HM, Cruz MG, Pierce JA. Elevated cadmium concentration in emphysematous lungs. Am Rev Respir Dis 1973; **108:** 30-39

Hochstrasser K, Reichert R, Schwarz S, Werle E. Isolierung und charakterisierung eines protease inhibitors aus menschlichem bronchialsekret. Hoppe-Seylers Z Physiol Chem 1972; **353**: 221-226.

Hoidal JR, Fox RB, Le Marbe PA et al. Altered oxidative metabolic responses in vitro of alveolar macrophages from asymptomatic cigarette smokers. Am Rev Respir Dis 1981; **123**: 85-89.

(a)Hoidal JR, Niewoehner, DE. Pathogenesis of emphysema. Chest 1983; 83: 679-685.

(b)Hoidal JR, Niewoehner, DE. Cigarette smoke potentiates elastase-induced emphysema in hamsters. Am Rev Respir Dis 1983; **127**: 478-481.

Hoidal JR, Niewoehner DE, Rao NV, Hibbs, MS. The role of neutrophils in the development of cadmium chloride-induced emphysema in lathyrogen-fed hamsters. Am J Pathol 1985; **120**: 22-29.

Höök M, Björk I, Hopwood J, Lindahl U. Anticoagulant activity of heparin: Separation of highactivity and low-activity heparin species by affinity chromatography on immobilised antithrombin. FEBS Lett 1976; **66**: 90-93. Höök M, Kjellen L, Johansson S, Robinson J. Cell surface glycosaminoglycans. Ann Rev Biochem 1984; **53**: 847-69.

Hruban RH, Meziane MA, Zerhouni EA, Khouri NF, Fishman EK, Wheeler PS, Dumler JS, Hutchins GM. High resolution computed tomography of inflation fixed lungs: pathologic - radiologic correlation of centrilobular emphysema. Am Rev Respir Dis 1987; **136**: 935-940.

Hubbard RC, Sellers S, Czerski D, Stephens L, Crystal RG. Biochemical efficacy and safety of monthly augmentation therapy for α_1 -antitrypsin deficiency. JAMA 1988; **260**: 1259-1264.

Hubbard RC, Brantley ML, Sellers S, Mitchell ME, Crystal, RG. Delivery of proteins for therapeutic purposes by aerosolization: direct augmentation of anti-neutrophil elastase defence of the lower Respiratory tract in α_1 -antitrypsin deficiency with an aerosol of α_1 -antitrypsin. Ann Intern Med 1989; **111**: 206-212.

(a) Hubbard, RC, Brantley, ML, Crystal, RG. Proteases. In: Crystal RG, West JB, et al, eds. The Lung. New York: Raven Press, 1991; 1763-1773.

(b) Hubbard RC, Fells G, Gadek J, Pacholok S, Humes J, Crystal, R.G. Neutrophil accumulation in the lung in α_1 -antitrypsin deficiency. Spontaneous release of leukotriene B₄ by alveolar macrophages. J Clin Invest 1991; **88**: 891-897

Huber GL, Davies, P, Zwilling GR, Pochay VE, Hinds WC, Nicholas HA, Mahajan, VK, Hayashi, M, First MW. A morpological and physiological bioassay for quantifying alterations in the lung following experimental chronic inhalation of tobacco smoke. Bull Eur Physiopathol Respir 1981; **17**: 269-327.

Hunninghake GW, Crystal, RG. Cigarette smoking and lung destruction: accumulation of neutrophils in the lungs of cigarette smoker. Am Rev Respir Dis 1983; **128**: 833-838

Idell S, Kucich U, Fein A, Kueppers F, James HL, Walsh, PN, Weinbaum G, Colman RW, Cohen AB. Neutrophil elastase-releasing factors in bronchoalveolar lavage from patients with adult Respiratory distress syndrome. Am Rev Respir Dis 1985; **132**:1098-1105.

Ingram RJ. In: Harrison, ed. Principles of Internal Medicine. 1994; 13th Edit. pp 1197

Jackson R.L, Busch SJ, Cardin AD. Glycosaminoglycans: molecular properties, protein interactions, and role in physiological processes. Physiological Rev 1991; **71:** 481-539.

Jallat S, Carvallo D, Tessier LH, Roecklin D, Roitsch C, Ogushi F, Crystal RG, Courtney M. Altered specificities of genetically engineered α1-antitrypsin variants. Protein Eng 1986; 1: 29-35.

Janoff A. Elastases and emphysema: Current assessment of the protease-antiprotease hypothesis. Am Respir Dis 1985; **132:** 417-433.

Janoff A, Sloan B, Weinbaum G, Damiano V, Sandhaus RA, Elias J, Kimbel P. Experimental emphysema induced with purified human neutrophil elastase: tissue localization of the instilled protease. Am Rev Respir Dis 1977; **115**: 461.

Janoff A, Carp H, lee DK, Drew RT. Cigarette smoke inhalation decreases alpha 1-antitrypsin activity in rat lung. Science 1979; **206**: 1313-1314.

(a) Janoff A, Raju L, Dearing R. Levels of elastase activity in bronchoalveolar fluids of healthy smokers and nonsmokers. Am Rev Respir Dis 1983; **127**: 540-544.

(b) Janoff A, Chanana AD, Joel DD, Susskind H, Laurent P, Yu SY, Dearing R. Evaluation of the urinary desmosine radioimmunoassy as a monitor of lung injury after endobronchial elastase installation in sheep. Am Rev Respir Dis 1983; **128**: 545-551.

Johnson D, Travis J. Structural evidence for methionine at the reactive site of human alpha 1 proteinase inhibitor. J Biol Chem 1978; **253**: 7142-7144.

Johnson D, Travis J. The oxidative inactivation of human α -1-proteinase inhibitor. Further evidence for methionine at the reactive center. J Biol Chem 1979; **254:** 4022-4026.

Johnston ID, Strachan DP, Anderson HR. Effect of pneumonia and whooping cough in childhood on adult lung function. N Engl J Med 1998; **338**: 581-587.

Jori G, Galiazzo G, Marzotto A, Scoffone E. Selective and reversible photo-oxidation of the methionyl residues of lysozyme. J Biol Chem 1968; **243**: 4272-4278.

Juhos LT, Green DP, Furiosi NJ, Freeman G. A quantitative study of stenosis in the respiratory bronchiole of the rat in NO₂ induced emphysema. Am Rev Respir Dis 1980; **121**: 541-549.

Kao RC, Wehner NG, Skubitz KM, Gray BH, Hoidal JR. Proteinase 3. A distinct human polymorphonuclear leukocyte proteinase that produces emphysema in hamsters. J Clin Invest 1988; **82:** 1963-73.

Karlinsky JB, Fredette J, Davidovits G, Catanese A, Snider R, Faris B, Snider GL, Franzblau C. The balance of the lung collective tissue elements in elastase-induced emphysema. J Lab Clin Med 1983; **102:** 151-162.

Kauffmann F, Frette C, Pham QT, Nafissi S, Bertrand JP, Oriol R. Associations of blood grouprelated antigens to FEV1, wheezing, and asthma. Am J Respir Crit Care Med 1996; **153**: 76-82.

Keil M, Lungarella G, Cavarra E, van Even P, Matorana PA. A scanning electron microscopic investigation of genetic emphysema in tight-skin, pallid, and beige mice, three different C57 BL/6J mutants. Lab Invest 1996; **74:** 353-362.

Kennedy AJ, Cline A, Ney UM, Johnson WH, Roberts NA. The effect of a peptide aldehyde reversible inhibitor of elastase on a human leucocyte elastase-induced model of emphysema in the hamster. Eur J Respir Dis 1987; **71:** 472-478.

Kida K, Thurlbeck WM. The effects of β -aminoproprionitrile on the growing rat lung. Am J Pathol 1980; **101:** 693-707.

Kilroe-Smith TA, Dowdeswell RJ, Gaillard M. Elastase binding capacity of alpha 2-macroglobulin in plasma of patients with asthma or chronic obstructive pulmonary disease, without alpha 1-protease inhibitor deficiency. Clin Chim Acta 1989; **185**: 81-90.

Klebanoff SJ, Clark RA. Neutrophil function and clinical disorders. Amsterdam: North-Holland Pulb; 1978.

Kleinerman J, Ip MPC. Effect of nitrogen dioxide on elastin and collagen contents of lung. Arch Environ Health 1979; **34**: 228-232.

Klumpp T, Bieth JG. Automated measurement of the elastase-inhibitory capacity of plasma with a centrifugal analyser. Clin Chem 1979; **25:** 969-972.

Knight KR, Burdon JG, Cook L, Brenton S, Ayad M, Janus ED. The proteinase-antiproteinase theory of emphysema: a speculative analysis of recent advances into the pathogenesis of emphysema. Respirology 1997; **2**: 91-95.

Koopman P, Povey S, Lovell-Badge RH. Widespread expression of human α 1-antitrypsin in transgenic mice revealed by in situ hybridization. Genes Dev 1989; **3:** 16-25.

Kramps JA, Franken C, Meijer CJLM, Dijkman JH. Localisation of a low-molecular-weight protease inhibitor in serous cells of the Respiratory tract. J Histochem Cytochem 1981; **29:** 712-719.

Kramps JA, Franken C, Dijkman JH. Quantity of anti-leukoprotease relative to α_1 -proteinase inhibitor in peripheral airspaces of the human lung. Clin Sci 1988; **75**: 351-353.

Kramps JA, Applehaus H, Nikiforov T, Dijkman JH. Trypsin, chymotrypsin and leukocyte elastase are inhibited by the second COOH-terminal domaine of antileukoprotease. Am Rev Respir Dis 1989; **139:** A201.

Kramps JA, Rudolphus A, Stolk J, Willems LNA, Dijkman JH. Role of antileukoprotease in the lung. Ann NY Acad Sci 1991; **624:** 97-108.

Kruger U. Chronic obstructive lung disease and alpha-2-macroglobulin deficiency in serum – case report. Pneumologie 1993; **47:** 531-534.

Kubes P, Smith R, Grisham MD, Granger DN. Neutrophil-mediated proteolysis. Differential roles for cathepsin G and elastase. Inflammation 1993; **17**: 321-332.

Kucich U, Christner P, Lippman M, Kimbel P, Williams G, Rosenbloom J, Weinbaum G. Utilization of a peroxidase antiperoxidase complex in an enzyme-linked immunosorbent assay of elastin-derived peptides in human plasma. Am Rev Respir Dis 1985; **131**: 709-713.

Kuhn C, Yu SY, Chraplyvy M, Linder RE, Senior RM. The induction of emphysema with elastase II. Changes in connective tissue. Lab Invest 1976; **34:** 372-380.

Kuhn C, Starcher BC. The effect of lathyrogens on the evolution of elastase-induced emphysema. Am Rev Respir Dis 1980; **122**: 453-460.

Laennec RTH. A treatise on the diseases of the chest and on mediate ausculation. John Forbes, trans. From the 4th London ed. Philadelphia: Desilar, Thomas & Co., 1835; 135-163.

Lafuma C, Frisdal E, Harf A, Robert L, Hornebeck W. Prevention of leucocyte elastase-induced emphysema in mice by heparin fragments. Eur Respir J 1991; **4:** 1004-1009.

Lamb D. Chronic bronchitis, emphysema, and the pathological basis of chronic obstructive pulmonary disease. In: Spencer's pathology of the lung. 1996; pp. 597-629.

Lane DA, Lindahl U, eds. Heparin: Chemical and biological applications, Edward Arnold, 1989.

Langunof'f D, Phillips MT, Iseri OA, Benditt EP. Isolation and preliminary characterisation of rat mast cell granules. Lab Invest 1964; **13**: 1331-1344.

Laurell CB, Eriksson S. The electrophoretic alpha-1-globulin pattern of serum in alpha-1antitrypsin deficiency. Scand J Clin Lab Invest 1963; **15**: 132-140.

Laurent P, Janoff A, Kagan HM. Cigarette smoking blocks cross-linking of elastin in vitro. Am Rev Respir Dis 1983; **127:** 189-192.

Leaderer BP, Zagraniski RT, Berwick M, Stolwijk JA. Assessment of exposure to indoor air contaminants from combustion sources: methodology and application. Am J Epidemiol 1986; **124**: 275-289.

Lee CT, Fein AM, Lippmann M, Holtzman H, Kimbel P, Weinbaum G. Elastolytic activity in pulmonary lavage fluid from patients with adult Respiratory distress syndromes. N Engl J Med 1981; **304:** 192-196

Lellouch J, Claude JR, Martin JP, Orssaud G, Zaoui D, Bieth JG. Smoking does not reduce the functional activity of serum alpha-1-proteinase inhibitor. Am Rev Respir Dis 1985; **132:** 818-820.

Lentini A, Ternai B, Ghosh P. Synthetic inhibitors of human leukocyte elastase. Biochem Int 1985; 10: 221-232.

Lesser M, Galicki N, Cardoza C, Gariola CG. Cathepsin L activity in alveolar macrophages of rats: Respironse to cigarette smoke. Am J Respir Cell Mol Biol 1989; 371-376.

Lesser M, Padila ML, Cardoza C. Induction of emphysema in hamsters by intratracheal instillation of cathepsin B. Am Rev Respir Dis 1992; **145**: 661-668.

Lestienne P, Bieth JG. Activation of human leukocyte elastase activity by excess substrate, hydrophobic solvents, and ionic strength. J Biol Chem 1980; **255**: 9289-9294.

Lewis RW, Harwood JL, Tetley TD, Harris E Richards RJ. Degradation of human and rat surfactant apoprotein by neutrophil elastase and cathepsin G. Biochem Soc Trans 1992; **21**: 206S.

Lindahl U, Hook M. Glycosaminoglycans and their binding to biological macromolecules. Ann Rev Biochem 1978; **47:** 385-417.

Lindmark BE, Arborelius M, Erikssen SG. Pulmonary function in middle-aged women with heterozygous deficiency of the serine protease inhibitor alpha₁ -antichymotrypsin. Am Rev Respir Dis 1990; **141**: 884-888.

Loeb LM. The aetiology of emphysema. Arch Int Med 1930; 45: 464-472

Lohse DL, Linhardt RJ. Purification and characterisation of heparin lyases from Flavobacterium heparinum. J Biol Chem 1992; **267:** 24347-24355.

Loike JD, Silverstein R, Wright SD, Weitz JI, Huang AJ, Silverstein SC. The role of protected extra-cellular compartments in interactions between leukocytes and platelets and fibrin/fibrinogen matrices. Ann NY Acad Sci 1992; **667**: 163-172.

Lomas DA, Evans DL, Stone SR, Chang WSW, Carrell RW. Effect of the Z mutation on the physical and inhibitory properties of alpha-l-antitrypsin. Biochemistry 1993; **32**: 500-508.

Longstaff C, Gaffney PJ. Serpin-serine protease binding kinetics: α_2 - antiplasmin as a model inhibitor. Biochemistry 1991; **30**: 979-986.

Lonky SA, Marsh J, Wohl H. Stimulation of human granulocyte elastase by platelet factor 4 and heparin. Biochem Biophys Res Commun 1978; **85:** 1113-8.

Lonky SA, Wohl H. Regulation of elastolysis of insoluble elastin by human leukocyte elastase: stimulation by lysine rich ligands, anionic detergents, and ionic strength. Biochemistry 1983; **22**: 3174-3720.

Lowry OH, Rosenbrough NJ, Farr AL, Randall RJ. Protein measurement with the Folin phenol reagent. J Biochem 1952; **192:** 265-275.

Lucey EC, Stone PJ, Christensen TG, Turner JL, Snider GL. Human neutrophil elastase and ozoneinduced emphsema in hamsters are additive. Am Rev Respir Dis 1990; **141:** A736.

Lucey EC, Stone PJ, Snider GL. Consequences of proteolytic injury. In: Crystal RG, West JB, Barnes PJ, Cherniak NS, Weibel ER, eds. The lung: scientific foundations. New York: Raven Press, 1991; 1789-1801.

Luisetti M, Travis, J. Bioengineering: α_1 -Proteinase inhibitor site-specific mutagenesis. The prospect for improving the inhibitor. Chest 1996; **110**: 278s-283s.

Lundquist P, Martensson J, Sorbo B, Ohman S. Turbidimetry of inorganic sulfate, ester sulfate, and total sulfur in urine. Clin Chem 1980; **26**: 1178-1181.

MacKenzie ID, Bartecchi CE, Schrier RW. The human costs of tobacco abuse. N Engl J Med 1994; **330:** 907-912, 975-980.

MacNee W, Gould G, Lamb D. Quantifying emphysema by CT scanning. Clinicopathological correlations. Ann NY Acad Sci 1991; **624:** 179-194.

Marcum J, Rosenberg RD. Anticoagulantly active heparin-like molecules from vascular tissue. Biochemistry 1984; 23: 1730-1737.

Marossy K. Interaction of the chymotrypsin- and elastase-like enzymes of the human granulocyte with glycosaminoglycans. Biochim Biophys Acta 1981; **659**: 351-361.

Martodam RR, Baugh RJ, Twumasi DY, Liener IE. A rapid procedure for the large scale purification of elastase and cathepsin G from human sputum. Prep. Biochem. 1979; 9: 15-31.
Martorana, PA, Brand T, Gardi C, van Even P, de Santi MM, Calzoni P, Marcolongo P, Lungarella G. The pallid mouse: a model of genetic α !-antitrypsin deficiency. Lab Invest 1993; **68**: 233-241.

McGowan SE, Stone PJ, Calore JD, Snider GL, Franzblau C. The fate of neutrophil elastase incorporated by human alveolar macrophages. Am Rev Respir Dis 1983; **127:** 449-455.

McGuire WM, Spragg RG, Cohen AB, Cochrane CG. Studies on the pathogenesis of the adult respiratory distress syndrome. J Clin Invest 1982; 69: 543-553.

McLennan G, Walsh RL, Thomas PD, Stevens M. Soluble proteins in bronchoalveolar lavage fluid in the adult respiratory distress syndrome. Am Rev Resp Dis 1984; **129**: A123.

McLennan G, Walsh RL. Soluble proteins in bronchoalveolar lavage and serum in ARDS, IPF, sarcoidosis, and normals. Am Rev Respir Dis 1995; A205.

Mendelssohn A. Der mechanismus der Respiriation und circkulation oder das explicirte wesen der lungenhyperamien. Berlin: B. Behrs, 1845.

Merchant RK, Schwartz DA, Helmers RA, Dayton CS, Hunninghake GW. Bronchoalveolar lavage cellularity. Am Rev Respir Dis 1992; **146:** 48-53.

Moczar E, Hornebeck W. A lipophilic heparin derivative protects elastin against degradation by leucocyte elastase. Int J Biol Macromol 1991; **13:** 261-262.

Morrison HM, Kramps JA, Burnett D, Stockley RA. Lung lavage fluid from patients with α_1 -proteinase inhibitor deficiency or chronic obstuctive bronchitis: antielastase function and cell profile. Clin Sci 1987; **72:** 373-381.

Morrison HM, Welgus HG, Stockley RA, Burnett D, Campbell EJ. Inhibition of human leukocyte elastase bound to elastin: Relative ineffectiveness and two mechanisms of inhibitory activity. Am J Respir Cell Mol Biol 1990; **2:** 263-269.

Mortality Patterns--United States. JAMA 1993; 270: 2916-2917.

Murphy G, Cockett MI, Ward RV, Docherty AJP. Matrix metalloproteinase degradation of elastin, type IV collagen and proteoglycan. Biochem J 1991; 277: 277-279.

Murphy G, Docherty AJP. The matrix metalloproteinases and their inhibitor. Am J Respir Cell Mol Biol 1992; 7: 120-125.

Nader HB, Dietrich CP. Natural occurrence, and possible biological role of heparin. In: Lane, DA Lindahl, U, eds. Heparin: Chemical and biological applications, Edward Arnold, 1989; 81-96.

Nakajima K, Powers JC, Ashe BM, Zimmerman M. Mapping the extended substrate binding site of cathepsin G and human leukocyte elastase. Studies with peptide substrates related to the alpha 1 protease inhibitor reactive site. J Biol Chem 1979; **254:** 4027-4031.

Navia MA, McKeever BM, Springer JP, Lin TY, Williams HR, Fluder EM, Dorn CP, Hoogsteen K. Structure of human neutrophil elastase in complex with a peptide chloromethyl ketone inhibitor at 1.84 - Å resolution. Proc Natl Acad Sci U.S.A. 1989; **86:** 7-11.

Nelson RM, Cecconi O, Roberts WG, Aruffo A, Linhardt RJ, Bevilacqua MP. Heparin oligosaccharides bind L- and P- selectin and inhibit acute inflammation. Blood 1993, **11**: 3253-3258.

Niederman MS, Fritts LL, Merrill WW, Fick RB, Matthay RA, Reynolds HY, Gee JB. Demonstration of a free elastolytic metalloenzyme in human lung lavage fluid and its relation to alpha₁ -antiprotease. Am Rev Respir Dis 1984; **129**: 943-947.

Niewoehner DE, Hoidal JR. Lung fibrosis and emphysema: divergent responses to a common injury ? Science 1982; **217**: 359-360.

Niewoehner DE, Kleinerman J, Rice DB. Pathologic changes in the peripheral airways of young cigarette smokers. N Engl J Med 1974; **291**: 755-758.

Niewoehner DE, Kleinerman J. Morphometric study of elastic fibers in normal and emphysematous human lungs. Am Rev Respir Dis 1977; 115: 15-21.

Noertjojo HK, Dimich-Ward H, Peelen S, Dittrick M, Kennedy SM, Chan-Yeung M. Western red cedar dust exposure and lung function: A dose-response relationship. Am J Respir Crit Care Med 1996; **154**: 968-973.

Ockene JK, Kuller LH, Svendsen KH, Meilhan E. The relationship of smoking cessation to coronary heart disease and lung cancer in the Multiple Risk Factor Intervention Trial (MRFIT). Am J Public Health 1990; **80:** 954-958.

O'Dell BL, Kilburn KH, McKenzie WN, Thurston RJ. The lung of the copper deficent rat: a model for developmental pulmonary emphysema. Am J Pathol 1978; **91:** 413-432.

Ofulue AF, Abboud RT. Time course of neutrophil and macrophage elastinolytic activities in cigarette smoke-induced emphysema. Am J Physiol 1998; 275: L1134-L1144.

Ogushi F, Hubbbard RC, Vogelmeier GA, Fells GA, Crystal RG. Risk factors for emphysema. Cigarette smoking is associated with a reduction in the association rate constant of lung α_1 -antitrypsin for neutrophil elastase. J Clin Invest 1991; **87**: 1060-1065.

Ohlsson K. Interaction of granulocyte neutral proteases with alpha 1-antitrypsin, alpha 2macroglobulin and alpha 1-antichymotrypsin. In: Havemann K, Janoff A, eds. Neutral proteases of human polymorphonuclear leukocytes. Baltimore: Urban and Schwarzenberg. 1978; 167-77.

Oppenheim JL, Zachariae COC, Mukaida N, Matsushima K. Properties of the novel proinflammatory supergene "intercrine" cytokine family. Ann Rev Immunol 1991; **9:** 617-648.

Orie NGM, Sluiter HJ, de Vreis K, et al. The host factor in bronchitis. In Orie, NGM, Sluiter, HJ (eds): Bronchitis. Assen, The Netherlands, Royal VanGorcum, Ltd., 1961, pp 43-59.

Orsos F. Uber das elastiche Gerust der normalen and der emphysematosen Lungen Beitr Pathol 1907; **41:** 95-121.

Osman M, Cantor JO, Roffman S, Keller S, Turino GM, Mandl I. Cigarette smoke impairs elastin resynthesis in lungs of hamsters with elastase-induced emphysema. Am Rev Respir Dis 1985; **132**: 640-643.

Otto-Verberne CJM, Ten Have-Opbroek AAW, Willems LNA, Francken C, Kramps JA Dijkman JH. Lack of type II cells and emphysema in human lungs. Eur Respir J 1991; 4: 316-323.

Otto-Verberne CJM, Ten Have-Opbroek AAW, Franken C, Hermans J, Diijkman JH. Protective effect of pulmonary surfactant on elastase-induced emphysema in mice. Eur Respir J 1992; **5**: 1223-1230.

(a) Owen CA, Campbell MA, Boukedes SS, Stockley RA, Campbell EJ. A discrete subpopulation of human monocytes expresses a neutrophil-like proinflammatory (P) phenotype. Am J Physiol 1994; **267:** L775-785.

(b) Owen CA Campbell MA, Boukedes SS, Campbell EJ. Monocytes recruited to sites of inflammation express a distinctive proinflammatory (P) phenotype. Am J Physiol 1994; **267:** L786-796.

Oxman AD, Muir DCF, Shannon HS, Stock SR, Hnizdo E, Lange HJ. Occupational dust exposure and chronic obstuctive pulmonary disease. Am Rev Respir Dis 1993; **148**: 38-48.

Paakko P, Kokkonen P, Anttila S, Kalliomaki PL. Cadmium and chromium as markers of smoking in human lung tissue. Environ Res 1989; **49:** 197-207.

Paakko P, Kirby M, du Bois RM, Gillissen A, Ferrans VJ, Crystal RG. Activated neutrophils secrete stored α_1 -antitrypsin. Am J Respir Crit. Care Med 1996; **154:** 1829-1833.

Padrines M, Bieth JG. Elastin decreases the efficiency of neutrophil elastase inhibitors. J Respir Cell Mol Biol 1991; **4:** 187-193.

Pare PD, Brooks LA, Bates J, Lawson LM, Nelems JMB, Wright JL, Hogg JC. Exponential analysis of the lung pressure-volume curve as a predictor of pulmonary emphysema. Am Rev Respir Dis 1982; **126**: 54-61.

Parekh RJ, Edge CJ. Selectins - glycoprotein targets for therapeutic intervention in inflammation. Trends Biotech 1994; **12:** 339-345.

Park SS, Kikkawa Y, Goldring IP, Daly MM, Zelefsky M, Shim C, Speirer M, Morita T. An animal model of cigarette smoking in beagle dogs. Am Rev Respir Dis 1977; **115**: 971

Partridge SM, Davis HF, Adair GS. The chemistry of connective tissues: 2 soluble proteins derived from partial hydrolysis of elastin. Biochem J 1985; **61:** 11-21.

Pelham F, Wewers M, Crystal R. Buist AS, Janoff, A. Urinary excretion of desmosine (elastin cross-links) in subjects with PiZZ alpha-1-antitrypsin deficiency, a phenotype associated with heriditary predisposition to pulmonary emphysema. Am Rev Respir Dis 1985; **132**: 821-823.

Perez-Padilla R, Regaldo J, Vedal S, Paré P, Chapela R, Sansores, R, Selman M. Exposure to biomass smoke and chronic airway disease in mexican women. Am J Respir Crit Care Med 1996; **154:** 701-706

Perlmutter DH, The cellular basis for liver injury in α 1-antitrypsin deficiency. Hepatology 1991; 13: 172-185.

Perlmutter DH, Pierce JA. The α 1-antitrypsin gene and emphysema. Am J Physiol 1989; **257**: L147-L162.

Peto R, Speizer FE, Cochrane AL, Moore F, Fletcher CM, Tinker CM, Higgins IT, Gray RG, Richards SM, Gilliland J, Norman-Smith B. The relevance in adults of air-flow obstruction, but not of mucus hypersecretion, to mortality from chronic lung disease. Results from 20 years of prospective observation. Am Rev Respir Dis 1983; **128**: 491-500.

Petty TL, Silvers GW, Standford RE. Mild emphysema is associated with reduced elastic recoil and increased lung size, but not with airflow limitation. Am Rev Respir Dis 1987; **136:** 867-871.

Pierce JA, Hocott JB, Ebert RV. The collagen and elastin content of the lung in emphysema. Ann Intern Med 1961; **55**: 210-222.

Piitulainen E, Eriksson, S. Decline in FEV_1 related to smoking status in individuals with severe alpha1-antitrypsin deficiency. Eur Respir J 1999; **13**: 247-251.

Pison U, Tam EK, Caughey GH, Hawgood S. Proteolytic inactivation of dog lung surfactantassociated proteins by neutrophil elastase. Biochim Biophys Acta 1989; **992**: 251-257.

Poller W, Barth J, Voss B. Detection of an alteration of the alpha 2-macroglobulin gene in a patient with chronic lung disease and serum alpha 2-macroglobulin deficiency. Hum Genet 1989; **83**: 93-96.

(a)Poller W, Faber JP, Olek K. Highly variable course in severe α_1 -antitrypsin deficiency - use of polymerase chain reaction for the detection of rare deficiency alleles. Klin Wochenschr 1990; **68**: 857-863.

(b)Poller W, Meison C, Olek K. DNA polymorphisms of the α_1 -antitrypsin gene region in patients with chronic obstructive pulmonary disease. Eur J Clin Invest 1990; **20:** 1-7.

Poller W, Faber JP, Weidinger S, et al. A leucine-to-proline substitution causes a defective α_1 antichymotrypsin allele associated with familial obstructive lung disease. Genomics 1993; **17**: 740-743.

Potempa J, Fedak D, Dubin A, Mast A, Travis J. Proteolytic inactivation of alpha-1antichymotrypsin: sites of cleavage and generation of chemotactic activity. J Biol Chem 1991; **266**: 21482-21487.

Powers JC Boone R, Carroll DL, et al. Reaction of azapeptides with human leukocyte elastase and porcine pancreatic elastase: new inhibitors and active titrants. J Biol Chem 1984; **259**: 4288-94.

Powers JC, Harper JW. Inhibitors of serine proteinases. In: Proteinase Inhibitors. A.J. Barrett, G.J. Salvesen, eds. Elsevier Press, Amsterdam. 1986; pp 55-112.

Pryde NB, Burrows B. Development of impaired lung function: Natural history and risk factors. In: Calverley P, and Pryde N. eds. Chronic obstructive pulmonary disease. London, Chapman Hall, 1995; pp 69-91.

Quazi U, Gettins PG, Stoops JK. On the sructural changes of native human alpha2-macroglobulin upon proteinase entrapment. Three dimensional structure of the half-transformed molecule. J Biol Chem 1998; **273**: 8987-8993.

Raitjar G, Marchi E, De Gaetano G, Cerletti C. Effects of glycosaminoglycans on platelet and leucocyte function: role of N-sulfation. Biochem Pharmacol 1993; **46:** 958-960.

Rao NV, Kennedy TP, Rao G, Ky N, Hoidal JR. Sulfated polysaccharides prevent human leukocyte elastase-induced acute lung injury and emphysema in hamsters. Am Rev Respir Dis 1990; **142**: 407-412.

Rao NV, Wehner NG, Marshall BC, Sturrock AB, Hueckstead TP, Rao GV, et al. Proteinase 3 (PR-3): a polymorphonuclear leukocyte proteinase. Ann NY Acad Sci 1991; **624:** 60-68.

Rautalahti M, Virtamo J, Haukka J, Heinonen OP, Sundvall J, Albanes D, Huttunen JK. The effect of alpha-tocopherol and beta-carotene supplementation on COPD symptoms. Am J Respir Crit Care Med 1997; **156**: 1447-1452.

(a)Redini F, Tixier J-M, Petitou M, Choay J, Robert L, Hornebeck W. Inhibition of leucocyte elastase by heparin and its derivatives. Biochem J 1988; **252**: 515-519.

(b)Redini F, Lafuma C, Hornebeck W, Choay J. and Robert, L. Influence of heparin fragments on the biological activities of elastase(s) and α_1 proteinase inhibitor. Biochem. Pharmacol., 1988; **37**: 4257-4261.

Reichert E, Boerkmann P, Plenat F. Parenteral administration of trypsin triggers lung emphysema. Eur Respir J 1992; **5:** 810-814

Reid L. In: The pathology of emphysema. London, Lloyd Luke, 1967; pp 1-21.

Reilly CF, Tewksbury D, Schechter N, Travis J. Rapid conversion of angiotensin I to angiotensin II by neutrophil and mast cell proteinases. J Biol Chem 1982; **257**: 8619-8622.

Reilly CF, FukunagaY, Powers JC, Travis J. Effect of neutrophil cathepsin G on elastin degradation by neutrophil elastase. Hoppe-Seyler's Physiol Chem 1984; **365**: 1131-1135.

Remold-O'Donnel E, Nixon JC, Rose RM. Elastase inhibitor. Characterization of the human elastase inhibitor molecule associated with monocytes, macrophages, and neutrophils. J Exp Med 1989; **169**: 1071-1086.

Remold-O'Donnel E, Chin J, Alberts M. Sequence and molecular charecterization of human monocyte/neutrophil elastase inhibitor. Proc Natl Acad Sci USA 1992; **89:** 5635-5639.

Rice KG, Rottink MK, Linhardt RJ. Fractionation of heparin-derived oligosaccharides by gradient polyacrylamide-gel electrophoresis. Biochem J 1987; **244:** 515-522.

Rice WG, Weiss SJ. Regulation of proteolysis at the neutrophil-substrate interface by secretory leukoprotease inhibitor. Science 1990; **249**: 178-181.

Riley DJ, Berg RA, Edelman NH, Prockop DJ. Prevention of collagen deposition following pulmonary oxygen toxicity in the rat by cis-4-hydroxy-1-proline. J Clin Invest 1980; 65: 643-651.

Rosenberg S, Barr PJ, Najarian RC, Hallewell RA. Synthesis in yeast of a functional oxidantresistant mutant of human α 1-antitrypsin. Nature 1984; **312**: 77-80.

Rudolphus A, Stolk J. Dijkman JH, Kramps JA. Inhibition of lipopolysaccharide-induced pulmonary emphysema by intratracheally instilled recombinant secretory leukocyte proteinase inhibitor. Am Rev Respir Dis 1993; **147**: 442-447.

Rudolphus A, Kramps JA, Mauve I, Dijkman JH. Intracheally-instilled antileukoprotease and alpha 1-proteinase inhibitor: effect on human neutrophil elastase-induced experimental emphysema and pulmonary location. Histochem J 1994; **26**: 817-824.

Rybojad M, Baumann C, Godeau G, Moraillon I, Prigent F, Morel P, Bourrat E. Congenital generalized cutis laxa: 5 cases. Ann Dermatol Venereol 1999; **126**: 317-319.

Sakamaki F, Ishizaki A, Urano T, Sayama K, Nakamura H, Terashima T, Waki Y, Tasaka S, Hasegawa N, Sato K, Nakagawa N, Obata T, Kanazawa M. Effect of a specific neutrophil elastase inhibitor, ONO-546, on endotoxin-induced acute lung injury. Am J Respir Crit Care Med 1996; **153:** 391-397.

Sallenave JM, Silva A, Marsden ME, Ryle AP. Secretion of mucus proteinase inhibitor and elafin by Clara cell and type II pneumocyte cell lines. Am J Respir Cell Mol. Biol 1993; 8: 126-133.

(a)Sallenave JM, Si-Ta har M, Cox G, Chignard M, Gauldie J. Secretory leukocyte proteinase inhibitor is a major leukocyte elastase in human neutrophils. J Leukoc Biol 1997; **61**: 695-702.

(b)Sallenave JM, Porteus DJ, Haslett C. Gene therapy for lung inflammatory diseases: not so far away? Thorax 1997; **52**: 742-744.

Samet JM, Marbury MC, Spengler JD. Health effects and sources of indoor air pollution. Part I. Am Rev Respir Dis 1987; **136**: 1486-1508.

Sandberg LB, Soskel NT, Leslie JG. Elastin structure, biosynthesis, and relation to disease states. N Engl J Med 1981; **304:** 566-579.

Sanborg RR, Smolen JE. Biology of disease: Early biochemical events in leukocyte activation. Lab Invest 1988; **59**: 300-320.

Savill J. Macrophage recognition of senescent neutrophils. Clin Sci 1992; 83: 649-655.

Schiessler H, Hochstrasser K, Ohlsson K. Acid stable inhibitors of granulocyte neutral proteases in human mucous secretions: biochemistry and possible biological functions. In: Havemann K, Janof A. (eds). Neutral proteases of human polymorphonuclear leukocytes. Urban & Schwarzenberg, Munchen Wien Baltimore, 1978; pp195-207.

Schnebli HP. Recombinant elastase inhibitors for therapy. Ann NY Acad Sci 1991; 624: 212-218.

Schreider JP, Raabe OG. Structure of the Respiratory acinus. Am J Anat 1957; 162: 221-232.

Schriver EE, Davidson JM, Sutcliffe MC, Swindell BB, Bernard GR. Comparison of elastin peptide concentrations in body fluids from healthy volunteers, smokers, and patients with chronic obstructive pulmonary disease. Am Rev Respir Dis 1992; 145: 762-766.

Seersholm N, Kok-Jensen A, Dirksen A. Decline in FEV1 among patients with severe heriditary alpha 1-antitrypsin deficiency type PiZ. Am J Crit Care Med 1995; **152**: 1922-1925.

Seersholm N, Wencker M, Banik N, Viskum K, Dirksen A, Kok-Jensen A, Konietzko N. Does alpha 1-antitrypsin augmentation therapy slow the annual decline in FEV1 in patients with severe hereditary alpha 1-antitrypsin deficiency? Eur Respir J 1997; **10**: 2260-2263.

Senior RM, Tegner H, Kuhn C, Ohlsson K, Starcher BC, Pierce JA. The induction of pulmonary emphysema induced with human leukocyte elastase. Am Rev Respir Dis 1977; **116**: 469-475.

Senior RM, Griffin GL, Mecham RP, Wrenn DS, Prasad KU, Urry DW. Val-Gly-Val-Ala-Pro-Gly, a repeating peptide in elastin, is chemotactic for fibroblasts and monocytes. J Cell Biol 1984; **99**: 870-874.

Senior RM, Griffin GL, Fliszar CJ, Shapiro SD, Goldberg GI, Welgus HG. Human 92- and 72-kilodalton type IV collagenases are elastases. J Biol Chem 1991; **266**: 7870-7875.

Shaheen SO, Sterne JA, Florey CD. Birth weight, childhood lower respiratory tract infection, and adult lung function. Thorax 1998; **53**: 549-553.

Shapiro SD. Elastolytic metalloproteinases produced by human mononuclear phagocytes. Potential roles in destructive lung disease. Am J Respir Crit Care Med 1994; **150:** S160-S164.

Shapiro SD. The pathogenesis of emphysema: the elastase:antielastase hypothesis 30 years later. Proc Assoc Am Physiol 1995; **107**: 346-352.

Shapiro SD. Mighty mice: Transgenic technology "knocks out" questions of matrix metalloproteinase function. Matrix Biology 1997; **15**: 527-533.

Sharp HL, Bridges RA, Krivit W. Cirrhosis with alpha 1 antitrypsin deficiency. A previously unrecognised inherited disorder. J Lab Clin Med 1969; 73: 934-939.

Siafakas NM, Vermeire P, Pride NB, Paoletti P, Gibson J, Howard P, Yernault JC, Decramer M, Higenbottam T, Postma DS, Rees J, on behalf of the Task Force. Optimal assessmment and management of chronic obstructive pulmonary disease (COPD). Eur Respir J 1995; **8**: 1398-1420.

Silverman EK, Speizer FE. Risk factors for the development of chronic obstructive pulmonary disease. Med Clin North America 1996; **80:** 501-522.

Silvers GW, Petty TL, Stanford RE. Elastic recoil changes in early emphysema. Thorax 1980; 35: 490-495.

Smoking and health. Report of the advisory committee to the surgeon general of the public health service, Washington DC: US department of health, education, and welfare, USDE publication 1103; 1964.

Snider GL, Hayes JA, Korthy Al, Lewis GP. Centrilobular emphysema experimentally induced by cadmium chloride aerosol. Am Rev Respir Dis 1973; **108**: 40-48.

Snider GL, Hayes JA, Franzblau C, Kagan HM, Stone PJ, Korthy AL. Relationship between elastolytic activity and experimental emphysema-inducing properties of papain preparations. Am Rev Respir Dis 1974; **110:** 254-262.

Snider GL, Lucey EC, Christensen TG, et al.: Emphysema and bronchial secretory cell metaplasia induced in hamsters by human neutrophil products. Am Rev Respir Dis 1984; **129**: 155-160.

Snider GL. Chronic obstructive pulmonary disease-A continuing challenge. Am Rev Respir Dis 1986; **133**: 942-944.

Snider GL, Kleinerman J, Thurlbeck WM. The definition of emphysema. Report of the National Heart, Lung and Blood Institute, Division of Lung Diseases Workshop. Am Rev Respir Dis 1985; **132**: 182-185.

Snider GL, Lucey EC, Stone PJ. Animal models of emphysema. Am Rev Respir Dis 1986; **133**: 149-169.

Snider GL, Lucey EC, Faris B, Jung-Legg Y, Stone PJ, Franzblau C. Cadmium chloride-induced airspace enlargement with interstitial pulmonary fibrosis is not associated with destruction of lung elastin. Implications for the pathogenesis of human emphysema. Am Rev Respir Dis 1988; **137**: 918-923.

Snider GL. Chronic obstructive pulmonary disease: risk factors, pathophysiology and pathogenesis. Annu Rev Med 1989; **40**: 41-429.

Snider GL. Emphysema; the first two centuries and beyond. Am Rev Respir Dis 1992; **146**: 1334-44, 1615-22

Snider GL, Stone PJ, Lucey EC. Development and evaluation of antiproteases as drugs for preventing emphysema. In: Grassi C, Travis J, Casali L, Luisetti M, eds. Biochemistry of pulmonary emphysema. Verona: Bi & Gi Publishers; 1992: 143-158.

Snider GL, Stone PJ, Morris TA, Franzblau C. Evidence that augmentation therapy diminishes degradation of cross-linked elastin in alpha-1-antitrypsin deficient humans. Am Rev Respir Dis 1993; 147: A674.

Sommerhoff CP, Krell RD, Williams JL, Gomes BC, Strimpler AM, Nadel JA. Inhibition of human neutrophil elastase by ICI 200,355. Eur J Pharmacol 1991; **193**: 153-158.

Soskel NT, Watanabe S, Hardie R, Shenvi AB, Punt JA, Kettner C. Effects of dosage and timing of administration of a peptide boronic acid inhibitor on lung mechanics and morphometrics in elastase-induced emphysema in hamsters. Am Rev Respir Dis 1986; **133**: 635-638.

Spitznagel NK, Dalldorf .G, Leffel MS, Folds JD, Welsh IRH, Cooney MH, Martin LE. Lab Invest 1974; **30**: 774-785.

Staprans I, Felts M. Isolation and characterisation of glycosaminoglycans in human plasma. J Clin Invest 1985; **76**: 1984-1991.

Starcher B, Scott M. Fractionation of urine to allow desmosine analysis by radioimmunoassay. Ann Clin Biochem 1992; **29:** 72-78.

Starkey PM, Barrett AJ. Human lysomal elastase: catalytic and immunological properties. Biochem J 1976; **155**: 265-271.

Stein RL, Strimpler AM, Hori H, Powers JC. Catalysis by human leukocyte elastase: Mechanistic insights into specificity requirements. Biochemistry 1987; **26:** 1301-1305.

Stockley RA, Mistry M, Bradwell AR, Burnett D. A study of plasma proteins in the sol phase of sputum from patients with chronic bronchitis. Thorax, 1979; **34:** 777-782.

Stockley RA, Ohlsson K. Serum studies of leukocyte elastase in acute and chronic lung diseases. Thorax 1982; **37:** 114-117.

Stockley RA, Afford SC. Qualitative studies of lung lavage α_1 -proteinase inhibitor. Hoppe Seylers Z. Physiol Chem 1984; **365**: 503-510.

Stockley RA, Morrison HM, Kramps JA, Dijkman JH, Burnett D. Elastase inhibitors of sputum sol phase: variability, relation to neutrophil elastase inhibition, and effect of corticosteroid treatment. Thorax 1986; **41:** 442-447.

Stolk J, Camps J, Feitsma HI, Dijlman JH, Pauwels EK. Pulmonary deposition and disappearance of aerosolised secretory leucocyte protease inhibitor. Thorax 1995; **50**: 645-650.

Stone PJ, Calore JD, Snider GL, Franzblau, C. The dose-dependent fate of enzymatically active and inactivated tritiated methylated pancreatic elastase administered intratracheally in the hamster Am Rev Respir Dis 1979; **120**: 577-587.

Stone PJ, Lucey EC, Calore JD, Snider GL, Franzblau C, Castillo MJ, Powers JC. The moderation of elastase induced emphysema in the hamster by intratracheal pretreatment or post-treatment with succinyl alanyl prolyl value chloromethyl ketone. Am Rev Respir Dis 1981; **124**: 56-59.

Stone PJ, Calore, JD, Snider GL, Franzblau C. The dose-dependent fate of enzymatically active and inactivated tritiated methylated pancreatic elastase administered intratracheally in the hamster Am Rev Respir Dis 1979; **120**: 577-587.

Stone PJ, Calore JD, Snider GL, Franzblau C. Role of α -macroglobulin-elastase complexes in the pathogenesis of elastase-induced emphysema in hamsters. J Clin Invest 1982; **69**: 920-931.

Stone P, Calore, JD, McGowan SE, Bernardo J, Snider GL, Franzblau C. Functional alphaantiprotease inhibitor in the lower Respiratory tract of smokers is not decreased. Science 1983; **221**: 1187-1189.

Stone PJ, Lucey EC, Snider GL. Induction and exacerbation of emphysema in hamsters with human neutrophil elastase inactivated reversibly by a peptide boronic acid. Am Rev Respir Dis 1990; 141: 47-52.

Stone PJ, Bryan-Rhadfi J, Lucey EC, Ciccolella DE, Crombie G, Faris B, Snider GL, Franzblau C. Measurement of urinary desmosine by isotope dilution and high performance chromatography.

Correlation between elastase-induced air-space enlargement in the hamster and elevation of urinary desmosine. Am Rev Respir Dis 1991; 144: 284-290.

(a)Stone PJ, Gottlieb DJ, O'Connor GT, Ciccolella DE, Breur R, Bryan-Rhadfi J, Shaw HA, Franzblau C, Snider GL. Elastin and collagen degradation products in urine of smokers with and without chronic obstructive pulmonary disease. Am J Respir Crit Care Med 1995; **151**: 952-959.

(b) Stone PJ, Konstan MW, Berger HL, Dorkin C, Franzblau C, Snider GL. Elastin and collagen degradation products in urine of patients with cystic fibrosis. Am J Respir Crit Care Med 1995; **152**: 157-162.

(c) Stone PJ, Morris TA, Franzblau C, Snider GL. Preliminary evidence that augmentation therapy diminishes degradation of cross-linked elastin in alpha-1-antitrypsin deficient humans. Respiration 1995; **62:** 76-79.

Streeten EA, Murphy EA, Pyeritz RE. Pulmonary function in the Marfan syndrome. Chest 1987; **91:** 408-412.

Szapiel S, Fulmer JD, Hunninghake GW, Elson NA, Kawanami O, Ferrans VJ, Crystal RJ. Hereditary emphysema in the tight skin (Tsk/+) mouse. Am Rev Respir Dis 1981; **123:** 680-685.

Tardiff J, Krauter KS. Divergent expression of alpha1-protease inhibitor in mouse and human. Nucleic Acids Res 1998; **26**: 3794-3799.

Tarjan E, Magyar P, Vaczi Z, Lantos, A, Vaszar L. Longitudinal lung function study in heterozygous PIMZ phenotype subjects. Eur Respir J 1994; 7: 2199-2204.

Tetley T, Proteinase imbalance: its role in lung disease. Thorax 1993; 48: 560-565

Thurlbeck WM. The incidence of pulmonary emphysema. Am Rev Respir Dis 1963; 87: 206-215.

Thurlbeck WM. Chronic airflow obstruction. In: Pathology of the lung. ed. WM Thurlbeck. 1988; pp.519-578. New York: Theime Medical.

Thurlbeck WM. Chronic airflow obstruction. In: Pathology of the lung. 2nd Edition. eds. WM Thurlbeck, AM Churg. 1995; pp.739-784. New York: Theime Medical.

Thurlbeck WM, Ryder RC, Sternby N. A comparative study of the severity of emphysema in necropsy populations in three different countries. Am Rev Respir Dis 1974; **109:** 239-248

Thurlbeck WM, Simon G. Radiographic appearance of the chest in emphysema. Am J Roentgenol 1978; 14: 429-440.

Tollefsen T, Saltvedt E. Comparitive analysis of gingival fluid and plasma by crossed immunoelectrophoresis. J Periodontal Res 1980; **15**: 96-106.

Travis J. Structure, function, and control of neutrophil proteinases. Am J Med 1988; **84** (Suppl. 6A): 37-42.

Travis J, Bowen J, Bough R. Human alpha- 1 antichymotrypsin: interaction with chymotrypsin like proteinases. Biochemistry 1978; 17: 5651-5656.

Travis J, Salvesen GS. Human plasma proteinase inhibitors. Ann Rev Biochem 1983; 52: 655-709.

Travis J, Owen M, George P, Carrell R, Rosenberg S, Hallewell RA, Barr PJ. Isolation and properties of recombinant DNA-produced variants of human α_1 -proteinase inhibitor. J Biol Chem 1985; **260:** 4384-4389.

Travis J, Fritz H. Potential problems in designing elastase inhibitors for therapy. Am Rev Respir Dis 1991; 143: 1412-1415.

Trefz G, Schlieber B, Heck B, Schulz V, Ebert W. α_1 -Proteinase inhibitor and mucus proteinase inhibitor in human lung emphysema. Clin Invest 1992; **70**: 269-276.

Tremblay GM, Sallenave JM, Israel-Assayag E, Cormier Y, Gauldie J.Elafin/elastase-specific inhibitor in bronchoalveolar lavage of normal subjects and farmer's lung. Am J Respir Crit Care Med 1996; **154:** 1092-1098.

Trevani AS, Andonegui G, Giodani M, Nociari M, Fontan P, Dran G, Geffner, JR. Neutrophil apoptosis induced by proteolytic enzymes. Lab Invest 1996; **711**: 711-721.

Turino GM, Barker AF, Brantly ML, Cohen AB, Connelly RP, Crystal RG, Eden E, Schlutcher MD, Stoller JK. Clinical features of individuals with PI*SZ phenotype of alpha 1-antitrypsin deficiency. alpha 1-Antitrypsin Deficiency Registry Study Group. Am J Respir Crit Care Med 1996; **154**: 1718-1725.

Turner-Stokes L, Turton C, Pope FM, Green M. Emphysema and cutis laxa. Thorax 1983; **38**: 790-792.

Twumasi DY, Liener IE, Galdston M, Levytska V. Activation of human leukocyte elastase by human α_2 -macroglobulin. Nature 1977; **267:** 61-63.

Tyagi SC, Simon SR. Inhibitors to binding domains in neutrophil elastase. Biochemistry 1990; 29: 9970-9977.

Tyagi SC, Simon SR. Regulation of neutrophil elastase activity by elastin-derived peptide. J Biol Chem 1993; **268**: 16513-16518.

U.S. Department of Health and Human Services, 1989. Reducing the health consequences of smoking: 25 years of progress. A report of the Surgeon General. Center for Chronic Disease Prevention and Health Promotion. Office on Smoking and Health. DHHS Publication No. (CDC) 89-8411.

U.S. Surgeon General. The health cosequences of smoking, nicotine addiction. Rockville MD: U.S. Department of Health and Human Services, 1988. USDHS No. (CDC) 88-8406 115, 213.

Van Furth R. Development and distribution of mononuclear phagocytes. In: Gallin, JI, Goldstein, IM, Snyderman, R, eds. Inflammation: basic principles and clinical correlates. New York: Raven Press, Ltd., 1992: 325-340.

Van Steenbergen W. α1-Antitrypsin deficiency: an overview. Acta Clinica Belgica 1993; **48:** 171-189. Verbeken EK., Cauberghs MD, Mertens I, Clement J., Lauweryns JM, Van de Woestijne KP. The senile lung. comparison with normal and emphysematous lungs. 2. Functional aspects. Chest, 1992; **101:** 800-809.

Vlodavosky I, Eldor A, Haimovitz-Freidman A, Matzner Y, Ishai-Michaeli R, Lider O, Naparstek Y, Cohen IR, Fuks Z. Expression of heparanase by platelets and circulating cells of the immune system: possible involvement in diapedesis and extravasation. Invasion Metastasis 1992; **12**: 112-127.

Vogelmeier C, Hubbard RC, Fells GA, Schnebli HP, Thompson RC, Fritz H, Crystal RG. Antineutrophil elastase defence of the normal human respiratory epithelial surface provided by the secretory leukoprotease inhibitor. J Clin Invest 1991; **87:** 482-488.

Vogelmeier C, Gillissen A, Buhl R. Use of secretory leukoprotease inhibitor to augment lung antineutrophil elastase activity. Chest 1996; **110**: 261S-266S.

Wachi H, Seyama Y, Yamashita S, Suganami H, Uemura Y, Okamato K, Yamada H, Tajima S. Stimulation of cell proliferation and autoregulation of elastin expression by elastin peptide VPGVG in cultured chick vascular smooth muscle cells. FEBS lett 1995, **368**: 215-219.

Walsh RL, Dillon TJ, McLennan G. Interference of functional α_1 -protease inhibitor activity by heparin. Am Rev Resp Dis 1988; **137**: A208.

(a)Walsh RL, Dillon TJ, Scicchitano R, McLennan G. Heparin and heparan sulphate are inhibitors of human leukocyte elastase. Clin Sci 1991; **81**: 341-346.

(b)Walsh RL, Dillon T, Scicchitano R, McLennan G. Interaction of surfactants with heparin and heparan sulphate inhibition of human leukocyte elastase. Am Rev Resp Dis 1991; **143**: A326.

Walsh RL, Dillon TJ, McLennan G. A rapid procedure for the measurement of elastase inhibitory function of plasma and bronchoalveolar lavage fluid. Clin Biochem 1992; **25**: 263-270

Walsh RL, Edwards JB, Holmes M, Scicchitano R, McLennan G. Inhibition of human neutrophil elastase by heparin fractions. Potential implications for therapy of inflammatory lung disease. Am Rev Resp Dis 1993; **147:** A304.

Walter M, Plotnick M, Schecher NM. Inhibition of human mast cell chymase by secretory leukocyte proteinase inhibitor: enhancement of the interaction by heparin. Arch Biochem Biophys 1996; **327:** 81-88.

Wang Z, Zheng T, Zhu Z, Homer RJ, Riese RJ, Chapman HA Jr, Shapiro SD, Elias JA. Interferon gamma induction of pulmonary emphysema in the adult murine lung. J Exp Med 2000; **192:** 1587-1600.

Waters ATH. Researches on the nature, pathology and treatment of emphysema of the lungs, and its relation with other diseases of the chest. London: J. Churchil, 1862.

Weibel ER. Morphometry of the human lung. Berlin, Springer Verlag, 1963.

Weidinger S, Poller W, Faber JP, Schwarzfischer F. Improved phenotyping of alpha 1antichymotrypsin (ACT) by isoelectric focusing and immunoprinting: first demonstration of a deficient protein variant in the ACT system. Electrophoresis 1992 13: 78-780.

Weiland JE, Davis WB, Holter JF, Mohammed JR, Dorinsky PM, Gadek JE. Lung neutrophils in the adult Respiratory distress syndrome: clinical and pathophysiologic influence. Am Rev Respir Dis 1986; **133**: 218-225

Weiss SJ. Tissue destruction by neutrophils. N Engl J Med 1989; 320: 365-376.

Weiss SJ, Regiani S. Neutrophils degrade subendothelial matrices in the prescence of alpha-1proteinase inhibitor: cooperative use of lysozymal proteinases and oxygen metabolites. J Clin Invest 1984; 73: 1297-1303.

Weitz JI, Landman SL, Crowley KA, Birken S, Morgan FJ. Development of an assay for in vivo human neutrophil elastase activity in patients with alpha-1-antiprotease inhibitor deficiency. J Clin Invest 1986; **78**: 155-162.

Weitz JI, Crowley KA, Landman SL, Lipman, BI, Yu J. Increased neutrophil elastase activity in cigarette smokers. Ann Intern Med 1987; **107:** 680-682.

Weitz JI, Silverman EK, Thong B, Campbell EJ. Plasma levels of elastase-specific fibrinopeptides correlate with proteinase inhibitor phenotype. Evidence for increased elastase activity in subjects with homozygous and heterozygous deficiency of alpha 1-proteinase inhibitor. J Clin Invest 1992; **89:** 766-773.

Welgus HG, Campbell EJ, Cury, JD, Eisen AZ, Senior RM, Wilhelm SM, Goldberg GI. Neutral metalloproteinases produced by human mononuclear phagocytes: enzyme profile, regulation, and expression during cellular development. J Clin Invest 1990; **86**:1496-1502.

Wenzel HR, Feldmann A, Engelbrecht S, Tschesche H. Activation of human leucocyte proteinases elastase and cathepsin G by various surfactants. Biol. Chem. Hoppe-Seyler, 1990; **371**: 721-724.

Wewers MD. Pathogenesis of emphysema. Assessment of basic science concepts through clinical investigation. Chest 1989; **95**: 190-195.

Wewers MD, Casolara MA, Crystal RG. Comparison of alpha-1-antitrypsin levels and neutrophil elastase capacity of blood and lung in a patient with the alpha-1-antitrypsin phenotype null-null before and during alpha-1-antitrypsin augmentation therapy. Am Rev Respir Dis 1987; **135**: 539-543.

Wewers MD, Herzyk DJ, Gadek JE. Alveolar fluid neutrophil elastase activity in the adult respiratory distress syndrome is complexed to α -2-macroglobulin. J Clin Invest 1988; **82:** 1260-1267.

Wewers MD, Herzyk DJ, Gadek JE. Comparison of smoker and nonsmoker lavage fluid for the rate of association with neutrophil elastase. Am J Respir Cell Mol Biol 1989; 1: 423-429.

White R, Habicht GS, Godfrey HP, Janoff A, Barton E, Fox C. Secretion of elastase and alpha-2-macroglobulin by cultured murine peritoneal macrophages: studies on their interactions. J Lab Clin Med 1981; **97:** 718-728.

Willems LNA, Otto-Verberne CJM, Kramps JA, ten Have Opbroeck AAW, Dijkman JH. Detection of antileukoprotease in connective tissue of the lung. Histochemistry 1986; **86:** 165-168.

Willems LNA, Kramps JA, Stijnen T, Sterk PJ, Weening JJ, Dijkman JH. Antileukoproteasecontaining bronchiolar cells. Am Rev Respir Dis 1989; **139**: 1244-1250.

Williams HR, Lin T-Y, Navia MA, Springer JP, McKeever BM, Hoogsteen K, Dorn CP, Jr. Crystallization of human neutrophil elastase. J Biol Chem 1987; **262**: 17178-17181.

Williams JC, Falcone RC, Knee C, Stein RL, Strimpler AM, Reaves B, Giles RE, Krell RD. Biologic characterisation of ICI 200,880 and ICI 200,355, novel inhibitors of human neutrophil elastase. Am Rev Respir Dis 1991; **144**: 875-883.

Wilson TA, Hyatt RE, Rodarte JR. The mechanisms that limit expiratory flow. Lung 1980; **158**: 193-200.

Wittels EH, Coalson JJ, Welch MH, Guenter CA. Pulmonary intravascular leukocyte sequestration: a potential mechanism of lung injury. Am Rev Respir Dis 1974; **109:** 502-509.

Wright GW, Kleinerman J. A consideration of the aetiology of emphysema in terms of contemporary knowledge. Am Rev Respir Dis 1963; **88:** 605

Wright JL Small airways disease: it role in chronic airflow obstruction. Semin Respir Med 1992; 13: 72-85

Wright JL, Hobson J, Wiggs BR, Hogg JC. Comparison of inflammatory cells in bronchoalveolar fluid with those in the lumen and tissue peripheral airways and alveolar air space. Lung 1988; **166**: 75-83.

Wright JL, Churg, A. Cigarette smoke causes physiologic and morphologic changes of emphysema in the guinea pig. Am Rev Respir Dis 1990; **142**: 1422-1428.

Wright JL, Sun JP. Effect of smoking cessation on pulmonary and cardiovascular function and structure: analysis of guinea pig model. J Appl Physiol 1994; 2163-2168.

Wright RR. Elastic tissue of normal and emphysematous lungs. A tridimensional histologic study. Am J Pathol 1961; **39:** 355-367.

Yamada S, Sakamoto K, Tsuda H, Yoshida K, Sugiura M, Sugahara K. Srtuctural studies of octasaccharides derived from the low-sulfated repeating disaccharide region and octasaccharide serines derived from the protein linkage region of porcine intestinal heparin. Biochemistry 1999; **38**: 838-847.

Yasutake A, Powers JC. Reactivity of human leucocyte elastase and porcine pancreatic elastase toward peptide 4-nitroanilides containing desmosine residues. Evidence that human leucocyte elastase is selective for cross linked regions of elastin. Biochemistry 1981; **20:** 3675-3679.

Yoshioka A, Betsuyaku T, Nishimura M, Miyamoto K, Kondo T, Kawakami Y. Excessive neutrophil elastase in bronchoalveolar lavage fluid in subclinical emphysema. Am J Respir Crit Care Med 1995; **152**: 2127-2132.

Zaslow MC, Clark RA, Stone PJ, Calore JD, Snider GL, Franzblau C. Human neutrophil elastase does not bind to alpha 1 protease inhibitor that has been ex-posed to activated human neutrophils. Am Rev Respir Dis 1983; **128**: 434-439.

Zheng T, Zhu Z, Wang Z, Homer RJ, Ma B, Riese RJ, Chapman HA Jr, Shapiro SD, Elias JA. Inducible targeting of IL-13 to the adult lung causes matrix metalloproteinase- and cathepsindependent emphysema. J Clin Invest 2000; **106**: 1081-1093.